UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended October 3, 2015

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to ____

Commission File Number 001-36801



Qorvo, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

46-5288992

(I.R.S. Employer Identification No.)

7628 Thorndike Road, Greensboro, North Carolina 27409-9421

and

2300 N.E. Brookwood Parkway, Hillsboro, Oregon 97124

(Address of principal executive offices)

(Zip Code)

(336) 664-1233 and (503) 615-9000

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \blacksquare

Accelerated filer \Box

Non-accelerated filer □ (Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

As of November 6, 2015, there were 141,341,120 shares of the registrant's common stock outstanding.

INDEX

PART I - FINANCIAL INFORMATION

Item 1. Financial Statements (Unaudited).	
Condensed Consolidated Balance Sheets as of October 3, 2015 and March 28, 2015	<u>3</u>
Condensed Consolidated Statements of Income for the three and six months ended October 3, 2015 and September 27, 2014	<u>4</u>
Condensed Consolidated Statements of Comprehensive Income for the three and six months ended October 3, 2015 and September 27, 2014	<u>5</u>
Condensed Consolidated Statements of Cash Flows for the six months ended October 3, 2015 and September 27, 2014	<u>6</u>
Notes to Condensed Consolidated Financial Statements	<u>7</u>
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.	<u>17</u>
Item 3. Quantitative and Qualitative Disclosures About Market Risk.	<u>25</u>
Item 4. Controls and Procedures.	<u>25</u>
PART II — OTHER INFORMATION	<u>25</u>
Item 1A. Risk Factors.	<u>25</u>
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.	<u>25</u>
Item 6. Exhibits.	<u>27</u>
<u>SIGNATURES</u>	<u>28</u>
EXHIBIT INDEX	<u>29</u>

2

Page

PART I — FINANCIAL INFORMATION

ITEM 1.

QORVO, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands) (Unaudited)

Current assets: Important asset: Importasset: Important asset: Importa		 October 3, 2015	 March 28, 2015
Cash and cash equivalents S 169,597 S 299,814 Short-term investments (Note 6) 25,060 244,830 Accounts receivable, less allowance of \$162 and \$539 as of October 3, 2015 and March 28, 2015, respectively 438,266 353,830 Inventories (Note 3) 393,700 346,000 Prepaid expenses 55,550 52,169 Other receivables 27,245 25,816 Other current assets 22,532 26,538 Deferred tax assets (Note 5) 143,072 150,008 Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Log-term investments (Note 6) 28,761 4,083 71 5,030 Other non-current assets S 6,532,007 5 6,892,371 Total assets S 6,532,007 5 6,892,371 Carrent liabilities 3,291 1,01,911 5,000 303,183 310,189 Log-term davalles S 2,04,857 \$ <th>ASSETS</th> <th></th> <th></th>	ASSETS		
Short-term investments (Nore 6) 25,960 244,830 Accounts receivable, less allowance of \$162 and \$539 as of October 3, 2015 and March 28, 2015, respectively 383,760 348,266 353,830 Inventories (Note 3) 393,760 346,900 Prepaid expenses 55,550 52,169 Other receivables 27,245 25,816 0ther current assets 22,532 26,538 Deferred tax assets (Note 5) 143,072 150,208 15,00,105 Total current assets 20,15 21,15,00,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,055,228 2,307,229 1,40,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,40,550 2,410,586 Other our-rurent assets 64,101 57,000 78,00 50,000 78,00 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,000 79,			
Accounts receivable, less allowance of \$162 and \$539 as of October 3, 2015 and March 28, 2015, respectively 438,266 353,830 Inventories (Note 3) 393,760 346,900 Prepaid expenses 55,550 52,169 Other receivables 27,245 25,810 Other receivables 22,532 26,638 Deferred tax assets (Note 5) 143,072 150,208 Total current assets 1275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,097 2,140,506 1275,982 2,307,229 Long-term investments (Note 6) 28,761 4,083 4,083 Other current assets 64,101 57,009 7,005 Total assets § 6,532,097 \$ 0,6892,379 LIABULTITES AND STOCKHOLDERS' EQUITY 330,183 310,189 Current liabilities 32,91 10,971 Total assets \$ 204,857 \$ 182,468 Accurds payable \$ 2,048,871 \$ 182,468 Accurds	· · ·	\$	\$ 299,814
respectively 438,266 353,830 Inventories (Note 3) 393,760 346,900 Other receivables 27,245 25,550 Other current assets 22,332 26,538 Deferred tax assets (Note 5) 143,072 150,208 Total current assets 1,275,982 1,000,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 1,413,597 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$609,576 at March 28, 2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 0ther non-current assets 64,101 57,005 Total assets 64,101 57,005 5 6,582,979 5 6,892,379 LABILITIES AND STOCKHOLDERS'EQUITY 213,547 \$ 182,468 Accured liabilities: 32,91 10,971 Total assets \$ 204,857 \$ 182,468 Accured liabilities 325,310 16,971 10	Short-term investments (Note 6)	25,960	244,830
Prepaid expenses 55,550 52,169 Other cecivables 27,245 25,816 Other current assets 22,532 26,538 Deferred tax assets (Note 5) 143,072 150,208 Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$268,926 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 128,2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 2,055,228 2,307,229 Long-term investments (Note 6) 28,701 4,083 2,055,228 2,307,229 Long-term investments (Note 6) 28,701 4,083 2,055,228 2,307,229 Long-term investments (Note 6) 28,701 4,083 2,055,228 2,307,299 Long-term investments (Note 6) 28,701 4,083 2,055,228 2,307,299 Current liabilities 6,532,097 \$ 6,532,097 \$ 8,204,857 \$ 8,204,857 \$ 182,468 Accrurend tiabilitities 3,210 10,971		438,266	353,830
Oher receivables 27,245 25,816 Other current assets 22,532 26,538 Deferred tax assets (Note 5) 1133,072 150,208 Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 28,761 4,083 Other non-current assets 64,101 57,005 10 57,005 Total assets 64,101 57,005 10 58,2319 LABILITIES AND STOCKHOLDERS' EQUITY 5 6,892,379 11,871 Other current liabilities 5 0,31,871 31,871 Other current liabilities 3,291 10,971 Total current liabilities 33,183 310,189 Other current liabilities 33,183 310,189 Other current liabilities 86,342 83,720 Total current liabilities 86,342 83,720	Inventories (Note 3)	393,760	346,900
Other current assets 22,532 26,538 Deferred tax assets (Note 5) 143,072 150,208 Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets 8 6,532,097 \$ 6,892,379 LIABILITIES AND STOCKHOLDERS' EQUITY 2 147,326 131,871 Other current liabilities 3,291 10,971 355,474 325,310 Other current liabilities 33,133 310,189 310,189 Long-term dats liabilities (Note 5) 303,183 310,189 310,189 Deferred ats liabilities (Note 5) 303,183 310,189 310,189 310,189 Long-term liabililities 86,342 88,	Prepaid expenses	55,550	52,169
Deferred tax assets (Note 5) 143,072 150,208 Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 72,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets 64,101 57,005 Total assets 5 6,532,097 5 6,892,379 LIABILITIES AND STOCKHOLDERS' EQUITY 2 113,871 0 5 6,892,379 Current liabilities 3,291 10,971 10,971 10,971 10,971 Total current liabilities 31,871 0,964 325,474 325,310 9 Deferred tax liabilities (Note 5) 303,183 310,189 310,189 310,189 310,189 9 9 9719,219 Stockholders' equity <td< td=""><td>Other receivables</td><td>27,245</td><td>25,816</td></td<>	Other receivables	27,245	25,816
Total current assets 1,275,982 1,500,105 Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Long-term investments (<i>Note 6</i>) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets 5 6,532,097 5 6,892,379 LABLITIES AND STOCKHOLDERS' EQUITY 2 10,971 5 10,971 Current liabilities 3 2,91 10,971 10,971 Total current liabilities 3 3,10,189 3,10,189 Long-term liabilities 3,291 10,971 10,971 Total current liabilities 3,53,474 325,310 Deferred tax liabilities (<i>Note 5</i>) 303,183 310,189 Long-term liabilities 819,999 719,219 Stockholders' equity: 9 719,219 Preferred stock, \$0,001 par va	Other current assets	22,532	26,538
Property and equipment, net of accumulated depreciation of \$682,261 at October 3, 2015 and \$609,576 at March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Long-term investments (<i>Note 6</i>) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets 5 6,532,097 5 6,822,979 LABILITIES AND STOCKHOLDERS' EQUITY 2 182,468 447,326 1182,468 Accround payable \$ 204,857 \$ 182,468 Accrued liabilities 147,326 131,871 010,971 Total current liabilities 335,474 325,310 - Deferred tax liabilities (<i>Note 5</i>) 303,183 310,189 301,189 Long-term liabilities 86,342 83,720 - Other current liabilities 819,999 719,219 Stockholders' equity: - Preferred stock, \$0001 par value; 5,000 shares authorized; 140,976 and 149,059 shares issued and outstanding outstanding at October 3, 201	Deferred tax assets (Note 5)	143,072	150,208
March 28, 2015 972,328 883,371 Goodwill 2,135,697 2,140,586 Intangible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets 5 6,532,097 5 6,892,379 LIABLITIES AND STOCKHOLDERS' EQUITY 147,326 131,871 Other current liabilities 147,326 131,871 0ther current liabilities 3,291 10,971 Total current liabilities (Note 5) 3303,183 310,189 310,189 Long-term debt (Note 4) 75,000 - - Other ong-term liabilities (Note 5) 303,183 310,189 33,201 Long-term debt (Note 4) 75,000 - - - Other ong-term liabilities (Note 5) 819,999 719,219 Stockholders' equity: - - - Preferred stock, \$.0001 par value; 5,000 shares authorized; 140,976 and 149,059 shares issued and outstanding ot Ustanding at October 3,2015	Total current assets	 1,275,982	 1,500,105
Intagible assets, net of accumulated amortization of \$520,548 at October 3, 2015 and \$268,926 at March 28, 20152,055,2282,307,229Long-term investments (Note 6)28,7614,083Other non-current assets64,10157,005Total assets\$ 6,532,097\$ 6,892,379LIABILITIES AND STOCKHOLDERS' EQUITY5204,857\$ 182,468Accounts payable\$ 204,857\$ 182,468Accrued liabilities147,326131,871Other current liabilities3,29110,971Total current liabilities (Note 5)303,183310,189Long-term det (Note 4)75,000-Other during this (Note 5)86,34283,720Total liabilities819,999719,219Stockholders' equity:Preferred stock, \$,0001 par value; 5,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,858Accumulated dificit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160		972,328	883,371
28, 2015 2,055,228 2,307,229 Long-term investments (Note 6) 28,761 4,083 Other non-current assets 64,101 57,005 Total assets \$ 6,532,097 \$ 6,892,379 LIABLITIES AND STOCKHOLDERS' EQUITY V V Current liabilities:	Goodwill	2,135,697	2,140,586
Other non-current assets $64,101$ $57,005$ Total assets \$ $6,532,097$ \$ $6,892,379$ LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities: 147,326 131,871 Other current liabilities 3,291 10,971 Total current liabilities 3,291 10,971 Total current liabilities (<i>Note 5</i>) 303,183 310,189 Long-term debt (<i>Note 4</i>) 75,000 - Other long-term liabilities 86,342 83,720 Total liabilities 819,999 719,219 Stockholders' equity: - - Preferred stock, \$.0001 par value; 5,000 shares authorized; 140,976 and 149,059 shares issued and outstanding outstanding at October 3, 2015 and March 28, 2015, respectively - - Accumulated other comprehensive loss, net of tax (1,281) (124) (124) Accumulated deficit (404,479) (410,963) -		2,055,228	
Stal assets S 6,532,097 5 6,892,379 LIABILITIES AND STOCKHOLDERS' EQUITY	Long-term investments (Note 6)	28,761	4,083
LIABILITIES AND STOCKHOLDERS' EQUITY (1,1,2,1,0) Current liabilities: \$ 204,857 \$ 182,468 Accouds payable \$ 204,857 \$ 182,468 Accrued liabilities 147,326 131,871 Other current liabilities 3,291 10,971 Total current liabilities (Note 5) 303,183 310,189 Long-term debt (Note 4) 75,000 - Other liabilities 86,342 83,720 Total liabilities 819,999 719,219 Stockholders' equity: - - Preferred stock, \$.0001 par value; 5,000 shares authorized; 140,976 and 149,059 shares issued and outstanding outstanding at October 3, 2015 and March 28, 2015, respectively 6,117,858 6,584,247 Accumulated other comprehensive loss, net of tax (1,281) (124) Accumulated deficit (400,479) (410,963) Total stockholders' equity 5,712,098 6,173,160	Other non-current assets	64,101	57,005
Current liabilities: \$ 204,857 \$ 182,468 Accounts payable \$ 204,857 \$ 182,468 Accrued liabilities 147,326 Other current liabilities 147,326 Total current liabilities 3,291 Total current liabilities (<i>Note 5</i>) 303,183 Deferred tax liabilities (<i>Note 5</i>) 303,183 Long-term debt (<i>Note 4</i>) 75,000 Other long-term liabilities 86,342 Rockholders' equity: 819,999 Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding - Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding outstanding at October 3, 2015 and March 28, 2015, respectively 6,117,858 Accumulated other comprehensive loss, net of tax (1,281) (124) Accumulated deficit (404,479) (410,963) Total stockholders' equity 5,712,098 6,173,160	Total assets	\$ 6,532,097	\$ 6,892,379
Accounts payable \$ 204,857 \$ 182,468 Accrued liabilities 147,326 131,871 Other current liabilities 3,291 10,971 Total current liabilities (Note 5) 303,183 310,189 Long-term debt (Note 4) 75,000 — Other long-term liabilities 86,342 83,720 Total liabilities 819,999 719,219 Stockholders' equity: — — Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding outstanding at October 3, 2015 and March 28, 2015, respectively 6,117,858 6,584,247 Accumulated other comprehensive loss, net of tax (1,281) (124) Accumulated deficit (404,479) (410,963) Total stockholders' equity — 5,712,098 6,173,160	LIABILITIES AND STOCKHOLDERS' EQUITY		
Accrued liabilities 147,326 131,871 Other current liabilities 3,291 10,971 Total current liabilities 355,474 325,310 Deferred tax liabilities (Note 5) 303,183 310,189 Long-term debt (Note 4) 75,000 — Other long-term liabilities 86,342 83,720 Total liabilities 86,342 83,720 Total liabilities 819,999 719,219 Stockholders' equity:	Current liabilities:		
Other current liabilities $3,291$ $10,971$ Total current liabilities $355,474$ $325,310$ Deferred tax liabilities (Note 5) $303,183$ $310,189$ Long-term debt (Note 4) $75,000$ $-$ Other long-term liabilities $86,342$ $83,720$ Total liabilities $819,999$ $719,219$ Stockholders' equity: $ -$ Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding $-$ Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively $6,117,858$ Accumulated other comprehensive loss, net of tax $(1,281)$ (124) Accumulated deficit $(404,479)$ $(410,963)$ Total stockholders' equity $5,712,098$ $6,173,160$	Accounts payable	\$ 204,857	\$ 182,468
Total current liabilities355,474325,310Deferred tax liabilities (Note 5)303,183310,189Long-term debt (Note 4)75,000—Other long-term liabilities86,34283,720Total liabilities819,999719,219Stockholders' equity:——Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding——Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding——Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Accrued liabilities	147,326	131,871
Deferred tax liabilities (Note 5)303,183310,189Long-term debt (Note 4)75,000Other long-term liabilities86,34283,720Total liabilities819,999719,219Stockholders' equity:819,999719,219Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstandingCommon stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,8586,584,247Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Other current liabilities	3,291	10,971
Long-term debt (Note 4)75,000—Other long-term liabilities86,34283,720Total liabilities819,999719,219Stockholders' equity:819,999719,219Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding——Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,8586,584,247Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Total current liabilities	 355,474	 325,310
Other long-term liabilities86,34283,720Total liabilities819,999719,219Stockholders' equity:Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding outstanding at October 3, 2015 and March 28, 2015, respectivelyAccumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Deferred tax liabilities (Note 5)	303,183	310,189
Total liabilities819,999719,219Stockholders' equity:Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding——Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,8586,584,247Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Long-term debt (Note 4)	75,000	_
Stockholders' equity:Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding—Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,858Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Other long-term liabilities	86,342	83,720
Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding——Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,8586,584,247Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Total liabilities	 819,999	 719,219
Common stock, \$.0001 par value; 405,000 shares authorized; 140,976 and 149,059 shares issued and outstanding at October 3, 2015 and March 28, 2015, respectively6,117,8586,584,247Accumulated other comprehensive loss, net of tax(1,281)(124)Accumulated deficit(404,479)(410,963)Total stockholders' equity5,712,0986,173,160	Stockholders' equity:		
outstanding at October 3, 2015 and March 28, 2015, respectively 6,117,858 6,584,247 Accumulated other comprehensive loss, net of tax (1,281) (124) Accumulated deficit (404,479) (410,963) Total stockholders' equity 5,712,098 6,173,160	Preferred stock, \$.0001 par value; 5,000 shares authorized; no shares issued and outstanding	_	_
Accumulated deficit (404,479) (410,963) Total stockholders' equity 5,712,098 6,173,160		6,117,858	6,584,247
Accumulated deficit (404,479) (410,963) Total stockholders' equity 5,712,098 6,173,160	Accumulated other comprehensive loss, net of tax	(1,281)	(124)
Total stockholders' equity 5,712,098 6,173,160	•		\$ × *
		\$ 6,532,097	\$ 6,892,379

See accompanying Notes to Condensed Consolidated Financial Statements.

QORVO, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF INCOME (In thousands, except per share data)

(Unaudited)

		Three Mo	nths	Ended		Six Mon	Six Months Ended				
	0	October 3, 2015		September 27, 2014		October 3, 2015		September 27, 2014			
Revenue	\$	708,335	\$	362,667	\$	1,381,976	\$	678,988			
Cost of goods sold		423,487		195,216		817,611		369,268			
Gross profit		284,848		167,451		564,365		309,720			
Operating expenses:											
Research and development		118,293		48,567		235,503		93,153			
Marketing and selling		105,925		19,179		215,570		38,069			
General and administrative		29,069		17,754		65,152		36,819			
Other operating expense (Note 10)		13,522		6,695		31,436		20,303			
Total operating expenses		266,809		92,195		547,661		188,344			
Income from operations		18,039		75,256		16,704		121,376			
Interest expense		(660)		(195)		(1,208)		(669)			
Interest income		472		40		864		75			
Other income, net		381		137		4,500		521			
Income before income taxes		18,232		75,238		20,860		121,303			
Income tax expense (Note 5)		(13,784)		(11,927)		(14,376)		(19,345)			
Net income	\$	4,448	\$	63,311	\$	6,484	\$	101,958			
Net income per share (Note 2):											
Basic	\$	0.03	\$	0.88	\$	0.04	\$	1.42			
Diluted			_		_		_				
Diluted	\$	0.03	\$	0.85	\$	0.04	\$	1.38			
Weighted average shares of common stock outstanding (<i>Note 2</i>):											
Basic		146,053		72,211		147,627		71,888			
Diluted		150,783		74,134		152,562		73,897			

See accompanying Notes to Condensed Consolidated Financial Statements.

QORVO, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (In thousands)

(Unaudited)

		Three Mo	nths Ended			led		
	October	3, 2015	Septem 20		Octob	er 3, 2015	Sej	ptember 27, 2014
Net income	\$	4,448	\$	63,311	\$	6,484	\$	101,958
Other comprehensive (loss) income:								
Unrealized gain (loss) on marketable securities, net of tax		(105)		3,139		707		3,138
Foreign currency translation adjustment, including intra-entity foreign currency transactions that are of a long-term-investment nature		(97)		(141)		25		(107)
Reclassification adjustments, net of tax:								
Realized gain on sale of marketable securities		(30)		—		(1,958)		
Amortization of pension actuarial loss		34		7		69		14
Other comprehensive (loss) income		(198)		3,005		(1,157)		3,045
Total comprehensive income	\$	4,250	\$	66,316	\$	5,327	\$	105,003

See accompanying Notes to Condensed Consolidated Financial Statements.

QORVO, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

		Six Mon	ths Ended	
	Oc	tober 3, 2015	September 2	7, 2014
Cash flows from operating activities:				
Net income	\$	6,484	\$	101,958
Adjustments to reconcile net income to net cash provided by operating activities:				
Depreciation		88,807		24,074
Amortization and other non-cash items		250,940		14,163
Excess tax benefit from exercises of stock options		—		(2,681)
Deferred income taxes		2,388		7,734
Foreign currency adjustments		427		(462)
Loss on assets and other, net		872		1,617
Realized gain on sale of marketable securities		(4,004)		
Stock-based compensation expense		83,900		18,712
Changes in operating assets and liabilities:				
Accounts receivable, net		(84,496)		(80,583)
Inventories		(51,803)		(33,674)
Prepaid expense and other current and non-current assets		(5,469)		(11,155)
Accounts payable and accrued liabilities		31,599		43,634
Income tax payable/recoverable		(585)		18,082
Other liabilities		(8,833)		(6,458)
Net cash provided by operating activities		310,227		94,961
Investing activities:				
Purchase of property and equipment		(169,686)		(29,073)
Proceeds from sale of property and equipment		255		7,365
Purchase of securities available-for-sale		(175,104)	(183,109)
Proceeds from maturities of securities available-for-sale		370,067		131,432
Net cash provided by (used in) investing activities		25,532		(73,385)
Financing activities:				
Proceeds from debt		125,000		
Payment of debt		(50,000)		(87,503)
Debt issuance cost		(1,339)		
Excess tax benefit from exercises of stock options		_		2,681
Proceeds from the issuance of common stock		29,708		16,459
Repurchase of common stock, including transaction costs		(549,940)		
Tax withholding paid on behalf of employees for restricted stock units		(19,430)		(14,402)
Restricted cash associated with financing activities		106		109
Other financing		(23)		(34)
Net cash used in financing activities		(465,918)		(82,690)
Effect of exchange rate changes on cash		(58)		(109)
Net decrease in cash and cash equivalents		(130,217)		(61,223)
Cash and cash equivalents at the beginning of the period		299,814		171,898
Cash and cash equivalents at the end of the period	\$	169,597	-	110,675
Non-cash investing information:				
Capital expenditure adjustments included in liabilities	\$	5,217	\$	14,559
See accompanying Notes to Condensed Consolidated Fir		- , ,		,

See accompanying Notes to Condensed Consolidated Financial Statements.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

1. BASIS OF PRESENTATION AND SIGNIFICANT ACCOUNTING POLICIES

On February 22, 2014, RF Micro Devices, Inc. ("RFMD") and TriQuint Semiconductor, Inc. ("TriQuint") entered into an Agreement and Plan of Merger and Reorganization (as subsequently amended on July 15, 2014, the "Merger Agreement") providing for the business combination of RFMD and TriQuint ("Business Combination") under a new holding company named Qorvo, Inc. (formerly named Rocky Holding, Inc.) ("Qorvo"). The Business Combination closing was effective on January 1, 2015 (fourth quarter of fiscal 2015). For financial reporting and accounting purposes, RFMD was the acquirer of TriQuint. The results presented in the Condensed Consolidated Financial Statements, the Notes to the Condensed Consolidated Financial Statements and in the Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") reflect those of RFMD prior to the completion of the Business Combination. As used herein, all references to "the Company," "we," "us" and "our" prior to January 1, 2015 refer to RFMD and all such references on or after January 1, 2015 refer to Qorvo.

The accompanying Condensed Consolidated Financial Statements have been prepared in conformity with accounting principles generally accepted in the United States. The preparation of these financial statements requires management to make estimates and assumptions, which could differ materially from actual results. In addition, certain information or footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States have been condensed, or omitted, pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). In the opinion of management, the financial statements include all adjustments (which are of a normal and recurring nature) necessary for the fair presentation of the results of the interim periods presented. These Condensed Consolidated Financial Statements should be read in conjunction with Qorvo's audited consolidated financial statements and notes thereto included in Qorvo's Annual Report on Form 10-K for the fiscal year ended March 28, 2015.

The Condensed Consolidated Financial Statements include the accounts of the Company and its wholly-owned subsidiaries. All significant intercompany accounts and transactions have been eliminated in consolidation.

The Company uses a 52- or 53-week fiscal year ending on the Saturday closest to March 31 of each year. The first fiscal quarter of each year ends on the Saturday closest to June 30, the second fiscal quarter of each year ends on the Saturday closest to September 30 and the third fiscal quarter of each year ends on the Saturday closest to December 31. Fiscal 2016 is a 53-week fiscal year and as a result the second fiscal quarter ended October 3, 2015 included 14 weeks compared to 13 weeks for the second fiscal quarter ended September 27, 2014 and the six months ended October 3, 2015 included 27 weeks compared to 26 weeks for the six months ended September 27, 2014.

2. NET INCOME PER SHARE

Pursuant to the terms of the Merger Agreement, effective January 1, 2015, the Company effected a one-for-four reverse stock split of the Company's issued and outstanding shares of common stock. All share and per share information contained in the accompanying Condensed Consolidated Financial Statements, Notes to the Condensed Consolidated Financial Statements and MD&A have been retroactively adjusted to reflect the reverse stock split for all periods presented.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

The following table sets forth a reconciliation of the numerators and denominators in the computation of basic and diluted net income per share (in thousands, except per share data):

	 Three Mo	nths F	Ended		Six Mont	nths Ended			
	October 3, 2015	S	September 27, 2014	October 3, 2015			September 27, 2014		
Numerator:									
Numerator for basic and diluted net income per share — net income available to common stockholders	\$ 4,448	\$	63,311	\$	6,484	\$	101,958		
Denominator:									
Denominator for basic net income per share — weighted average shares	146,053		72,211		147,627		71,888		
Effect of dilutive securities:									
Stock-based awards	4,730		1,923		4,935		2,009		
Denominator for diluted net income per share — adjusted weighted average shares and assumed conversions	 150,783		74,134		152,562		73,897		
Basic net income per share	\$ 0.03	\$	0.88	\$	0.04	\$	1.42		
Diluted net income per share	\$ 0.03	\$	0.85	\$	0.04	\$	1.38		

In the computation of diluted net income per share for the three and six months ended October 3, 2015, outstanding stock options to purchase approximately 0.2 million shares and 0.1 million shares, respectively, were excluded because the exercise price of the options was greater than the average market price of the underlying common stock and the effect of their inclusion would have been anti-dilutive. In the computation of diluted net income per share for the three and six months ended September 27, 2014, outstanding stock options to purchase less than 0.1 million shares were excluded because the exercise price of their inclusion would have been anti-dilutive.

The computation of diluted net income per share for the three and six months ended September 27, 2014 does not assume the conversion of the Company's previously issued \$175 million initial aggregate principal amount of 1.00% convertible subordinated notes due 2014 (the "2014 Notes"). The 2014 Notes became due on April 15, 2014, and the remaining principal balance of \$87.5 million was paid with cash on hand.

3. INVENTORIES

Inventories are stated at the lower of cost or market determined using the average cost method. The components of inventories are as follows (in thousands):

	October 3, 2015	March	28, 2015
Raw materials	\$ 89,659	\$	71,863
Work in process	203,309		137,306
Finished goods	100,792		137,731
Total inventories	\$ 393,760	\$	346,900

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

4. DEBT

Credit Agreement

On April 7, 2015, the Company and certain material domestic subsidiaries of the Company (the "Guarantors") entered into a five-year unsecured senior credit facility with Bank of America, N.A., as administrative agent (in such capacity, the "Administrative Agent"), swing line lender, and L/C issuer, and a syndicate of lenders (the "Credit Agreement"). The Credit Agreement includes a \$300.0 million revolving credit facility, which includes a \$25.0 million sublimit for the issuance of standby letters of credit and a \$10.0 million sublimit for swingline loans. The Company may request, at any time and from time to time, that the revolving credit facility be increased by an amount not to exceed \$150.0 million. The revolving credit facility is available to finance working capital, capital expenditures and other lawful corporate purposes. The Company's obligations under the Credit Agreement are jointly and severally guaranteed by the Guarantors. As of October 3, 2015, the Company has an outstanding balance of \$75.0 million under the Credit Agreement.

At the Company's option, loans under the Credit Agreement will bear interest at (i) the Applicable Rate (as defined in the Credit Agreement) plus the Eurodollar Rate (as defined in the Credit Agreement) or (ii) the Applicable Rate plus a rate equal to the highest of (a) the federal funds rate plus 0.50%, (b) the prime rate of the Administrative Agent, or (c) the Eurodollar Rate plus 1.0% (the "Base Rate"). All swingline loans will bear interest at a rate equal to the Applicable Rate plus the Base Rate. The Eurodollar Rate is the rate per annum equal to the London Interbank Offered Rate, as published by Bloomberg, for dollar deposits for interest periods of one, two, three or six months, as selected by the Company. The Applicable Rate for Eurodollar Rate loans ranges from 1.50% per annum to 2.00% per annum. The Applicable Rate for Base Rate loans ranges from 0.50% per annum to 1.00% per annum. Interest for Eurodollar Rate loans will be payable at the end of each applicable interest period or at three-month intervals, if such interest period exceeds three months. Interest for Base Rate loans will be payable quarterly in arrears. The Company will pay a letter of credit fee equal to the Applicable Rate multiplied by the daily amount available to be drawn under any letter of credit, a fronting fee, and any customary documentary and processing charges for any letter of credit issued under the Credit Agreement.

The Credit Agreement contains various conditions, covenants and representations with which the Company must be in compliance in order to borrow funds and to avoid an event of default, including financial covenants that the Company must maintain a consolidated leverage ratio not to exceed 2.50 to 1.0 as of the end of any fiscal quarter and a consolidated interest coverage ratio of not less than 3.00 to 1.0 as of the end of any fiscal quarter.

The Credit Agreement also contains customary events of default, and the occurrence of an event of default will increase the applicable rate of interest by 2.00% and could result in the termination of commitments under the revolving credit facility, the declaration that all outstanding loans are due and payable in whole or in part and the requirement of cash collateral deposits in respect of outstanding letters of credit. Outstanding amounts are due in full on the maturity date of April 7, 2020 (with amounts borrowed under the swingline option due in full no later than ten business days after such loan is made).

Convertible Debt

In April 2007, the Company issued the 2014 Notes, which became due on April 15, 2014. The remaining principal balance of \$87.5 million plus interest of \$0.4 million was paid at maturity with cash on hand.

5. INCOME TAXES

Income Tax Expense

The Company's provision for income taxes for the three and six months ended October 3, 2015 and September 27, 2014 has been calculated by applying an estimate of the annual effective tax rate for the full fiscal year to "ordinary" income or loss (pre-tax income or loss excluding unusual or infrequently occurring discrete items) for the three and six months ended October 3, 2015 and September 27, 2014.

The Company's income tax expense was \$13.8 million and \$14.4 million for the three and six months ended October 3, 2015, respectively, and the Company's income tax expense was \$11.9 million and \$19.3 million for the three and six months ended September 27, 2014, respectively. The Company's effective tax rate was 75.7% and 69.0% for the three and six months ended October 3, 2015, respectively, and 15.9% for both the three and six months ended September 27, 2014. The Company's effective tax rate for both the second quarter of fiscal 2016 and the second quarter of fiscal 2015 differed from the statutory rate primarily due to tax rate differences in foreign jurisdictions, state income taxes, domestic tax credits generated, adjustments to



NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued)

(Unaudited)

deferred tax assets and liabilities primarily related to changes in state tax laws, adjustments to the valuation allowance limiting the recognition of the benefit of domestic deferred tax assets, changes in uncertain tax position exposure, and the domestic production activity deduction.

Deferred Taxes

The valuation allowance against net deferred tax assets increased in the second quarter of fiscal 2016 by \$2.4 million from the \$13.8 million balance as of the end of fiscal 2015, primarily due to an increase in domestic deferred tax assets related to domestic state tax credits. A valuation allowance remained against certain domestic and foreign net deferred tax assets as it is more likely than not that the related deferred tax assets will not be realized.

The Company has outstanding domestic federal and state tax net operating loss ("NOLs") carry-forwards that expire in fiscal years 2016 to 2035 if unused. The use of the NOLs that were acquired in prior year acquisitions is subject to certain annual limitations under Internal Revenue Code Section 382 and similar state tax provisions.

Uncertain Tax Positions

The Company's gross unrecognized tax benefits increased from \$59.4 million as of the end of fiscal 2015 to \$63.1 million as of the end of the second quarter of fiscal 2016, due to a \$4.4 million increase related to tax positions taken with respect to the current fiscal year and a \$0.7 million decrease related to tax positions taken with respect to prior fiscal years.

6. INVESTMENTS AND FAIR VALUE MEASUREMENTS

Available-For-Sale

The following is a summary of available-for-sale securities as of October 3, 2015 and March 28, 2015 (in thousands):

			Available-f	for-S	ale Securities		
	 Cost	Gross Unrealized Gains			Gross Unrealized Losses		Estimated Fair Value
October 3, 2015				_			
U.S. government/agency securities	\$ 21,573	\$	20	\$	—	\$	21,593
Auction rate securities	2,150		—		(364)		1,786
Marketable equity securities	856		3,511		—		4,367
Money market funds	168		—		—		168
	\$ 24,747	\$	3,531	\$	(364)	\$	27,914
March 28, 2015							
U.S. government/agency securities	\$ 197,516	\$	8	\$	(17)	\$	197,507
Auction rate securities	2,150				(400)		1,750
Corporate debt	43,164				(17)		43,147
Marketable equity securities	1,594		6,581		—		8,175
Money market funds	48,961		_				48,961
	\$ 293,385	\$	6,589	\$	(434)	\$	299,540

The estimated fair value of available-for-sale securities was based on the prevailing market values on October 3, 2015 and March 28, 2015. Determination of the cost for sold investments is based on the specific identification method.

The gross realized gains and losses recognized on available-for-sale securities for the three months ended October 3, 2015 were insignificant. There were \$4.0 million of gross realized gains and insignificant gross realized losses recognized on available-for-sale securities for the six months ended October 3, 2015. There were no gross realized gains and losses recognized on available-for-sale securities for the three and six months ended September 27, 2014.

The aggregate amount of available-for-sale securities in an unrealized loss position for fewer than 12 months as of October 3, 2015 was \$1.8 million, with \$0.4 million in unrealized losses. The aggregate amount of available-for-sale securities in an

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued)

(Unaudited)

unrealized loss position for fewer than 12 months as of March 28, 2015 was \$112.9 million, with \$0.4 million in unrealized losses. No available-for-sale investments were in a continuous unrealized loss position for 12 months or greater as of October 3, 2015 or as of March 28, 2015.

The amortized cost of available-for-sale investments with contractual maturities is as follows (in thousands):

	Octobe	r 3, 20	15		March	28, 2015		
	 Cost		Estimated Fair Value				Estimated Fair Value	
Due in less than one year	\$ 21,741	\$	21,761	\$	289,641	\$	289,615	
Due after ten years	2,150		1,786		2,150		1,750	
Total investments in debt securities	\$ 23,891	\$	23,547	\$	291,791	\$	291,365	

Other Investments

As previously disclosed, on August 4, 2015, Qorvo's wholly-owned subsidiary, TriQuint, invested \$25.0 million to acquire shares of Series F Preferred Stock of Cavendish Kinetics Limited, a private limited company incorporated in England and Wales. This investment was accounted for as a cost method investment and classified in "Long-term investments" on the Company's Condensed Consolidated Balance Sheet as of October 3, 2015. No impairment was recognized on the Company's cost-method investment during the second quarter of fiscal 2016.

Fair Value Measurements

On a quarterly basis, the Company measures the fair value of its marketable securities, which are comprised of U.S. government/agency securities, corporate debt, auction rate securities (ARS), marketable equity securities, and money market funds. Marketable securities are reported at fair value in "Cash and cash equivalents", "Short-term investments" and "Long-term investments" on the Company's Condensed Consolidated Balance Sheet. The related unrealized gains and losses are included in "Accumulated other comprehensive income (loss)", a component of "Stockholders' equity", net of tax.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

Recurring Fair Value Measurements

The fair value of the financial assets measured at fair value on a recurring basis was determined using the following levels of inputs as of October 3, 2015 and March 28, 2015 (in thousands):

	Total	Acti	oted Prices In ve Markets For entical Assets (Level 1)	nificant Other ervable Inputs (Level 2)
October 3, 2015				
Assets:				
Available-for-sale securities				
U.S. government/agency securities	\$ 21,593	\$	21,593	\$ —
Auction rate securities (1)	1,786		—	1,786
Marketable equity securities	4,367		4,367	—
Money market funds	 168		168	 —
Total available-for-sale securities	 27,914		26,128	1,786
Invested funds in deferred compensation plan (3)	4,929		4,929	 _
Total assets measured at fair value:	\$ 32,843	\$	31,057	\$ 1,786
Liabilities:				
Invested funds in deferred compensation plan (3)	4,929		4,929	_
Total liabilities measured at fair value:	\$ 4,929	\$	4,929	\$
March 28, 2015				
Assets:				
Available-for-sale securities				
U.S. government/agency securities	\$ 197,507	\$	197,507	\$ _
Auction rate securities (1)	1,750		—	1,750
Corporate debt (2)	43,147		—	43,147
Marketable equity securities	8,175		8,175	_
Money market funds	48,961		48,961	—
Total available-for-sale securities	 299,540		254,643	44,897
Invested funds in deferred compensation plan (3)	8,614		8,614	
Total assets measured at fair value:	\$ 308,154	\$	263,257	\$ 44,897
Liabilities:				
Invested funds in deferred compensation plan (3)	8,614		8,614	—
Total liabilities measured at fair value:	\$ 8,614	\$	8,614	\$

(1) ARS are debt instruments with interest rates that reset through periodic short-term auctions. The Company's Level 2 ARS are valued based on quoted prices for identical or similar instruments in markets that are not active.

(2) Corporate debt includes corporate bonds and commercial paper that are valued using observable market prices for identical securities that are traded in less active markets.
(3) The Company's non-qualified deferred compensation plan provides eligible employees and members of the Board of Directors with the opportunity to defer a specified percentage of their cash compensation. The Company includes the assets deferred by the participants in the "Other current assets" and "Other non-current assets" line items of its Condensed Consolidated Balance Sheets and the Company's obligation to deliver the deferred compensation in the "Other current liabilities" and "Other long-term liabilities" line items of its Condensed Consolidated Balance Sheets.

Other Fair Value Disclosures

The carrying values of cash and cash equivalents, accounts receivable, accounts payable and accrued liabilities approximate fair values because of the relatively short-term maturities of these instruments.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

7. SHARE REPURCHASES

On February 5, 2015, the Company's Board of Directors authorized the repurchase of up to \$200.0 million of the Company's outstanding common stock, exclusive of related fees, commissions or other expenses. On August 11, 2015, the Company announced the completion of this \$200.0 million share repurchase program having repurchased approximately 2.4 million shares of the Company's common stock for approximately \$150.0 million during fiscal 2016. On August 11, 2015, the Company announced that its Board of Directors authorized a new share repurchase program to repurchase up to \$400.0 million of the Company's outstanding common stock. On September 10, 2015, the Company announced the completion of this \$400.0 million share repurchase program having repurchased approximately 7.3 million shares of common stock for approximately \$400.0 million during the three months ended October 3, 2015.

8. RECENT ACCOUNTING PRONOUNCEMENTS

In July 2015, Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2015-11, "Inventory (Topic 330): Simplifying the Measurement of Inventory ("ASU 2015-11"). Entities that measure their inventory other than the last-in, last-out and retail inventory methods will measure their inventory at the lower of cost or net realized value. Net realized value is the estimated selling price in the ordinary cost of business less reasonably predictable costs to completion, transportation, or disposal. Currently, inventory is required to be measured at the lower of cost or market where market could be the replacement cost, net realizable value, or net realizable value less an approximated normal profit margin. The new authoritative guidance is effective for fiscal years beginning after December 15, 2016 and interim periods within fiscal years beginning after December 15, 2017. Early adoption is permitted and implementation will be applied on a prospective basis. The Company will adopt the provisions of ASU 2015-11 in the first quarter of fiscal 2018, and is currently evaluating the impact on its consolidated financial statements.

In April 2015, the FASB issued ASU 2015-03, "Interest - Imputation of Interest (Topic 835-30): Simplifying the Presentation of Debt Issuance Costs ("ASU 2015-03"). ASU 2015-03 requires debt issuance costs related to a recognized debt liability be presented in the balance sheet as a direct deduction from the related debt liability's carrying value, which is consistent with the presentation of debt discounts. Given the absence of authoritative guidance within ASU 2015-03 for debt issuance costs related to line-of-credit arrangements, in August 2015, the FASB issued ASU 2015-15, "Presentation and Subsequent Measurement of Debt Issuance Costs Associated with Line-of-Credit Arrangements" ("ASU 2015-15"), which clarifies ASU 2015-03 by stating that the SEC staff would not object to an entity deferring and presenting debt issuance costs as an asset and subsequently amortizing the deferred debt issuance costs ratably over the term of the line-of-credit arrangement, regardless of whether there are any outstanding borrowings on the line-of-credit arrangement. The Company adopted the provisions of ASU 2015-15 and continued to present its debt issuance costs related to its revolving credit facility as an asset and is amortizing these costs ratably over the term of its revolving credit arrangement.

In May 2014, the FASB issued ASU 2014-09, "Revenue from Contracts with Customers" ("ASU 2014-09") that amends existing guidance on revenue recognition. The new guidance is based on principles that an entity will recognize revenue to depict the transfer of goods and services to customers at an amount the entity expects to be entitled to in exchange for those goods and services. The guidance requires additional disclosures regarding the nature, amount, timing, and uncertainty of cash flows and both qualitative and quantitative information about contracts with customers and applied significant judgments. In July 2015, the FASB deferred the effective date for the adoption of ASU 2014-09 by one year. Early adoption is permissible but not before its original effective date of annual reporting periods beginning after December 15, 2016. The new authoritative guidance will become effective for the Company in the first quarter of fiscal 2019, using one of two retrospective methods of adoption. The Company has not determined which method it will adopt and is evaluating the effects the new guidance will have on its consolidated financial statements.

9. OPERATING SEGMENT INFORMATION

The Company's operating segments as of October 3, 2015 are Mobile Products (MP) and Infrastructure and Defense Products (IDP). In the fourth quarter of fiscal 2015, the Company renamed its reportable segments from Cellular Products Group to MP, and Multi-Market Products Group to IDP, as a result of the Business Combination. Additionally, the chief operating decision maker (CODM) elected to discontinue reporting Compound Semiconductor Group as an operating segment.



NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

MP is a leading global supplier of radio frequency (RF) solutions that perform various functions in the cellular radio front end section of smartphones, tablets and other mobile devices. These RF solutions are required for devices that operate under 4G, Wi-Fi and other communications standards. These solutions include various discrete RF components and module configurations, including complete RF front end modules that combine high-performance filters, power amplifiers and switches into single placement solutions.

IDP is a leading global supplier of a broad array of RF solutions to wireless network infrastructure, defense and aerospace markets and short-range connectivity applications for commercial, consumer, industrial and automotive markets. IDP's solutions include high power gallium arsenide ("GaAs") and gallium nitride ("GaN") components and various multichip and hybrid assemblies.

As of October 3, 2015, MP and IDP are separate reportable segments based on the organizational structure and information reviewed by the Company's Chief Executive Officer, who is the Company's CODM, and are managed separately based on the end markets and applications they support. The CODM allocates resources and assesses the performance of each operating segment primarily based on non-GAAP operating income (loss) and non-GAAP operating income (loss) as a percentage of revenue.

The "All other" category includes operating expenses such as stock-based compensation, amortization of purchased intangible assets, acquisition and integration-related costs, intellectual property rights (IPR) litigation costs, restructuring and disposal costs, start-up costs, gain (loss) on assets, and other miscellaneous corporate overhead expenses that the Company does not allocate to its reportable segments because these expenses are not included in the segment operating performance measures evaluated by the Company's CODM. The CODM does not evaluate operating segments using discrete asset information. The Company's operating segments do not record inter-company revenue. The Company does not allocate gains and losses from equity investments, interest and other income, or taxes to operating segments. Except as discussed above regarding the "All other" category, the Company's accounting policies for segment reporting are the same as for the Company as a whole.

The following tables present details of the Company's reportable segments and a reconciliation of the "All other" category (in thousands):

	Three Mo	Ended	Six Months Ended				
	 October 3, 2015	September 27, 2014			October 3, 2015		September 27, 2014
Net revenue:				-			
MP	\$ 578,160	\$	298,461	\$	1,129,047	\$	559,577
IDP	129,205		63,517		250,989		118,704
All other	970		689		1,940		707
Total net revenue	\$ 708,335	\$	362,667	\$	1,381,976	\$	678,988
Income (loss) from operations:							
MP	\$ 171,974	\$	86,725	\$	345,717	\$	156,341
IDP	22,850		15,198		36,922		26,069
All other	(176,785)		(26,667)		(365,935)		(61,034)
Income from operations	 18,039		75,256		16,704		121,376
Interest expense	(660)		(195)		(1,208)		(669)
Interest income	472		40		864		75
Other income, net	381		137		4,500		521
Income before income taxes	\$ 18,232	\$	75,238	\$	20,860	\$	121,303

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued)

(Unaudited)

		Three Mo	Ended	Six Mont	Ended		
	October 3, September 2 2015 2014		September 27, 2014	 October 3, 2015		September 27, 2014	
Reconciliation of "All other" category:							
Stock-based compensation expense	\$	(35,729)	\$	(9,543)	\$ (83,900)	\$	(18,712)
Amortization of intangible assets		(128,028)		(6,801)	(251,230)		(13,767)
Acquisition and integration related costs		(5,589)		(5,461)	(16,003)		(13,914)
Restructuring and disposal costs		(2,403)		(262)	(3,830)		(1,577)
IPR litigation costs		(192)		(1,992)	(340)		(8,006)
Start-up costs		(3,496)		(211)	(7,206)		(325)
Other expenses (including gain (loss) on assets and other miscellaneous							
corporate overhead)		(1,348)		(2,397)	(3,426)		(4,733)
Loss from operations for "All other"	\$	(176,785)	\$	(26,667)	\$ (365,935)	\$	(61,034)

10. BUSINESS ACQUISITION

Effective January 1, 2015, pursuant to the Merger Agreement, RFMD and TriQuint completed a strategic combination of their respective businesses through the "merger of equals" Business Combination under a new holding company named Qorvo, Inc.

The total estimated purchase price was approximately \$5,254.4 million. The allocation of the purchase price reflected in the accompanying financial statements is preliminary and is based upon estimates and assumptions that are subject to change within the measurement period (up to one year from the acquisition date pursuant to ASC 805). The measurement period remains open pending the completion of procedures related to deferred taxes.

The Business Combination resulted in the recognition of \$470.0 million of in-process research and development (IPRD) of which \$350.0 million relates to the MP operating segment and \$120.0 million relates to the IDP operating segment. The IPRD encompasses a broad technology portfolio of product innovations in RF applications for MP and IDP. These technologies include a variety of semiconductor processes in GaAs and GaN for power and switching applications and surface acoustic wave (SAW) and bulk acoustic wave (BAW) structures for filter applications. Included in IPRD are continuous improvements in the process for design and manufacturing as well as innovation in fundamental research areas such as materials, simulation and modeling, circuit design, device packaging and test. As of October 3, 2015, the IPRD projects for the MP operating segment are between 40% to 90% complete with estimated completion dates through the end of fiscal 2017. As of October 3, 2015, the IPRD projects associated with the IDP operating segment are between 60% to 95% complete with estimated completion dates through the end of fiscal 2017. Remaining costs to complete the IPRD projects for the MP operating segment and the IDP operating segment are approximately \$15.0 million to \$25.0 million and \$5.0 million to \$15.0 million, respectively. Upon completion of the development, acquired IPRD assets will be transferred to finite-lived intangible assets and amortized over their useful lives.

The remaining IPRD asset is classified as an indefinite lived intangible asset that is not currently subject to amortization but is reviewed for impairment annually or whenever events or changes in circumstances indicate that the carrying value of such asset may not be recoverable. The IPRD asset will be subject to amortization upon completion of its respective research project and at the start of commercialization. The fair value assigned to the IPRD asset was determined using the income approach based on estimates and judgments regarding risks inherent in the development process, including the likelihood of achieving technological success and market acceptance. If the IPRD project is abandoned, the acquired technology attributable to the project will be expensed in the Condensed Consolidated Statements of Operations.

During the three and six months ended October 3, 2015, the Company incurred approximately \$5.6 million and \$16.0 million, respectively, of integration costs (including stock-based compensation) and approximately \$3.4 million and \$6.3 million, respectively, of restructuring costs (including stock-based compensation) associated with the Business Combination (primarily related to employee termination benefits). During the three and six months ended September 27, 2014, the Company incurred approximately \$1.3 million and \$3.8 million, respectively, of acquisition costs and approximately \$6.1 million and \$12.1 million, respectively, of integration costs associated with the Business Combination.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (continued) (Unaudited)

The acquisition, integration and restructuring costs are being expensed as incurred and are presented in the Condensed Consolidated Statements of Income as "Other operating expense."

11. SUBSEQUENT EVENTS

On November 5, 2015, the Company announced that its Board of Directors has authorized the repurchase of up to \$1.0 billion of the Company's outstanding common stock through November 4, 2016.

Under the share repurchase program, share repurchases will be made in accordance with applicable securities laws on the open market or in privately negotiated transactions. The extent to which the Company repurchases its shares, the number of shares and the timing of any repurchases will depend on general market conditions, regulatory requirements, alternative investment opportunities and other considerations. The program does not require the Company to repurchase a minimum number of shares, and may be modified, suspended or terminated at any time without prior notice.

On November 6, 2015, the Company announced that it proposes to offer, subject to market conditions and other factors, \$1.0 billion aggregate principal amount of senior notes due 2023 and 2025 (the "Notes") to qualified institutional buyers pursuant to Rule 144A under the Securities Act of 1933, as amended (the "Securities Act"), and to certain non-U.S. persons in accordance with Regulation S under the Securities Act. The Company expects to use the net proceeds of the offering for general corporate purposes, including share repurchases and repayment of any amounts outstanding under its revolving credit facility. The Notes will be senior unsecured obligations of the Company and will be guaranteed, jointly and severally, by each of the Company's existing and future direct and indirect wholly-owned U.S. subsidiaries that guarantee the Company's obligations under its existing revolving credit facility.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS.

SAFE HARBOR FOR FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, that relate to our plans, objectives, estimates and goals. Statements expressing expectations regarding our future and projections relating to products, sales, revenues and earnings are typical of such statements and are made under the Private Securities Litigation Reform Act of 1995. Words such as "expect," "anticipate," "intend," "plan," "believe," and "estimate," and variations of such words and similar expressions, identify such forward-looking statements. Our business is subject to numerous risks and uncertainties, including, but not limited to the factors listed below:

- changes in business and economic conditions, including downturns in the semiconductor industry and/or the overall economy;
- our ability to accurately predict market requirements and evolving industry standards in a timely manner;
- our ability to accurately predict customer demand and thereby avoid the possibility of obsolete inventory, which would reduce our profit margins;
- our customers' and distributors' ability to manage the inventory they hold and forecast their demand;
- our ability to successfully integrate acquired businesses, operations, product technologies and personnel as well as achieve expected synergies;
- our ability to achieve cost savings and improve yields and margins on our new and existing products;
- our ability to respond to possible downward pressure on the average selling prices of our products caused by our customers or our competitors;
- our ability to efficiently utilize our capacity, or to acquire or source additional capacity, in response to customer demand;
- the inability of one or more of our customers to access their traditional sources of credit, which could lead them to reduce their level of purchases or seek credit or other accommodations from us;
- our ability to continue to improve our product designs, develop new products in response to new technologies, and achieve design wins;
- our dependence on a limited number of customers for a substantial portion of our revenue;
- our reliance on the U.S. government and on U.S government sponsored programs (principally for defense and aerospace applications) for a portion
 of our revenue;
- our ability to bring new products to market in response to market shifts and to use technological innovation to shorten time-to-market for our products;
- the risks associated with our wafer fabrication facilities, our assembly facilities and our test and tape and reel facilities;
- variability in manufacturing yields;
- variability in raw material costs and availability of raw materials;
- our dependence on third parties, including wafer foundries, wafer starting material suppliers, passive component manufacturers, assembly and packaging suppliers and test and tape and reel suppliers;
- our ability to manage platform provider and customer relationships;



- our ability to procure, commercialize and enforce intellectual property rights (IPR) and to operate our business without infringing on the unlicensed IPR of others;
- the risks associated with security breaches and other similar disruptions, which could compromise our information and expose us to liability and could cause our business and reputation to suffer;
- currency fluctuations, tariffs, trade barriers, tax and export license requirements and health and security issues associated with our foreign
 operations;
- our ability to attract and retain skilled personnel and develop leaders for key business units and functions;
- · failure to realize the anticipated benefits of the Business Combination, including difficulty in integrating the businesses of RFMD and TriQuint; and
- failure to realize the expected amount and timing of cost savings and operating synergies related to the Business Combination.

These and other risks and uncertainties, which are described in more detail in our most recent Annual Report on Form 10-K and in other reports and statements that we file with the SEC, could cause the actual results and developments to be materially different from those expressed or implied by any of these forward-looking statements. Forward-looking statements speak only as of the date they were made and we undertake no obligation to update or revise such statements, except as required by the federal securities laws.

OVERVIEW

Company

On February 22, 2014, RF Micro Devices, Inc. ("RFMD") entered into an Agreement and Plan of Merger and Reorganization, (as subsequently amended on July 15, 2014, the "Merger Agreement"), with TriQuint Semiconductor, Inc. ("TriQuint") providing for the combination of RFMD and TriQuint in a merger of equals ("Business Combination") under a new holding company named Qorvo, Inc. (the "Company" or "Qorvo"). The transactions contemplated by the Merger Agreement were consummated on January 1, 2015, and as a result, TriQuint's results of operations are included in Qorvo's Condensed Consolidated Statements of Income for the three and six months ended October 3, 2015.

For financial reporting and accounting purposes, RFMD was the acquirer of TriQuint in the Business Combination. Unless otherwise noted, "we," "our" or "us" in this report refers to RFMD and its subsidiaries prior to the closing of the Business Combination and to Qorvo and its subsidiaries after the closing of the Business Combination.

The following MD&A is intended to help the reader understand the consolidated results of operations and financial condition of Qorvo. MD&A is provided as a supplement to, and should be read in conjunction with, our Condensed Consolidated Financial Statements and accompanying notes.

We are a leading provider of core technologies and radio frequency ("RF") solutions for mobile, infrastructure and defense and aerospace applications. We have more than 7,000 global employees dedicated to delivering solutions for everything that connects the world. Our design and manufacturing expertise encompasses many semiconductor process technologies, which we source both internally and through external suppliers. We operate worldwide, with design, sales and manufacturing facilities located in Asia, Europe and North America.

We design, develop, manufacture and market our products to leading U.S. and international original equipment manufacturers ("OEMs") and original design manufacturers ("ODMs") in the following operating segments:

• *Mobile Products (MP)* - MP is a leading global supplier of RF solutions that perform various functions in the cellular radio front end section of smartphones, tablets and other mobile devices. These RF solutions are required for devices that operate under 4G, Wi-Fi and other communications standards. These solutions include various discrete RF components and module configurations, including complete RF front end modules that combine high-performance filters, power amplifiers and switches into single placement solutions.



Infrastructure and Defense Products (IDP) - IDP is a leading global supplier of a broad array of RF solutions to wireless network infrastructure, defense and aerospace markets and short-range connectivity applications for commercial, consumer, industrial and automotive markets. IDP's solutions include high power gallium arsenide ("GaAs") and gallium nitride ("GaN") components and various multichip and hybrid assemblies.

As of October 3, 2015, our reportable segments are MP and IDP. These business segments are based on the organizational structure and information reviewed by our Chief Executive Officer, who is our chief operating decision maker (or CODM), and are managed separately based on the end markets and applications they support. The CODM allocates resources and evaluates the performance of each operating segment primarily based on operating income and operating income as a percentage of revenue. In connection with the Business Combination, in the fourth quarter of fiscal 2015 we renamed our Cellular Products Group operating segment as MP and our Multi-Market Products Group operating segment as IDP. Additionally, the CODM elected to discontinue reporting Compound Semiconductor Group as an operating segment (see Note 8 of the Notes to the Condensed Consolidated Financial Statements in Part I, Item 1 of this report for additional information regarding our operating segments).

SECOND QUARTER FISCAL 2016 FINANCIAL HIGHLIGHTS:

- Quarterly revenue increased 95.3% as compared to the second quarter of fiscal 2015, primarily due to the inclusion of TriQuint revenue in the three months ended October 3, 2015 as well as increased demand for our cellular RF solutions for smartphones.
- Gross margin for the quarter was 40.2% as compared to 46.2% for the second quarter of fiscal 2015. This decrease was primarily due to costs related to the Business Combination (including intangible amortization and stock-based compensation) and average selling price erosion. This decrease was partially offset by manufacturing and sourcing-related cost reductions.
- Operating income was \$18.0 million for the second quarter of fiscal 2016 as compared to operating income of \$75.3 million for the second quarter of fiscal 2015. This decrease was primarily due to costs related to the Business Combination (including intangible amortization and stock-based compensation), which was partially offset by increased revenue and manufacturing and sourcing-related cost reductions.
- Diluted earnings per share for the second quarter of fiscal 2016 was \$0.03 as compared to \$0.85 for the second quarter of fiscal 2015 after giving retroactive effect to the one-for-four reverse stock split related to the Business Combination.
- Cash flow from operations was \$168.8 million for the second quarter of fiscal 2016 as compared to \$58.7 million for the second quarter of fiscal 2015. This year-over-year increase was primarily attributable to improved profitability exclusive of non-cash Business Combination expenses.
- Capital expenditures were \$80.3 million for the second quarter of fiscal 2016 as compared to \$19.3 million for the second quarter of fiscal 2015. This year-over-year increase was primarily related to projects for increasing manufacturing capacity.
- During the second quarter of fiscal 2016, we repurchased approximately 9.1 million shares of our common stock for approximately \$500.0 million.
- During the second quarter of fiscal 2016, we recorded merger-related expenses, integration costs and restructuring expenses totaling \$9.0 million related to the Business Combination. We expect merger and integrated-related expenses associated with the Business Combination to decrease in future periods.



RESULTS OF OPERATIONS

Consolidated

The following tables presents a summary of our results of operations for the three and six months ended October 3, 2015 and September 27, 2014 (in thousands, except percentages):

	Three Months Ended								
	 October 3, 2015	% of Revenue		September 27, 2014	% of Revenue		Increase (Decrease)	Percentage Change	
Revenue	\$ 708,335	100.0%	\$	362,667	100.0%	\$	345,668	95.3 %	
Cost of goods sold	423,487	59.8		195,216	53.8		228,271	116.9	
Gross profit	284,848	40.2		167,451	46.2		117,397	70.1	
Research and development	118,293	16.7		48,567	13.4		69,726	143.6	
Marketing and selling	105,925	15.0		19,179	5.3		86,746	452.3	
General and administrative	29,069	4.1		17,754	4.9		11,315	63.7	
Other operating expense	13,522	1.9		6,695	1.8		6,827	102.0	
Operating income	\$ 18,039	2.5%	\$	75,256	20.8%		(57,217)	(76.0)	

		Six Months Ended							
	0	ctober 3, 2015	% of Revenue	Sept	ember 27, 2014	% of Revenue	Increa (Decrea		Percentage Change
Revenue	\$	1,381,976	100.0%	\$	678,988	100.0%	\$ 702	2,988	103.5 %
Cost of goods sold		817,611	59.2		369,268	54.4	448	3,343	121.4
Gross profit		564,365	40.8		309,720	45.6	254	4,645	82.2
Research and development		235,503	17.0		93,153	13.7	142	2,350	152.8
Marketing and selling		215,570	15.6		38,069	5.6	177	7,501	466.3
General and administrative		65,152	4.7		36,819	5.4	28	3,333	77.0
Other operating expense		31,436	2.3		20,303	3.0	11	,133	54.8
Operating income	\$	16,704	1.2%	\$	121,376	17.9%	(104	1,672)	(86.2)

Revenue increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to the inclusion of TriQuint revenue in the three and six months ended October 3, 2015. The remaining increase was primarily due to increased demand for our cellular RF solutions for smartphones.

Gross margin and operating income decreased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to costs related to the Business Combination (including intangible amortization and stock-based compensation) and average selling price erosion. This decrease was partially offset by manufacturing and sourcing-related cost reductions.

Operating Expenses

Research and development expenses increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to the inclusion of TriQuint research and development expenses in the three and six months ended October 3, 2015, and expenses resulting from new product development for mobile products.

Marketing and selling expenses increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to marketing-related intangible asset amortization resulting from the Business Combination and the inclusion of TriQuint marketing and selling expenses in the three and six months ended October 3, 2015.

General and administrative expenses increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, due to the inclusion of TriQuint general and administrative expenses in the three and six months ended October 3, 2015.

Other operating expense increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to expenses associated with the Business Combination.

Segment Product Revenue, Operating Income and Operating Income as a Percentage of Revenue

Mobile Products

	Three Months Ended							
(In thousands, except percentages)	October 3, 2015	5	September 27, 2014		Increase	Percentage Change		
Revenue	\$ 578,160	\$	298,461	\$	279,699	93.7%		
Operating income	171,974		86,725		85,249	98.3		
Operating income as a % of revenue	29.7%		29.1%)				

	Six Months Ended									
(In thousands, except percentages)		October 3, 2015	5	September 27, 2014		Increase	Percentage Change			
Revenue	\$	1,129,047	\$	559,577	\$	569,470	101.8%			
Operating income		345,717		156,341		189,376	121.1			
Operating income as a % of revenue		30.6%		27.9%)					

The increase in MP revenue for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, was primarily due to the inclusion of TriQuint revenue in the three and six months ended October 3, 2015. The remaining increase is primarily due to increased demand for our cellular RF solutions for smartphones.

The increase in MP operating income for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, was primarily due to higher revenue and improved gross margin resulting from manufacturing- and sourcing-related cost reductions, which were partially offset by average selling price erosion.

Infrastructure and Defense Products

	Three Months Ended							
(In thousands, except percentages)	October 3, 2015	s	eptember 27, 2014		Increase	Percentage Change		
Revenue	\$ 129,205	\$	63,517	\$	65,688	103.4%		
Operating income	22,850		15,198		7,652	50.3		
Operating income as a % of revenue	17.7%		23.9%					

	Six Months Ended							
(In thousands, except percentages)	October 3, 2015		September 27, 2014		Increase	Percentage Change		
Revenue	\$ 250,989	\$	118,704	\$	132,285	111.4%		
Operating income	36,922		26,069		10,853	41.6		
Operating income as a % of revenue	14.7%		22.0%)				

Revenue increased for the three and six months ended October 3, 2015 as compared to the three and six months ended September 27, 2014, primarily due to the inclusion of TriQuint revenue in the three and six months ended October 3, 2015. IDP's revenue and operating income for the three and six months ended October 3, 2015 were adversely impacted by lower

global demand for wireless infrastructure products, which is currently expected to improve gradually beginning in the third quarter of fiscal 2016.

See Note 9 to the Condensed Consolidated Financial Statements for a reconciliation of segment operating income to the consolidated operating income for the three and six months ended October 3, 2015 and September 27, 2014.

OTHER (EXPENSE) INCOME AND INCOME TAXES

		Three Months Ended			Six Months Ended				
(In thousands)	00	tober 3, 2015		September 27, 2014		October 3, 2015		September 27, 2014	
Interest expense	\$	(660)	\$	(195)	\$	(1,208)	\$	(669)	
Interest income		472		40		864		75	
Other income		381		137		4,500		521	
Income tax expense		(13,784)		(11,927)		(14,376)		(19,345)	

Other Income

During the first quarter of fiscal 2016, we sold equity securities and recognized a gain of approximately \$4.0 million.

Income Taxes

Our provision for income taxes for the three and six months ended October 3, 2015 and September 27, 2014 has been calculated by applying an estimate of the annual effective tax rate for the full fiscal year to "ordinary" income or loss (pre-tax income or loss excluding unusual or infrequently occurring discrete items) for the three and six months ended October 3, 2015 and September 27, 2014.

Income tax expense was \$13.8 million and \$14.4 million for the three and six months ended October 3, 2015, which was comprised primarily of tax expense related to domestic and international operations generating pre-tax book income offset by a tax benefit related to international operations generating pre-tax book losses. Income tax expense was \$11.9 million and \$19.3 million for the three and six months ended September 27, 2014, which was comprised primarily of tax expense related to domestic and international operations offset by a tax benefit related to changes in the domestic deferred tax asset valuation allowance.

The valuation allowance against net deferred tax assets as of October 3, 2015 increased by \$2.4 million from the \$13.8 million balance as of the end of fiscal 2015, with the change primarily arising from an increase in domestic deferred tax assets related to state tax credits. A valuation allowance remained against certain domestic and foreign net deferred tax assets as it is more likely than not that the related deferred tax assets will not be realized.

LIQUIDITY AND CAPITAL RESOURCES

We have funded our operations to date through revenue from product sales, sales of equity and debt securities, bank borrowings and capital equipment leases. As of October 3, 2015, we had working capital of approximately \$920.5 million, including \$169.6 million in cash and cash equivalents (\$224.3 million was received from TriQuint in the Business Combination), compared to working capital of approximately \$453.7 million at September 27, 2014, including \$110.7 million in cash and cash equivalents. Working capital increased year-over-year primarily due to the Business Combination.

Our total cash, cash equivalents and short-term investments were \$195.6 million as of October 3, 2015. This balance includes approximately \$86.9 million held by our foreign subsidiaries. If these funds held by our foreign subsidiaries are needed for our operations in the U.S., we would be required to accrue and pay U.S. taxes to repatriate these funds. However, under our current plans, we expect to permanently reinvest these funds outside of the U.S. and do not expect to repatriate them to fund our U.S. operations.

Stock Repurchases

On February 5, 2015, the Company's Board of Directors authorized the repurchase of up to \$200.0 million of the Company's outstanding common stock, exclusive of related fees, commissions or other expenses. On August 11, 2015, the Company announced the completion of this \$200.0 million share repurchase program having repurchased approximately 2.4 million shares of the Company's common stock for approximately \$150.0 million during fiscal 2016. On August 11, 2015, the Company announced that its Board of Directors authorized a new share repurchase program to repurchase up to \$400.0 million of the Company's outstanding common stock. On September 10, 2015, the Company announced the completion of this \$400.0

million share repurchase program having repurchased approximately 7.3 million shares of common stock for approximately \$400.0 million during the three months ended October 3, 2015. These repurchases were partially funded through our revolving credit facility.

On November 5, 2015, the Company announced that its Board of Directors has authorized the repurchase of up to \$1.0 billion of the Company's outstanding common stock through November 4, 2016. Under the share repurchase program, share repurchases will be made in accordance with applicable securities laws on the open market or in privately negotiated transactions. The extent to which the Company repurchases its shares, the number of shares and the timing of any repurchases will depend on general market conditions, regulatory requirements, alternative investment opportunities and other considerations. The program does not require the Company to repurchase a minimum number of shares, and may be modified, suspended or terminated at any time without prior notice.

Cash Flows from Operating Activities

Operating activities for the six months ended October 3, 2015 generated cash of \$310.2 million, compared to \$95.0 million for the six months ended September 27, 2014. This year-over-year increase was primarily attributable to improved profitability exclusive of non-cash Business Combination expenses.

Cash Flows from Investing Activities

Net cash provided by investing activities for the six months ended October 3, 2015 was \$25.5 million, compared to net cash used in investing activities of \$73.4 million for the six months ended September 27, 2014. This change was primarily due to increased proceeds from maturities of available-for-sale securities for the six months ended October 3, 2015 as compared to the six months ended September 27, 2014, which was partially offset by higher capital expenditures for the six months ended October 3, 2015 as compared to the six months ended September 27, 2014.

Cash Flows from Financing Activities

Net cash used in financing activities was \$465.9 million for the six months ended October 3, 2015, compared to \$82.7 million for the six months ended September 27, 2014. Net cash used in financing activities was higher during the six months ended October 3, 2015 as we repurchased approximately 9.7 million shares of our common stock at an average price of \$56.81 on the open market. In addition, during the three months ended October 3, 2015, the Company borrowed \$125.0 million and made payments of \$50.0 million under its Credit Agreement, while during the six months ended September 27, 2014, the remaining principal balance of \$87.5 million plus interest of \$0.4 million was paid.

COMMITMENTS AND CONTINGENCIES

<u>Credit Agreement</u> On April 7, 2015, we and certain of our material domestic subsidiaries (the "Guarantors") entered into a five-year unsecured senior credit facility with Bank of America, N.A., as administrative agent (in such capacity, the "Administrative Agent"), swing line lender, and L/C issuer, and a syndicate of lenders (the "Credit Agreement"). The Credit Agreement includes a \$300.0 million revolving credit facility, which includes a \$25.0 million sublimit for the issuance of standby letters of credit and a \$10.0 million sublimit for swingline loans. We may request, at any time and from time to time, that the revolving credit facility be increased by an amount not to exceed \$150.0 million. The revolving credit facility is available to finance working capital, capital expenditures and other lawful corporate purposes. Our obligations under the Credit Agreement are jointly and severally guaranteed by the Guarantors. As of October 3, 2015, \$75.0 million was outstanding under the Credit Agreement.

The Credit Agreement contains various conditions, covenants and representations with which we must be in compliance in order to borrow funds and to avoid an event of default, including financial covenants that we must maintain a consolidated leverage ratio not to exceed 2.5 to 1.0 as of the end of any fiscal quarter and a consolidated interest coverage ratio not to be less than 3.0 to 1.0 as of the end of any fiscal quarter. At October 3, 2015, we were in full compliance with these covenants.

<u>Proposed Notes Offering</u> On November 6, 2015, the Company announced that it proposes to offer, subject to market conditions and other factors, \$1.0 billion aggregate principal amount of senior notes due 2023 and 2025 (the "Notes") to qualified institutional buyers pursuant to Rule 144A under the Securities Act of 1933, as amended (the "Securities Act"), and to certain non-U.S. persons in accordance with Regulation S under the Securities Act. The Company expects to use the net proceeds of the offering for general corporate purposes, including share repurchases and repayment of any amounts outstanding under its revolving credit facility. The Notes will be senior unsecured obligations of the Company and will be guaranteed, jointly and severally, by each of the Company's existing and future direct and indirect wholly-owned U.S. subsidiaries that guarantee the Company's obligations under its existing revolving credit facility.

The Notes will not be registered under the Securities Act or any state securities laws and may not be offered or sold in the United States absent registration or an applicable exemption from such registration requirements.

<u>Capital Commitments</u> At October 3, 2015, we had capital commitments of approximately \$56.2 million primarily related to projects for increasing manufacturing capacity. In November 2015, our Board of Directors authorized up to an additional \$128.0 million in capital expenditures in the current fiscal year to fund various manufacturing-related projects.

Future Sources of Funding Our future capital requirements may differ materially from those currently anticipated and will depend on many factors, including, but not limited to, market acceptance of our products, volume pricing concessions, capital improvements, demand for our products, technological advances and our relationships with suppliers and customers. Based on current and projected levels of cash flow from operations, coupled with our existing cash and cash equivalents and our revolving credit facility, we believe that we have sufficient liquidity to meet both our short-term and long-term cash requirements. However, if there is a significant decrease in demand for our products, or in the event that growth is faster than we had anticipated, operating cash flows may be insufficient to meet our needs. If existing resources and cash from operations are not sufficient to meet our future requirements or if we perceive conditions to be favorable, we may seek additional debt or equity financing. We cannot be sure that any additional equity or debt financing will not be dilutive to holders of our common stock. Further, we cannot be sure that additional equity or debt financing, if required, will be available on favorable terms, if at all.

<u>Legal</u> We are involved in litigation and other legal proceedings in the ordinary course of business that have not been fully adjudicated. These actions, when finally concluded and determined, will not, in the opinion of management, have a material adverse effect on our consolidated financial position or results of operations.

<u>*Taxes*</u> We are subject to income and other taxes in the United States and in numerous foreign jurisdictions. Our domestic and foreign tax liabilities are subject to the allocation of revenues and expenses in different jurisdictions. Additionally, the amount of taxes paid is subject to our interpretation of applicable tax laws in the jurisdictions in which we operate. We are subject to audits by tax authorities. While we endeavor to comply with all applicable tax laws, there can be no assurance that a governing tax authority will not have a different interpretation of the law than we do or that we will comply in all respects with applicable tax laws, which could result in additional taxes. There can be no assurance that the outcomes from tax audits will not have an adverse effect on our results of operations in the period during which the review is conducted.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

There have been no material changes to our market risk exposures during the second quarter of fiscal 2016. For a discussion of our exposure to market risk, refer to Item 7A, "Quantitative and Qualitative Disclosures About Market Risk," contained in Qorvo's Annual Report on Form 10-K for the fiscal year ended March 28, 2015.

ITEM 4. CONTROLS AND PROCEDURES.

As of the end of the period covered by this report, the Company's management, with the participation of the Company's Chief Executive Officer and the Chief Financial Officer, evaluated the effectiveness of the Company's disclosure controls and procedures in accordance with Rule 13a-15(e) under the Securities Exchange Act of 1934, as amended (the Exchange Act). Based upon their evaluation, the Chief Executive Officer and the Chief Financial Officer concluded that the Company's disclosure controls and procedures were effective for the purpose of ensuring that the information required to be disclosed in the reports that the Company files or submits under the Exchange Act with the Securities and Exchange Commission (the SEC) (i) is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and (ii) is accumulated and communicated to the Company's management, including its Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

On January 1, 2015, the Business Combination was consummated between RFMD and TriQuint. As a result of the Business Combination, the Company has incorporated internal controls over significant processes specific to TriQuint and the Business Combination that it believes to be appropriate and necessary in consideration of the level of related integration. As the Company further integrates the TriQuint business, it will continue to review the internal controls and may take further steps to ensure that the internal controls are effective and integrated appropriately.

Except for the paragraph above, no change in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) occurred during the quarter ended October 3, 2015 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

ITEM 1A. RISK FACTORS.

In addition to the other information set forth in this report and in our other reports and statements that we file with the SEC, including our quarterly reports on Form 10-Q, careful consideration should be given to the factors discussed in Part I, Item 1A., "Risk Factors" in Qorvo's Annual Report on Form 10-K for the fiscal year ended March 28, 2015, which could materially affect our business, financial condition or future results. The risks described in Qorvo's Annual Report on Form 10-K are not the only risks that we face. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may materially affect our business, financial condition and/or operating results.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS.

(c) Issuer Purchases of Equity Securities

Purchases of Equity Securities

Period	Total number of shares purchased	Average price paid per share	Total number of shares purchased as part of publicly announced plans or programs	Approximate dollar value of shares that may yet be purchased under the plans or programs
June 28, 2015 through August 1, 2015	_	_	_	\$100.0 million
August 2, 2015 through August 29, 2015	5,025	\$55.31	5,025	\$222.0 million
August 30, 2015 through October 3, 2015	4,051	\$54.77	4,051	_
Total	9,076	\$55.07	9,076	_

On February 5, 2015, we announced that our Board of Directors authorized the repurchase of up to \$200.0 million of our outstanding common stock, exclusive of related fees, commissions or other expenses. On August 11, 2015, we announced completion of this share repurchase program and also announced that our Board of Directors authorized a new share repurchase

program to repurchase up to \$400.0 million of our outstanding common stock. This repurchase program did not have an expiration date. As of October 3, 2015, we completed this \$400.0 million share repurchase program.

Subsequent to the quarter end, on November 5, 2015, the Company announced that its Board of Directors has authorized the repurchase of up to \$1.0 billion of the Company's outstanding common stock through November 4, 2016 (see Note 11 of the Notes to the Condensed Consolidated Financial Statements in Part I, Item 1 of this report for additional information).

Table of Contents

ITEM 6. EXHIBITS.

2.1	Contingent Acquisition Implementation Deed by and among TriQuint Semiconductor, Inc., Cavendish Kinetics Limited and Certain Cavendish Shareholders, dated as of August 4, 2015 *+
31.1	Certification of Periodic Report by Robert A. Bruggeworth, as Chief Executive Officer, pursuant to Rule 13a-14(a) or 15d-14(a) of the Exchange Act, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
31.2	Certification of Periodic Report by Steven J. Buhaly, as Chief Financial Officer, pursuant to Rule 13a-14(a) or 15d-14(a) of the Exchange Act, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
32.1	Certification of Periodic Report by Robert A. Bruggeworth, as Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Periodic Report by Steven J. Buhaly, as Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
101	The following materials from our Quarterly Report on Form 10-Q for the quarter ended October 3, 2015, formatted in XBRL (eXtensible Business Reporting Language): (i) the Condensed Consolidated Balance Sheets as of October 3, 2015 and March 28, 2015; (ii) the Condensed Consolidated Statements of Income for the three and six months ended October 3, 2015 and September 27, 2014; (iii) the Condensed Consolidated Statements of Comprehensive Income for the three and six months ended October 3, 2015 and September 27, 2014; (iv) the Condensed Consolidated Statements of Cash Flows for the six months ended October 3, 2015 and September 27, 2014; (iv) the Condensed Consolidated Statements of Cash Flows for the six months ended October 3, 2015 and September 27, 2014; and (v) the Notes to

* Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission as part of an application for confidential treatment pursuant to the Securities Exchange Act of 1934, as amended.

the Condensed Consolidated Financial Statements

+ Schedules and certain exhibits to the Contingent Acquisition Implementation Deed have been omitted pursuant to Item 601(b)(2) of Regulation S-K. Qorvo hereby undertakes to furnish supplementally copies of any of the omitted schedules and exhibits upon request by the Securities and Exchange Commission.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: November 12, 2015

Qorvo, Inc.

/s/ Steven J. Buhaly

Steven J. Buhaly Chief Financial Officer

EXHIBIT INDEX

- 2.1 Contingent Acquisition Implementation Deed by and among TriQuint Semiconductor, Inc., Cavendish Kinetics Limited and Certain Cavendish Shareholders, dated as of August 4, 2015 *+
- 31.1 Certification of Periodic Report by Robert A. Bruggeworth, as Chief Executive Officer, pursuant to Rule 13a-14(a) or 15d-14(a) of the Exchange Act, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 31.2 Certification of Periodic Report by Steven J. Buhaly, as Chief Financial Officer, pursuant to Rule 13a-14(a) or 15d-14(a) of the Exchange Act, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 32.1 Certification of Periodic Report by Robert A. Bruggeworth, as Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
- 32.2 Certification of Periodic Report by Steven J. Buhaly, as Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
- 101 The following materials from our Quarterly Report on Form 10-Q for the quarter ended October 3, 2015, formatted in XBRL (eXtensible Business Reporting Language): (i) the Condensed Consolidated Balance Sheets as of October 3, 2015 and March 28, 2015; (ii) the Condensed Consolidated Statements of Income for the three and six months ended October 3, 2015 and September 27, 2014; (iii) the Condensed Consolidated Statements of Comprehensive Income for the three and six months ended October 3, 2015 and September 27, 2014; (iii) the Condensed Consolidated Statements of Comprehensive Income for the three and six months ended October 3, 2015 and September 27, 2014; (iv) the Condensed Consolidated Statements of Cash Flows for the six months ended October 3, 2015 and September 27, 2014; and (v) the Notes to the Condensed Consolidated Financial Statements

* Portions of this Exhibit have been omitted and filed separately with the Securities and Exchange Commission as part of an application for confidential treatment pursuant to the Securities Exchange Act of 1934, as amended.

+ Schedules and certain exhibits to the Contingent Acquisition Implementation Deed have been omitted pursuant to Item 601(b)(2) of Regulation S-K. Qorvo hereby undertakes to furnish supplementally copies of any of the omitted schedules and exhibits upon request by the Securities and Exchange Commission.

Our SEC file number for documents filed with the SEC pursuant to the Securities Exchange Act of 1934, as amended, is 001-36801.

CERTAIN CONFIDENTIAL MATERIAL APPEARING IN THIS DOCUMENT, MARKED BY [*****], HAS BEEN OMITTED AND FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION PURSUANT TO RULE 24B-2 PROMULGATED UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED.

August 4, 2015

CONTINGENT ACQUISITION IMPLEMENTATION DEED

BY AND AMONG

TRIQUINT SEMICONDUCTOR, INC.,

CAVENDISH KINETICS LIMITED,

[*****]

[*****]

EACH OF THE COMMITTED SHAREHOLDERS,

EACH OF THE EXECUTIVE OPTIONHOLDERS,

THE RESTRICTED SHAREHOLDERS,

THE WARRANTORS

AND

SHAREHOLDER REPRESENTATIVE SERVICES LLC,

as Securityholder Agent

	Page
1. INTERPRETATION	<u>5</u>
2. CONDITIONS	<u>6</u>
3. ADHERENCE TO THIS DEED	<u>7</u>
4. THE OPTION	<u>8</u>
5. GOVERNANCE AND CONDUCT OF BUSINESS	<u>14</u>
6. DUE DILIGENCE DURING THE DILIGENCE PERIOD	<u>18</u>
7. THE OFFER	<u>18</u>
8. TREATMENT OF THE CONTINGENT SECURITIES	<u>26</u>
9. COMPLETION OF THE TRANSFER OF SHARES	<u>33</u>
10. SQUEEZE-OUT ACQUISITION	<u>36</u>
11. DRAG-ALONG ACQUISITION	<u>37</u>
12. WARRANTIES	<u>38</u>
13. REMEDIES AND LIMITATIONS ON LIABILITY	<u>39</u>
14. TAX COVENANT	<u>48</u>
15. NON-SOLICITATION	<u>48</u>
16. RECOMMENDATION BY THE COMPANY DIRECTORS	<u>49</u>
17. ACCESS TO INFORMATION AND ACCOUNTS	<u>51</u>
18. RESTRICTIONS ON DEALING WITH SHARES	<u>52</u>
19. PERMITTED TRANSFERS	<u>52</u>
20. CONSENT TO TRANSFER FOR THE PURPOSES OF THE ARTICLES	<u>54</u>
21. GENERAL UNDERTAKINGS	<u>54</u>
22. UNDERTAKINGS BY THE COMPANY	<u>56</u>
23. RESTRICTIVE COVENANTS FOR RESTRICTED SHAREHOLDERS	<u>56</u>
24. DISCLAIMER OF CORPORATE OPPORTUNITY	<u>58</u>
25. CONFLICT WITH ARTICLES; INVESTMENT AGREEMENT	<u>59</u>
26. PRE-EMPTION RIGHTS	<u>60</u>
27. TERMINATION OF THIS DEED	<u>60</u>
28. ASSIGNMENT	<u>61</u>
29. GENERAL WARRANTIES	<u>61</u>
30. FURTHER ASSURANCES; SECURITY POWER OF ATTORNEY	<u>62</u>
31. LIMITATION OF RELATIONSHIP	<u>64</u>
32. SET-OFF AND WITHHOLDINGS	<u>64</u>
33. NOTICES	<u>64</u>
34. REMEDIES AND WAIVERS	<u>66</u>
35. INVALIDITY AND SEVERANCE	<u>66</u>
36. LANGUAGE	<u>67</u>
37. ENTIRE AGREEMENT	<u>67</u>
38. AMENDMENT AND VARIATION	<u>67</u>

p,

CONTENTS (CONT'D)

	<u>Page</u>
39. CONFIDENTIALITY	<u>68</u>
40. ANNOUNCEMENTS	<u>70</u>
41. COSTS AND EXPENSES	<u>70</u>
42. COUNTERPARTS	<u>70</u>
43. CHOICE OF GOVERNING LAW	<u>70</u>
44. ARBITRATION	<u>70</u>
45. AGENT FOR SERVICE	<u>71</u>
46. SECURITYHOLDER AGENT	<u>72</u>
47. THIRD PARTY RIGHTS	<u>74</u>
Schedule 1 DEFINITIONS AND INTERPRETATION	<u>75</u>
Schedule 2 SCHEDULE OF SHAREHOLDERS	<u>93</u>
Schedule 3 SCHEDULE OF CONTINGENT SECURITIES	<u>94</u>
Schedule 4 OFFER DOCUMENT	<u>95</u>
Schedule 5 FORM OF ACCEPTANCE	<u>96</u>
Schedule 6 CONDUCT OF BUSINESS	<u>97</u>
Schedule 7 ACTIONS ON COMPLETION	<u>102</u>
Schedule 8 WARRANTIES	<u>106</u>
Schedule 9 WARRANTORS	<u>128</u>
Schedule 10 FORM OF DEED OF ADHERENCE	<u>129</u>
Schedule 11 EFFECTIVE TIME STATEMENTS	<u>130</u>
Schedule 12 ARTICLES OF ASSOCIATION	<u>137</u>
Schedule 13 TAX COVENANT	<u>138</u>
Schedule 14 ACCOUNTING POLICIES	<u>151</u>
Schedule 15 DETERMINATION OF LEGACY PRODUCT REVENUE	<u>155</u>
Schedule 16 PREPRODUCTION RELEASE OF THE SPECIFIED PRODUCT	<u>157</u>
Schedule 17 FORM OF DEED OF RELEASE AND WAIVER	<u>158</u>
Schedule 18 AMENDMENT #1 TO EMPLOYEE SHARE SCHEME GOVERNING	
LTD OPTIONS	<u>159</u>
Schedule 19 FORM OF ESCROW AGREEMENT	<u>160</u>
Schedule 20 FORM OF PAYMENTS AGREEMENT	<u>161</u>

DATE: August 4, 2015

PARTIES:

- 1. **TRIQUINT SEMICONDUCTOR, INC.**, a corporation incorporated in the State of Delaware, United States of America, doing business as "Qorvo," whose principal place of business is at 7628 Thorndike Road, Greensboro, North Carolina 27409-9421 ("**Qorvo**");
- 2. **CAVENDISH KINETICS LIMITED**, a private limited company incorporated in England and Wales with registered number 02982696 whose registered office is at 5 New Street Square, London, EC4A 3TW (the "**Company**");
- 3. [*****]
- 4. [*****]
- 5. [*****]
- 6. [*****]
- 7. EACH OF THE PERSONS LISTED IN <u>PART A</u> OF <u>SCHEDULE 2</u>, (each such person and any other person who becomes a Committed Shareholder pursuant to <u>clause 3.2(a)</u>, a "Committed Shareholder" and, collectively, the "Committed Shareholders");
- 8. EACH OF THE PERSONS LISTED IN <u>PART A</u> OF <u>SCHEDULE 3</u>, (each such person and any other person who becomes an Executive Optionholder pursuant to <u>clause 3.2(b)</u>, an "Executive Optionholder" and, collectively, the "Executive Optionholders");
- 9. EACH OF THE RESTRICTED SHAREHOLDERS. Each Restricted Shareholder is also an Executive Optionholder and each such person enters into this Deed in his or her capacity as both a Restricted Shareholder and an Executive Optionholder;
- 10. **EACH OF THE PERSONS LISTED IN <u>SCHEDULE 9</u>** (each a "Warrantor" and, collectively, the "Warrantors"). Each Warrantor is also a Major Investor, a Committed Shareholder or an Executive Optionholder and each such person enters into this Deed in their capacity as both a Major Investor, a Committed Shareholder or an Executive Optionholder (as the case may be) and as a Warrantor; and
- 11. SHAREHOLDER REPRESENTATIVE SERVICES LLC, a Colorado limited liability company solely in its capacity as the Securityholder Agent.

RECITALS:

(A) Each of the Major Investors, the Committed Shareholders and the Other Shareholders are "Shareholders" who hold Shares. Each of the Contingent Shareholders (including the Executive Optionholders) holds Contingent Securities which, subject to the terms of such Contingent Securities, may entitle such Contingent Shareholders to be issued Shares. Each of the Major Investors, the Committed Shareholders, the Other Shareholders and the Contingent Shareholders (including the Executive Optionholders) are the "Selling Shareholders" for the purposes of this Deed.

- (B) On or about the date of this Contingent Acquisition Implementation Deed (this "Deed"), Qorvo and the Company, inter alia, have entered into a subscription agreement relating to Series F Preferred Shares in the Company (the "Subscription Agreement"). Pursuant to the Subscription Agreement, the Company shall issue to Qorvo [*****] Series F Preferred Shares in the Company, to be issued by the Company pursuant to the Series F Financing. The execution and delivery of this Deed by the parties hereto is a condition precedent to the completion of the Series F Financing under the Subscription Agreement. On completion of the issuance of the Series F Preferred Shares in the Company, Qorvo shall also become a "Shareholder" for the purposes of this Deed. For the avoidance of doubt, Qorvo is not a Selling Shareholder under this Deed.
- (C) In further consideration for the investment in the Company to be made by Qorvo pursuant to the Subscription Agreement, each of the Major Investors, each Committed Shareholder and each Executive Optionholder has agreed to enter into this Deed to grant to Qorvo the exclusive option and right, but not the obligation, to acquire (pursuant to any of the methods specified in this Deed) the entire issued share capital of the Company (comprising the Shares and the Contingent Securities (other than any Shares held by Qorvo or any of its Affiliates)) on the terms and conditions of this Deed.
- (D) In the event that Qorvo exercises the exclusive option and right to acquire the entire issued share capital of the Company (comprising the Shares and the Contingent Securities (other than any Shares held by Qorvo or any of its Affiliates)), then: (i) Qorvo shall be entitled to make an offer to the Committed Shareholders, the Major Investors and the Other Shareholders and any other person who holds Shares on the date of the Offer to acquire one hundred per cent (100%) of the Shares (other than any Shares held by Qorvo or any of its Affiliates); and (ii) Qorvo shall be entitled to require that the Company shall procure that the Contingent Securities shall be exchanged for either: (a) in the case of Contingent Securities which have vested or which are otherwise exercisable prior to Completion of the Acquisition – cash consideration paid to the Contingent Shareholders by Qorvo in accordance with this Deed; or (b) in the case of Contingent Securities which have not vested or which are not otherwise exercisable on or prior to Completion of the Acquisition – securities issued by Qorvo in accordance with clause 8; or (c) in the case of Contingent Securities which are exercised on or after the date of this Deed – Ordinary Shares or Preferred Shares, in which case such Ordinary Shares or Preferred Shares shall be made on the terms

and conditions of this Deed, the Offer Document and the Form of Acceptance and the exchange of the Contingent Securities will be made on the terms and conditions of this Deed and the Employee Share Schemes.

- (E) The initial parties to this Deed are Qorvo, the Company, the Major Investors, the Committed Shareholders, the Executive Optionholders and the Securityholder Agent. The provisions of <u>clause 3</u> provide for the accession to this Deed of other persons who may hold Shares or Contingent Securities following the date hereof upon execution and delivery by any such person of a Deed of Adherence to the Company and Qorvo (or a person nominated by Qorvo), whereupon such person will become a party to this Deed as a Committed Shareholder or an Executive Optionholder (as specified in the Deed of Adherence) and (as the case may be) a Warrantor.
- (F) The Major Investors and the Committed Shareholders hereby undertake to accept the Offer if it is made by Qorvo, by executing and delivering a Form of Acceptance to Qorvo and the Company, in accordance with the terms and conditions of this Deed. In the event that any Executive Optionholder is issued any Ordinary Shares prior to Completion of the Acquisition in any circumstances, such Executive Optionholder hereby also undertakes to accept the Offer if it is made by Qorvo in respect of such Ordinary Shares, by executing and delivering a Form of Acceptance to Qorvo and the Company, in accordance with the terms and conditions of this Deed.
- (G) If any Other Shareholder or any person who holds or is issued Shares on or after the date of this Deed does not accept the Offer by executing and delivering a Form of Acceptance to Qorvo and the Company during the Offer Period, Qorvo shall complete the acquisition of the Shares held by such persons via a Squeeze-Out Acquisition or a Drag-Along Acquisition in accordance with the terms and conditions of this Deed.
- (H) If Qorvo makes the Offer and Completion of the Acquisition occurs in accordance with the terms and conditions of this Deed, the Company shall procure (and each Executive Optionholder hereby undertakes to permit the Company to procure in respect of the Contingent Securities held by such Executive Optionholder) that on Completion of the Acquisition all Contingent Securities then outstanding in respect of any Ordinary Shares or Preferred Shares shall either: (i) convert into Ordinary Shares or Preferred Shares to be acquired via a Squeeze-Out Acquisition or a Drag-Along Acquisition; or (ii) be exchanged for either: (a) in the case of Contingent Securities which have vested or which are otherwise exercisable prior to Completion of the Acquisition cash consideration paid to the Contingent Shareholders by Qorvo in accordance with this Deed; (b) in the case of Contingent Securities which have not vested or which are not otherwise exercisable on or prior to Completion of the Acquisition equivalent securities issued by Qorvo of

equivalent value to the cash consideration due in respect of such Contingent Securities, in each case in accordance with this Deed and the Employee Share Schemes.

(I) The purpose of this Deed is to set out the terms and conditions of the Option to be granted to Qorvo, the terms and conditions of the governance and operations of the Company during the Option Period, the mechanics for exercise of the Option and the making of the Offer, the treatment of any Contingent Securities and certain other matters in relation to the implementation and Completion of the Acquisition.

IN WITNESS OF WHICH, THE PARTIES HAVE EXECUTED AND DELIVERED THIS DOCUMENT AS A DEED TO RECORD AS FOLLOWS:

1. INTERPRETATION

- 1.1 In this Deed, unless otherwise specified:
 - (a) capitalised terms identified in <u>Schedule 1</u> and applied in this Deed have the meanings ascribed to them in <u>Schedule 1</u>;
 - (b) references to clauses and paragraphs of, and Schedules to, are to clauses and paragraphs of, and Schedules to, this Deed;
 - (c) references to a "**company**" shall be construed so as to include any corporation or other body corporate, wherever and however incorporated or established;
 - (d) references to a "**person**" shall be construed so as to include any individual, firm, company, corporation, body corporate, government, state or agency of a state, local or municipal authority or government body or any joint venture, association or partnership (whether or not having separate legal personality);
 - (e) "**body corporate**" shall have the meaning given in section 1173 of the Companies Act and "**subsidiary**" and "**wholly-owned subsidiary**" shall have the meaning given in section 1159 of the Companies Act;
 - (f) any reference to a "**day**" (including within the phrase "**Business Day**") shall mean a period of twenty-four (24) hours from midnight to midnight;
 - (g) references to times are to time in New York, New York;
 - (h) references to "**indemnify**" any person against any circumstance shall include indemnifying and keeping him harmless from all actions, claims and proceedings from time to time made against him and all Losses suffered, made or incurred by him as a consequence of that circumstance; and
 - (i) a reference to any other document referred to in this Deed is a reference to that other document as amended, varied, novated or supplemented at any time.
- 1.2 All headings and titles are inserted for convenience only. They are to be ignored in the interpretation of this Deed.
- 1.3 The provisions of the Schedules to this Deed shall form part of and be incorporated into this Deed.
- 1.4 Any reference to "**writing**" or "**written**" means any method of reproducing words in a legible and non-transitory form (including, for the avoidance of doubt, e-mail).
- 1.5 References to "include" or "including" are to be construed without limitation.

- 1.6 Unless the context otherwise requires, words in the singular include the plural and vice versa and a reference to any gender includes all other genders.
- 1.7 References to any statute or statutory provision include a reference to that statute or statutory provision as amended, consolidated or replaced from time to time (whether before or after the date of this Deed) and include any subordinate legislation made under the relevant statute or statutory provision.
- 1.8 References to any English legal term for any action, remedy, method of financial proceedings, legal document, legal status, court, official or any legal concept or thing shall, in respect of any jurisdiction other than England, be deemed to include what most nearly approximates in that jurisdiction to the English legal term.
- 1.9 The rule known as the *ejusdem generis* rule shall not apply to this Deed and accordingly general words introduced by the word "other" shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things.
- 1.10 General words in this Deed shall not be given a restrictive meaning by reason of the fact that they are followed by particular examples intended to be embraced by the general words.
- 1.11 References to "USD" or "\$" are to the lawful currency of the United States of America. References to "GBP" or "£" are to the lawful currency of the United Kingdom of Great Britain and Northern Ireland. References to "EUR" or "€" are to the lawful currency of the European Monetary Union.
- 1.12 Other than as explicitly provided in this Deed, the obligations of each of the Major Investors and each Committed Shareholder under this Deed are several and any obligation expressed to be an obligation of an Major Investor or any Committed Shareholder is to be read as the sole obligation of each Major Investor or Committed Shareholder (as the case may be), and no Major Investor or Committed Shareholder shall be excused from compliance with this Deed by the action or omission of any other Major Investor or Committed Shareholder.
- 1.13 All financial and accounting calculations used in or made pursuant to this Deed shall be made in United States dollars. Any amounts that are denominated in a currency other than United States dollars shall be converted into United States dollars. Such conversion shall be based upon the average daily exchange rate for the month in which any item was recorded, and the exchange rate to be used with respect to any conversion shall be a rate provided by a reputable Third Party exchange rate service that Qorvo regularly uses for purposes of determining exchange rates.
- 1.14 References to the use of "reasonable endeavours" by the Major Investors, Committed Shareholders and Executive Optionholders shall not require such parties to make payments of any amounts to any other Shareholder or Contingent Shareholder, nor require the joinder of such party in any lawsuit.

2. CONDITIONS

- 2.1 This Deed (except for the provisions referred to in <u>clause 2.2</u>) is conditional on completion occurring under the Subscription Agreement (the "**Condition**").
- 2.2 The provisions of this <u>clause 2</u>, <u>clause 1</u>, <u>clause 33</u>, <u>clause 35</u>, <u>clause 36</u>, <u>clause 37</u>, <u>clause 38</u>, <u>clause 39</u>, <u>clause 40</u>, <u>clause 41</u>, <u>clause 42</u>, <u>clause 43</u>, <u>clause 44</u>, <u>clause 45</u> and <u>clause 46</u> shall take effect on the date of this Deed and shall be effective even if the Condition is not satisfied.
- 2.3 The Condition contained in <u>clause 2.1</u> may only be waived in whole or in part by the parties hereto upon the written agreement of the Company, Qorvo and each of the Major Investors.
- 2.4 The remaining clauses of this Deed take effect on the date on which the Condition in <u>clause 2.1</u> is satisfied or waived.

3. ADHERENCE TO THIS DEED

- 3.1 The rights and obligations set out in this Deed will arise:
 - (a) in respect of a Committed Shareholder, Executive Optionholder or Major Investor who is a party to this Deed on the date of this Deed on the execution and delivery to Qorvo and the Company of this Deed;
 - (b) in respect of any Other Shareholder on the execution and delivery to Qorvo and the Company of a Deed of Adherence to this Deed, which will constitute the accession of that person to this Deed as if it were named in it as a Committed Shareholder but not (for the avoidance of doubt) as a Warrantor;
 - (c) in respect of any Contingent Shareholder (other than an Executive Optionholder) on the execution and delivery to Qorvo and the Company of a Deed of Adherence to this Deed, which will constitute the accession of that person to this Deed as if it were named in it as an Executive Optionholder but not (for the avoidance of doubt) as a Warrantor;
 - (d) in respect of any Contingent Shareholder (other than an Executive Optionholder) who exercises any Contingent Securities on or after the date of this Deed and who is issued any Ordinary Shares or Preferred Shares – on the execution and delivery to Qorvo and the Company of a Deed of Adherence to this Deed, which will constitute the accession of that person to this Deed as if it were named in it as a Committed Shareholder in respect of such Ordinary Shares or Preferred Shares, but not (for the avoidance of doubt) as a Warrantor.
- 3.2 The Company shall procure that:
 - (a) any person who is issued or acquires any Shares on and from the date of this Deed and who is not at such time of issuance or acquisition a party to this Deed (either as a signatory hereto or by means of execution of a Deed of Adherence) shall, as a condition of being issued or acquiring such Shares, execute a Deed of Adherence to this Deed, which will constitute the accession of that person to this Deed as if it were named in it as a Committed Shareholder;

- (b) any person who is issued or acquires any Contingent Securities on and from the date of this Deed (whether under an Employee Share Scheme or otherwise) and who, at such time of issuance or acquisition, is not a party to this Deed (either as a signatory hereto or by means of execution of a Deed of Adherence) shall, as a condition of being issued or acquiring such Contingent Securities, execute a Deed of Adherence to this Deed in respect of such Contingent Securities, which will constitute the accession of that person to this Deed as if it were named in it as an Executive Optionholder and, if that person shall, after the issuance or acquisition of such Contingent Securities, hold, in aggregate (including the Contingent Securities issued on and from the date of this Deed and any other Contingent Securities held by such person prior to the date of this Deed), Contingent Securities granting rights to subscribe for in excess of 250,000 Ordinary Shares, such person shall also be required by such Deed of Adherence to accede to this Deed as a Warrantor as a condition of being issued or acquiring such further Contingent Securities.
- 3.3 If any Executive Optionholder exercises any Contingent Securities on or after the date of this Deed, that Executive Optionholder shall automatically and immediately, without the need for any further action or the execution of any additional document, assume the rights and obligations of a Committed Shareholder under this Deed with respect to any Ordinary Shares issued to such Executive Optionholder on the exercise of such Contingent Securities. The rights and obligations of the Executive Optionholder with respect to any remaining Contingent Securities held or acquired by such Executive Optionholder shall continue in full force and effect.
- 3.4 On the execution of a Deed of Adherence to this Deed by any person, the Company shall update <u>Schedule 2</u> or <u>Schedule 3</u> (as the case may be) and shall deliver to Qorvo such updated <u>Schedule 2</u> or <u>Schedule 3</u>.

4. THE OPTION

4.1 Each of the Major Investors, the Executive Optionholders, the Company and each Committed Shareholder hereby irrevocably grants to Qorvo the exclusive right and option, but not the obligation (the "**Option**"), to: (i) make an offer (either directly or through a wholly-owned subsidiary of Qorvo) (the "**Offer**") to the Committed Shareholders, the Major Investors, the Other Shareholders and any other person who holds Shares on the date of the Offer to acquire one hundred per cent (100%) of the issued share capital of the Company on the date of the Offer (other than any Shares held by Qorvo or any of its Affiliates); and (ii) require that the Company shall procure (and each Executive Optionholder shall undertake to permit the Company to procure in respect of the Contingent Securities then outstanding in respect of any Ordinary Shares or Preferred Shares shall either: (i) convert into Ordinary Shares or Preferred Shares to be acquired via a Squeeze-Out Acquisition or a Drag-Along Acquisition; or (ii) be exchanged for either: (a) in the case of Contingent Securities which have vested or which are otherwise exercisable prior to Completion of the Acquisition paid to the Contingent Shareholders by Qorvo in accordance with this Deed; (b) in the case of Contingent Securities which have not vested or which are not otherwise exercisable prior to

Completion of the Acquisition – securities issued by Qorvo in accordance with <u>clause 8</u> (such transactions, the "Acquisition"), on the following terms and conditions:

- (a) subject to <u>clause 4.2</u>, the Option may be exercised at any time on and from the date of this Deed until and including the earlier of:
 - (i) the date which falls sixty (60) days after Preproduction Release of the Specified Product;
 - (ii) 30 June 2017; and
 - (iii) the termination of this Deed,

(such period, the "**Initial Option Period**"). The Initial Option Period and, if applicable, the Extended Option Period (as defined in <u>clause 4.2</u>) shall constitute the "**Option Period**" for the purposes of this Deed). The Company shall deliver a notice to Qorvo within [*****] Business Days following the date of the Preproduction Release of the Specified Product, notifying Qorvo of the occurrence of the Preproduction Release of the Specified Product, and the date of such occurrence. For purposes of clarification, whether or not the Preproduction Release of the Specified Product has occurred is subject to, among other things, the Qorvo [*****] set forth in <u>Schedule 16</u>.

- (b) the Option may be exercised once only during the Option Period and may only be exercised in respect of an offer for all (and not some only) of the issued and to be issued share capital of the Company (other than any Shares held by Qorvo or any of its Affiliates);
- (c) Qorvo may exercise the Option by serving a written notice on the Company (with a copy to the Securityholder Agent) at any time during the Option Period stating that Qorvo wishes to exercise the Option in accordance with the terms of this Deed (the "**Option Notice**"). For the avoidance of doubt, service of the Option Notice shall not oblige Qorvo to make the Offer or complete the Acquisition;
- (d) as soon as practicable upon exercise of the Option by Qorvo and receipt by the Company of the Option Notice sent pursuant to <u>clause 4.1(c)</u>, the Company shall notify the Major Investors, the Executive Optionholders and the Committed Shareholders of receipt of the Option Notice from Qorvo;
- (e) subject to <u>clause 4.6</u>, following receipt of the Option Notice by the Company until the end of the Diligence Period (as defined below), the Company shall render all reasonable assistance to Qorvo to enable Qorvo to conduct a due diligence and disclosure exercise on the business and financial affairs of the Company in accordance with <u>clause 6</u>. At any time prior to the date which is [*****] Business Days prior to the end of the Diligence Period, the Company may (on behalf of the Warrantors) deliver to Qorvo an update to the Disclosure Letter in accordance with <u>clause 12</u> and <u>clause 13</u>. In the event that the Company (on behalf of the Warrantors) does not deliver an update to the Disclosure

Letter to Qorvo prior to the date which is [*****] Business Days prior to the end of the Diligence Period, the Company (and the Warrantors) shall be prohibited from making any further disclosures against the Warranties. Any disclosures sought to be made on or after the date which is [*****] Business Days prior to the end of the Diligence Period shall in no way limit or qualify the Warranties. Only those matters set out in any Disclosure Letter delivered to Qorvo prior to the date which is [*****] Business Days prior to the end of the Diligence Period shall qualify the Warranties given as at Completion of the Acquisition. "**Diligence Period**" shall mean a period of [*****] Business Days following the date on which all of the following have occurred: (i) Qorvo has delivered the Option Notice to the Company; (ii) the Company has populated the Data Room with any and all diligence materials reasonably requested by Qorvo (including any items listed on any update to the Disclosure Letter) to complete a due diligence and disclosure exercise on the business and financial affairs of the Company in accordance with <u>clause 6</u>; and (iii) the Data Room is made available to Qorvo and its advisers;

- (f) Qorvo shall serve a written notice on the Company at any time on and from the commencement of the Diligence Period to and including the date which is [*****] Business Days after the conclusion of the Diligence Period which either (i) confirms that Qorvo intends to proceed with the Offer and the Acquisition (subject to the further terms and conditions set out in this Deed) (the "Offer Confirmation Notice"), or (ii) states that Qorvo has determined not to proceed with the Offer and which rescinds the Option Notice, in which case the Option shall immediately cease to be exercisable and shall immediately lapse, and this Deed, and the rights and obligations of the parties to this Deed hereunder, shall terminate as provided in <u>clause 27</u>. For the sake of clarity, if, at 12:01 a.m. on the date which is [*****] Business Days after the conclusion of the Diligence Period, Qorvo has not served the written notice required by this <u>clause 4.1(f)</u>, then the Option shall immediately cease to be exercisable and shall immediately lapse, and the parties to this Deed hereunder, shall terminate as provided in clause of the parties to the exercisable and shall immediately lapse, and the zero hereunder, shall terminate as provided in <u>clause 27</u>. For the sake of clarity, if, at 12:01 a.m. on the date which is [*****] Business Days after the conclusion of the Diligence Period, Qorvo has not served the written notice required by this <u>clause 4.1(f)</u>, then the Option shall immediately cease to be exercisable and shall immediately lapse, and this Deed, and the rights and obligations of the parties to this Deed hereunder, shall terminate as provided in <u>clause 27</u>;
- (g) if the Company timely receives the Offer Confirmation Notice from Qorvo, then within [*****] Business Days of receipt of such notice:
 - (i) Qorvo shall distribute the Offer Document as prepared and approved by Qorvo (acting reasonably), in substantially the form set out in <u>Schedule 4</u> and on the requisite terms and conditions specified in this Deed and in particular <u>clause 7</u>, to each of the Major Investors, the Committed Shareholders and the Other Shareholders. The Company and the Major Investors shall, and the Committed Shareholders shall use reasonable endeavours to, render to Qorvo all assistance reasonably necessary to enable Qorvo to prepare the Offer Document and to comply with any and all Applicable Laws with respect to the content, distribution and registration of the Offer Document, including by obtaining for the Directors advice from a competent, independent person who has no substantial financial interest in the Company or Qorvo on the financial implications of the Offer, and by furnishing the consolidated audited accounts of the Company in respect of the

latest accounting reference period for which the period for laying and delivering the accounts under the Companies Act has passed or, if accounts in respect of a later accounting reference period have been delivered under the relevant legislation, as shown in those accounts and not the earlier accounts, the letter from the Company to be included in the Offer Document and all other information and documentation required for the Offer Document, and authorizing and approving the information relating to the Company and its Subsidiaries contained in the Offer Document;

- (ii) the Company, through all of its Directors (and each of the Major Investors and the Committed Shareholders, to the extent that they have appointed, or have a right to appoint, any Director), shall procure that each Director shall recommend the Offer as required in accordance with <u>clause 16</u>; and
- (iii) each of the Major Investors and the Committed Shareholders shall, and hereby irrevocably undertakes to Qorvo that it shall, upon receipt of the Offer Document, complete and return a valid Form of Acceptance with respect to all Shares then held by such Major Investor or Committed Shareholder, in accordance with and subject to the terms set out in the Offer Document promptly upon the Offer being made by Qorvo and the Offer Document being distributed. Each Major Investor and Committed Shareholder irrevocably acknowledges and agrees that, if such Major Investor or Committed Shareholder defaults on its obligation to promptly complete and return a valid Form of Acceptance, such Form of Acceptance shall be executed and returned by a Qorvo Officer on behalf of such defaulting Major Investor or Committed Shareholder in accordance with <u>clause 30</u> in favour of any Qorvo Officer to secure the performance of its obligations under this Deed;
- (iv) in the event that any Executive Optionholder has exercised any Contingent Securities and been issued any Ordinary Shares prior to the making of the Offer, the Executive Optionholders hereby irrevocably undertake to Qorvo that they shall, upon receipt of the Offer Document, complete and return a valid Form of Acceptance with respect to all Shares then held by such Executive Optionholder, in accordance with and subject to the terms set out in the Offer Document promptly upon the Offer being made by Qorvo and the Offer Document being distributed. Each Executive Optionholder irrevocably acknowledges and agrees that, if such Executive Optionholder defaults on its obligation to promptly complete and return a valid Form of Acceptance, such Form of Acceptance shall be executed and returned by a Qorvo Officer on behalf of such defaulting Executive Optionholder pursuant to the security power of attorney granted by each Executive Optionholder in accordance with <u>clause 30</u> in favour of any Qorvo Officer to secure the performance of its obligations under this Deed;

- (v) the Administrators of the Employee Share Schemes shall send a notice to each Contingent Shareholder (including the Executive Optionholders) notifying them of the Offer and setting out the proposed treatment of their Contingent Securities upon Completion of the Acquisition in accordance with <u>clause 8</u>; and
- (vi) the Company shall send a notice to [*****] notifying [*****] of the Offer, confirming that Qorvo shall not assume the [****] Warrants and requesting that [****] exercise the [****] Warrants within twenty (20) calendar days from the date of such notice. Upon exercise, the [****] Warrants shall be dealt with upon Completion of the Acquisition in accordance with clause 8;
- (h) the Offer shall be open to be accepted by the Major Investors and the Committed Shareholders, as well as each Other Shareholder and any other person who holds Shares on the date of the Offer (including any Executive Optionholder, if applicable), for a period of at least [*****] Business Days from the date on which the Offer Document is distributed to such persons, or such longer period as Qorvo may determine in its reasonable discretion (subject to Applicable Law), but which shall not exceed [*****] days (subject to any extension pursuant to clause 9.10(a)) (the "Offer Period"). During the Offer Period, the Company shall render all reasonably necessary assistance to Qorvo in soliciting Forms of Acceptance from any of the Other Shareholders and any other person who holds Shares on the date of the Offer;
- (i) at the conclusion of the Offer Period, each of the Major Investors and the Committed Shareholders (and, if applicable, any Executive Optionholder) undertakes to sell, and Qorvo shall purchase, all Shares held by such Major Investor or Committed Shareholder (or, if applicable, Executive Optionholder) with all rights attached or accruing to them at Completion of the Acquisition in accordance with the provisions of this Deed and the Offer Document. Each Major Investor and Committed Shareholder (and, if applicable, Executive Optionholder) irrevocably acknowledges and agrees that, if such Major Investor or Committed Shareholder (or, if applicable, Executive Optionholder) defaults on its obligation to transfer all the Shares held by such Major Investor or Committed Shareholder (or, if applicable, Executive Optionholder) defaulting Major Investor or Committed Shareholder (or, if applicable, Executive Optionholder) pursuant to the security power of attorney granted by each Major Investor and Committed Shareholder (or, if applicable, Executive Optionholder) pursuant to the security power of attorney granted by each Major Investor and Committed Shareholder (and each Executive Optionholder) in accordance with <u>clause 30</u> in favour of any Qorvo Officer to secure the performance of its obligations under this Deed;
- (j) at the conclusion of the Offer Period and on Completion of the Acquisition, the Company and Qorvo shall procure that all Contingent Securities (including the Contingent Securities held by the Executive Optionholders) are exchanged for either cash or securities issued by Qorvo in accordance with <u>clause 8</u>;
- (k) at the conclusion of the Offer Period, each Other Shareholder that has delivered a Form of Acceptance to Qorvo shall sell, and Qorvo shall purchase, all Shares held by such

Other Shareholder with all rights attached or accruing to them at Completion of the Acquisition in accordance with the provisions of this Deed and the Offer Document;

- the transfer of any Shares held by any Other Shareholder who has not delivered a Form of Acceptance to Qorvo and the Company during the Offer Period shall be completed in accordance with <u>clause 7.12</u>, via either a Squeeze-Out Acquisition or a Drag-Along Acquisition; and
- (m) throughout the Option Period and until Completion of the Acquisition in accordance with the Offer Document and this Deed, the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall comply with any and all obligations imposed on them and undertakings given by them in accordance with the terms of this Deed.
- 4.2 Subject to <u>clause 4.3</u>, in the event that the Preproduction Release of the Specified Product has not occurred prior to the 30 June 2017, Qorvo shall have the exclusive right and option, but not the obligation, to extend the Initial Option Period until and including the earlier of:
 - (a) the date which falls sixty (60) days after Preproduction Release of the Specified Product; and
 - (b) 30 June 2018,

(the "Extended Option Period"), by written notice to the Company and the Major Investors (with a copy to the Securityholder Agent) received by the Company and the Major Investors on or before [*****] (the "Additional Funding Notice") provided that in the event that Qorvo exercises its right to extend the Initial Option Period pursuant to this clause 4.2, the Company may request that Qorvo invest in the Company an additional amount as determined by the Company at such time, which amount shall be no more than USD \$25 million, in consideration for the issue by the Company to Qorvo of an additional amount of the Series F Preferred Shares in the Company, on equivalent terms and conditions as set out in the Subscription Agreement, *mutatis mutandis* and with all warranties repeated as at the date of issuance of such additional Series F Preferred Shares (the "Additional Funding"). For the sake of clarity, upon the Company's request for the Additional Funding, which shall be delivered by way of written notice to Qorvo within [*****] days of the Company's receipt of the Additional Funding Notice and shall state the additional investment amount desired by the Company, Qorvo shall be obliged to make the additional investment. The Additional Funding shall be completed on or before [*****]; provided, however, that such date may be extended for a reasonable period of time in order to accommodate the need to satisfy any regulatory obligations. In the event that (i) the Company requests Additional Funding, (ii) the Company complies with its obligations pursuant to and satisfies its closing conditions under the Subscription Agreement with respect to the Additional Funding and (iii) Qorvo fails to wire the additional funds to the Company prior to [*****] (as such date may be extended in accordance with the previous sentence) in accordance with the terms of this <u>clause 4.2</u>, then the Option shall immediately cease to be exercisable and shall immediately lapse, and this Deed, and the rights and obligations of the parties to this Deed hereunder, shall terminate as provided in clause 27.

- 4.4 If the Option is not duly exercised during the Option Period, the Option will cease to be exercisable and shall lapse, and this Deed, and the rights and obligations of the parties to this Deed hereunder, shall terminate as provided in <u>clause 27</u>. For purposes of clarification, if the Option is exercised during the Option Period, the Completion of the Acquisition may take place after the end of the Option Period.
- 4.5 The Company undertakes to Qorvo that it shall not (and each of the Major Investors, the Committed Shareholders and the Executive Optionholders undertakes to Qorvo that they shall take all appropriate action within its control or authority (taking into account its position as a shareholder or optionholder of the Company) to procure that the Company shall not) issue any Series F Preferred Shares to any person other than in accordance with the terms and provisions provided in the Subscription Agreement, or in accordance with the Additional Funding or <u>clause 4.3</u> of this Deed.
- 4.6 In the event that, in its reasonable judgment, Qorvo determines that the Company has not fully and fairly complied with its obligations under <u>clause 4.1(e)</u> or <u>clause 6</u> during the Diligence Period, or does not provide Qorvo with timely access to all information requested by Qorvo, Qorvo shall have the one-time right to extend the Diligence Period by a period of up to [*****] additional Business Days from the end of the initial Diligence Period, which it may elect to do upon written notice to the Company, identifying with specificity the diligence items Qorvo is requesting, delivered prior to the end of Working Hours on the last day of the initial Diligence Period. For purposes of clarification, if the Company has not fully and fairly complied in all material respects with its obligation under <u>clause 4.1(e)</u> or <u>clause 6</u>, during the Diligence Period, it shall be considered a material breach of this Deed in any Action brought by Qorvo following termination of this Deed.

5. GOVERNANCE AND CONDUCT OF BUSINESS

- 5.1 The Company hereby undertakes that on and from the date of this Deed until the earlier of (i) the termination of this Deed and (ii) the Completion of the Acquisition:
 - (a) each member of the Group shall carry on business in the normal and ordinary course. In particular, but without limitation and in compliance with antitrust Applicable Laws, acts or matters listed in <u>Part A</u> of <u>Schedule 6</u> shall require the consent in writing of Qorvo prior to the exercise of the Option by Qorvo and acts or matters listed in <u>Part B</u> of

<u>Schedule 6</u> shall require the consent in writing of Qorvo following the exercise of the Option by Qorvo; provided, that if a member of the Group determines, with the advice of counsel, that obtaining such consent would not be in compliance with antitrust Applicable Law, the Company and such member of the Group shall, within a reasonable time before taking, or permitting any other member of the Group to take, any action which would otherwise require the consent of Qorvo pursuant to this <u>clause 5.1(a)</u>, consult with Qorvo regarding such action and the parties shall use their reasonable endeavours to ensure that the intent of the parties in executing this Deed is preserved;

- (b) each member of the Group shall:
 - (i) maintain its assets and properties in the ordinary course of business in the manner historically maintained, reasonable wear and tear, damage by fire and other casualty excepted;
 - (ii) promptly repair, restore or replace all material assets and properties in the ordinary course of business consistent with past practice;
 - (iii) upon any damage, destruction or loss to any of its material assets or properties, apply any and all insurance proceeds, if any, received with respect thereto to the prompt repair, replacement and restoration thereof;
 - (iv) comply with all Applicable Laws;
 - (v) take all actions necessary to be in compliance in all material respects with all Material Contracts and to maintain the effectiveness of all Company Permits;
 - (vi) notify Qorvo in writing of the commencement, or any written threat of the commencement, of any action, suit, claim, investigation or other like proceeding against any member of the Group; and
 - (vii) pay accounts payable and pursue collection of its accounts receivable in the ordinary course of business, consistent with past practices;
- 5.2 The Company and Qorvo hereby undertake and agree that on and from the date of this Deed until the earlier to occur of (i) the termination of this Deed and (ii) the Completion of the Acquisition:
 - (a) Unless otherwise agreed in advance by Qorvo, an authorised Representative of Qorvo shall be present for any inperson meetings or scheduled telephone or conference calls between any member of the Group (or any employee, officer, director, adviser or consultant of any member of the Group) and [*****] with respect to the [*****]. In addition, unless otherwise agreed in advance by the Company, an authorised Representative of the Company shall be present for any in-person meetings or scheduled telephone or conference calls between Qorvo, Parent and any Affiliate of Qorvo or Parent (or any employee, officer, director, adviser or consultant of Qorvo, Parent or an affiliate of Qorvo or Parent) and [*****] with respect to the [*****]. In addition, the Company

and Qorvo shall provide to each other copies of any and all written communications between [*****] and each such party regarding the [*****]. Notwithstanding the foregoing, the above rights and obligations of the parties hereto shall be subject to [*****] prior approval and consent in each instance, such consent which may be expressly given in the [*****]; and

- (b) the Company shall hold [*****] with Representatives of Qorvo to ensure that Qorvo is kept aware of [*****], provided that during such meetings the Company shall not be required to share its [*****] with Qorvo.
- 5.3 Between the execution of this Deed and Completion of the Acquisition, (a) the Company shall each notify Qorvo promptly if they become aware of a fact or circumstance which constitutes or which would or might reasonably constitute a breach (whether material, repudiatory or remedial in nature or not) of the requirements of this <u>clause 5</u> and (b) the Major Investors, the Executive Optionholders and each Committed Shareholder shall each notify Qorvo promptly if they know that a breach of this <u>clause 5</u> has occurred.
- 5.4 Following the delivery of the Offer Confirmation Notice and so long as the Option has not been rescinded by Qorvo in accordance with the terms of this Deed (including <u>paragraph 1</u> of <u>Part B</u> of <u>Schedule 7</u>), the Company shall use all reasonable endeavours to take, or cause to be taken, all appropriate action to do, or cause to be done, all things necessary, proper or advisable under Applicable Law or otherwise to consummate and make effective the transactions contemplated by this Deed and the Transaction Documents as promptly as practicable, including to:
 - (a) obtain from Governmental Entities and other persons all consents, approvals, authorisations, qualifications and orders as are necessary for the consummation of the transactions contemplated by this Deed and the Transaction Documents;
 - (b) promptly make all necessary filings, and thereafter make any other required submissions, with respect to this Deed required under the HSR Act or any other Applicable Law; and
 - (c) have vacated, lifted, reversed or overturned any order, decree, ruling, judgment, injunction or other action (whether temporary, preliminary or permanent) that is in effect and that enjoins, restrains, conditions, makes illegal or otherwise restricts or prohibits the consummation of the transactions contemplated by this Deed and the Transaction Documents.

In furtherance and not in limitation of the foregoing, the Company shall permit Qorvo reasonably to participate in the defence and settlement of any claim, suit or cause of action against the Company or its Board of Directors brought prior to Completion which relates to this Deed, or the transactions contemplated hereby, and the Company shall not settle or compromise any such claim, suit or cause of action without Qorvo's written consent, which shall not be unreasonably withheld or delayed. Each party shall bear its own costs and expenses incurred in relation to its obligations under this clause 5.4.

- 5.5 Following the delivery of the Offer Confirmation Notice and so long as the Option has not been rescinded by Qorvo in accordance with the provisions of <u>Schedule 7</u>, <u>Part B(1)</u>, each Major Investor, Executive Optionholder and Committed Shareholder shall use all reasonable endeavours to take, or cause to be taken, all appropriate action within its control or authority (taking into account its position as a shareholder or optionholder of the Company) to consummate the Completion.
- 5.6 The Company shall, and shall cause its Subsidiaries to, give promptly such notice to third parties (collectively, the "**Required Notices**") and to obtain such third party consents and estoppel certificates (collectively, the "**Required Consents**") as Qorvo may in its reasonable discretion deem necessary or desirable in connection with the transactions contemplated by this Deed and the Transaction Documents. Qorvo shall cooperate with and assist the Company in giving the Required Notices and obtaining the Required Consents; provided, that Qorvo shall have no obligation, in connection with giving the Required Notices or obtaining the Required Consents, to:
 - (a) give any guarantee or other consideration of any nature; or
 - (b) consent to any change in the terms of any agreement or arrangement that Qorvo in its sole discretion may deem adverse to the interests of Qorvo or the Company or any of its Subsidiaries.
- 5.7 Notwithstanding anything herein to the contrary, Qorvo shall not be required by any provision of this Deed to take or agree to undertake any action, including entering into any consent decree, hold separate order or other arrangement, that would (i) require the divestiture of any assets of Qorvo, the Company or any of their respective Affiliates or (ii) limit Qorvo's freedom of action with respect to, or its ability to consolidate and control, the Company and its Subsidiaries or any of their assets or businesses or any of Qorvo's or its Affiliates' other assets or businesses.
- 5.8 Until the earlier of (i) the termination of this Deed and (ii) the Completion of the Acquisition, each Major Investor, each Executive Optionholder and each Committed Shareholder irrevocably and unconditionally agrees that it shall at any meeting of the shareholders of the Company, however called, or in connection with any written consent of shareholders of the Company, vote (or consent), or cause to be voted at such meeting (or validly execute and return and cause such consent to be granted with respect to), all Shares held by such Major Investor, Committed Shareholder or Executive Optionholder against:
 - (a) any proposal that would result in a Change of Control of the Company, (other than in accordance with the transactions contemplated by this Deed);
 - (b) any proposal for any recapitalisation, reorganisation, liquidation, dissolution, amalgamation, merger, sale of assets or other business combination between the Company and any other person (other than in accordance with the transactions contemplated by this Deed); or

(c) any other action that could reasonably be expected to impede, interfere with, delay, postpone or adversely affect the transactions contemplated by this Deed or any transaction that results in a breach in any material respect of any covenant, representation or warranty or other obligation or agreement of the Company or any of its Subsidiaries under this Deed.

6. DUE DILIGENCE DURING THE DILIGENCE PERIOD

- 6.1 In order to enable Qorvo to make an informed decision whether to proceed with the Offer on exercise of the Option, it will be essential that Representatives of Qorvo, its auditing firm and legal counsel be afforded an opportunity to undertake expedited and thorough due diligence on the Company and the Group during the Diligence Period.
- 6.2 In furtherance of <u>clause 6.1</u> and <u>clause 4.1(e)</u>, the Company undertakes to Qorvo that the Company shall, during the Diligence Period, maintain the Data Room created for the purpose of the Series F Financing and shall ensure that the information contained in the Data Room is accurate, up to date and complete in all material respects during the Diligence Period. For the sake of clarity, prior to the receipt by the Company of the Option Notice, the Company shall have no obligation to maintain the Data Room. In particular, following the delivery of the Option Notice, the Company shall populate the Data Room with any diligence materials previously withheld from Qorvo (including unredacted versions of any redacted materials) during the Series F Financing or during the period prior to the receipt by the Option Notice.
- 6.3 The Company shall grant to Qorvo, its auditing firm, financial advisers and legal counsel access to the Data Room during the Option Period at any time on reasonable notice, and shall permit Qorvo, its auditing firm, financial advisers and legal counsel unrestricted access to the Data Room during the Diligence Period.
- 6.4 The Company shall grant to Qorvo, its auditing firm, financial advisers and legal counsel during the Diligence Period at any time on reasonable notice access to the Directors, officers, employees, consultants, advisers and other Representatives of the Company for interviews and discussions relating to the operations and financial position of the Company and potential employment arrangements following the Completion of the Acquisition.

7. THE OFFER

- 7.1 The Offer shall be made on substantially the same terms as are set out in the Offer Document attached as <u>Schedule 4</u>, and in accordance with the terms and conditions specified in this <u>clause 7</u>. The Contingent Securities shall be exchanged on the terms and conditions specified in <u>clause 8</u> for the amounts specified in this <u>clause 7</u>.
- 7.2 The "**Aggregate Value**" to be ascribed to the issued and to be issued share capital of the Company for the purposes of the Offer and the exchange of Contingent Securities (including, for the avoidance of doubt, any Series F Preferred Shares (or Ordinary Shares issued upon conversion of any Series F Preferred Shares) held by Qorvo pursuant to the Subscription

Agreement or the Additional Funding (if any)) and which is to be allocated to the Shares and Contingent Securities in accordance with the remainder of this <u>clause 7</u> and for the purposes of <u>clause 8</u> shall be an amount equal to:

- (a) an amount of USD \$300,000,000 plus the Aggregate Exercise Price (the "Equity Consideration"); less
- (b) the Actual Net Debt Amount; less
- (c) the Working Capital Shortfall (if any); plus
- (d) the Working Capital Excess (if any); plus
- (e) an amount equal to four and one half (4.5) times the "Annualised Legacy Product Revenue", where Annualised Legacy Product Revenue is equal to
 - (i) the Legacy Product Revenue (determined on the basis of UK GAAP) earned by the Company in the fiscal quarter ended immediately prior to the date of the Option Notice; multiplied by
 - (ii) a factor of four (4),

(such amount determined in accordance with <u>clause 7.2(e)</u>, the "**Revenue Payment**"; as an example, if the Option Notice is sent on 15 May 2016 and Legacy Product Revenue earned by the Company in the fiscal quarter ended 31 March 2016 was USD \$[*****], the Revenue Payment would equal USD \$[*****]); less

- (f) the Share of Stamp Duty.
- 7.3 The Estimated Consideration and Expense Fund Amount shall be allocated and paid in accordance with this Deed (including clause 7.4 and clause 7.6) on Completion of the Acquisition. The Revenue Payment and any Balancing Amount payable to take account of the Actual Net Debt Amount, any Working Capital Shortfall or any Working Capital Excess, shall be allocated and paid (in accordance with this Deed, including clause 7.4 and clause 7.6) after final agreement on, or final determination of, the Payment Statement in accordance with <u>Schedule 11</u> (Effective Time Statements) and after determination of the Legacy Product Revenue in accordance with <u>Schedule 15</u> as follows:
 - (a) if the Balancing Amount is zero (0), Qorvo shall allocate and pay to the Selling Shareholders an amount equal to the Revenue Payment, such payment to be distributed in accordance with the remainder of this <u>clause 7</u> and this Deed;
 - (b) if the Balancing Amount is a positive number, Qorvo shall allocate and pay to the Selling Shareholders an amount equal to the Revenue Payment plus the Balancing Amount, such payment to be distributed in accordance with the remainder of this <u>clause 7</u> and this Deed; and

- (c) if the Balancing Amount is a negative number, Qorvo shall allocate and pay to the Selling Shareholders an amount equal to the Revenue Payment less the Balancing Amount (expressed as a positive number), such payment to be distributed in accordance with the remainder of this <u>clause 7</u> and this Deed, provided that if the Revenue Payment less the Balancing Amount (expressed as a positive number) results in a negative number (such difference, the "**Overpayment Amount**"), each Selling Shareholder shall be obliged to repay to Qorvo its pro rata proportion (calculated based upon each Selling Shareholder's fully-diluted ownership interest in the Company relative to the other Selling Shareholders) of the Overpayment Amount (expressed as a positive number).
- 7.4 Payment of the Estimated Consideration which is due to the Selling Shareholders in accordance with the principles of allocation set out in clause 7.6 shall be made on the Completion of the Acquisition by Qorvo by wire transfer to the Payments Administrator, the Company (for distribution pursuant to clause 8.3) and the Dutch Subsidiary (for distribution pursuant to <u>clause 8.6</u>) (including, for purposes of clarification, subject to (i) a deduction from any such amount of the amount allocable to the Shares held by Qorvo in accordance with <u>clause 7.6</u>, (ii) an allocation of a portion of such amount to the holders of Contingent Securities in accordance with clause 7.6 and clause 8, and (iii) a deduction of the amount to be placed into the Escrow Account in accordance with clause 7.5 in respect of any payment of the Estimated Consideration due to the Warrantors). Payment of the Expense Fund Amount by Qorvo shall be made on the Completion of the Acquisition by Qorvo by wire transfer to the Securityholder Agent to hold in compliance with the terms of clause 46.4. Any payment of the Balancing Amount or the Revenue Payment to be made under clause 7.3 which is due to the Selling Shareholders in accordance with the principles of allocation set out in clause 7.6 (including, for purposes of clarification, subject to (i) a deduction from any such amount of the amount allocable to the Shares held by Qorvo in accordance with clause 7.6, (ii) an allocation of a portion of such amount to the holders of Contingent Securities in accordance with clause 7.6 and clause 8, and (iii) a deduction of the amount to be placed into the Escrow Account in accordance with <u>clause 7.5</u> in respect of any payment of the Balancing Amount and the Revenue Payment due to the Warrantors) shall be made on the later of:
 - (a) the date which is [*****] Business Days after (and excluding) the date on which the Effective Time Statements are agreed or determined in accordance with <u>Schedule 11</u> (Effective Time Statements); and
 - (b) the Completion of the Acquisition and the payment of the Estimated Consideration in accordance with this Deed,

by Qorvo by wire transfer to the Payments Administrator, the Company (for distribution pursuant to <u>clause 8.3</u>) and the Dutch Subsidiary (for distribution pursuant to <u>clause 8.6</u>); provided, however, that amounts allocated and payable with respect to vested Contingent Securities shall be paid to the holders of such vested Contingent Securities pursuant to <u>clause 8.3(a)</u> and <u>clause 8.6(a)</u>.

In the event of an Overpayment Amount, the Securityholder Agent shall send a notice to each Selling Shareholder, notifying them of their respective amounts due to Qorvo, such notice which

shall be mailed within [*****] Business Days after (and excluding) the date on which the Effective Time Statements are agreed or determined in accordance with <u>Schedule 11</u> (Effective Time Statements). Such notice shall be delivered in accordance with the provisions of <u>clause 33</u>. Thereafter, each Selling Shareholder shall promptly pay such amounts to Qorvo by check or wire transfer according to instructions provided to each Selling Shareholder by the Securityholder Agent, the details of which shall be notified by Qorvo to the Securityholder Agent within two (2) Business Days after (and excluding) the date on which the Effective Time Statements are agreed or determined in accordance with <u>Schedule 11</u> (Effective Time Statements).

- 7.5 On the Completion of the Acquisition, and concurrently with the payment to the Warrantors (as Selling Shareholders) of that part of the Estimated Consideration allocated and due to be paid to the Warrantors (as Selling Shareholders) as contemplated by <u>clause 7.4</u> and in accordance with the principles of allocation set out in <u>clause 7.6</u>, Qorvo shall deposit an amount equal to 13.75% of the Estimated Consideration due to the Warrantors (both with respect to consideration for Shares and with respect to the Contingent Securities pursuant to clause 8.3(a) and clause 8.6(a)) in an escrow account (the "Escrow Account") with the Escrow Agent as a partial mechanism to satisfy the obligations of the Warrantors set forth in clause 12 and <u>clause 13</u>. Concurrently with the payment to the Warrantors (as Selling Shareholders) of that part of the Balancing Amount and Revenue Payment due to be allocated and paid to the Warrantors (as Selling Shareholders) as contemplated by clause 7.4 and in accordance with the principles of allocation set out in clause 7.6, Qorvo shall deposit an amount equal to [*****]% of any amount of the Revenue Payment or the Balancing Amount due to the Warrantors (both with respect to consideration for Shares and with respect to the Contingent Securities pursuant to clause 8.3(a) and clause 8.6(a)) in the Escrow Account. Promptly following the end of a period of 18 months after the Completion Date, and in accordance with the terms of the Escrow Agreement, the Escrow Agent shall pay to the Warrantors, by wire transfer, the amount placed in the Escrow Account less: (a) any portion of the Escrow Account previously credited or paid in final resolution of any claims for breach of Warranties or indemnification made under <u>clause 12</u> and <u>clause 13</u>; and (b) an amount sufficient to satisfy any pending claims for breach of Warranties or indemnification made under clause 12 and clause 13, provided that such amounts shall be reasonable estimates determined by Qorvo in good faith. Promptly following final resolution of, and full payment or credit in connection with, such pending claims for breach of Warranties or indemnification made under clause 12 and clause 13, any remaining amounts in the Escrow Account shall be paid by the Escrow Agent to the Warrantors in accordance with the terms of the Escrow Agreement; provided, however, that any payments made to the Warrantors attributable to the Ltd Options purchased pursuant to <u>clause 8.3</u> or the Dutch Options pursuant to <u>clause 8.6</u> shall be subject to applicable tax and social security contributions and, as such, such payment amounts shall be paid to the Company or the Subsidiaries for payment through its payroll system. Any interest that accrues on the Escrow Account shall become part of the Escrow Account.
- 7.6 It is the intention of the parties that the Aggregate Value will be allocated among all holders of Shares and Contingent Securities of the Company in accordance with the Articles of Association as if the Acquisition constituted a Liquidation Event. The Offer Document will specify that the Aggregate Value will be allocated in accordance with the Articles of Association which,

assuming there is no amendment or replacement of the Articles of Association in the form set forth on <u>Schedule 12</u>, as follows (and each of Qorvo, the Major Investors, the Executive Optionholders and the Committed Shareholders agrees by execution of this Deed, and each Other Shareholder shall agree by execution of a Form of Acceptance, that the Aggregate Value will be allocated as follows):

- (a) an amount equal to the Series F Liquidation Preference will be allocated to the holders of the Series F Preferred Shares as follows:
 - (i) in the case of any holder of Series F Preferred Shares other than Qorvo, an amount equal to their pro rata entitlement to the Series F Liquidation Preference shall be allocated to such holders of Series F Preferred Shares out of the Aggregate Value; and
 - (ii) in the case of Qorvo, an amount equal to the pro rata entitlement of Qorvo to the Series F Liquidation Preference shall be deducted from the Aggregate Value to be allocated to the Selling Shareholders in respect of the Offer (but, for the avoidance of doubt, no funds shall be transferred in respect of the Series F Preferred Shares held by Qorvo);
- (b) an amount equal to the Series E Liquidation Preference will be allocated to the holders of the Series E Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association). For the avoidance of doubt, [*****] shall be entitled to receive an allocation of the Aggregate Value on any Series E Preferred Shares which may be issued to [*****] on exercise of the [*****] Warrants pursuant to <u>clause 8</u>;
- (c) an amount equal to the Series D Liquidation Preference will be allocated to the holders of the Series D Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association);
- (d) an amount equal to the Series C Liquidation Preference will be allocated to the holders of the Series C Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association);
- (e) an amount equal to the Series B-1 Liquidation Preference will be allocated to the holders of the Series B-1 Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association). For the avoidance of doubt, [*****] shall be entitled to receive an allocation of the Aggregate Value on any Series B-1 Preferred Shares which may be issued to [*****] on exercise of the [*****] pursuant to <u>clause 8</u>;
- (f) an amount equal to the Series B Liquidation Preference will be allocated to the holders of the Series B Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association);

- (g) an amount equal to the Series A Liquidation Preference will be allocated to the holders of the Series A Preferred Shares out of the Aggregate Value and divided pro rata amongst them (as per the Articles of Association); and
- (h) after all such amounts have been allocated out of the Aggregate Value, any amount of the Aggregate Value remaining to be allocated to the Shares shall be allocated to the holders of the Preferred Shares, the holders of Ordinary Shares and the holders of Contingent Securities (as if such Contingent Securities had been exercised in full for Ordinary Shares or Preferred Shares, as applicable, immediately prior to completion of the Acquisition provided that the amount allocated to such Contingent Securities shall be settled in accordance with <u>clause 8</u>) in accordance with article 5.2(b) of the Articles of Association, with each Preferred Share, Ordinary Share or Contingent Security entitling the holder of such instrument to one pro rata proportionate (calculated on an asconverted to Ordinary Shares basis) allocation of the remaining Aggregate Value after deduction of all Liquidation Preferences. The amounts to be paid to the holders of any Contingent Securities in exchange for such Contingent Securities shall be satisfied in accordance with <u>clause 8</u>, provided that each Contingent Security exercisable for Ordinary Shares or Preferred Shares shall be allocated the same value as any other Ordinary Share or Preferred Share (of the same class), although the amounts payable to the holder of such Contingent Security may: (i) be reduced by the amount of any sum which would be payable by such holder on exercise of such Contingent Security for any Ordinary Share or Preferred Share; or (ii) may be satisfied by equivalent securities issued by Qorvo of equivalent value to the cash consideration due in respect of such Contingent Securities. For the avoidance of doubt, Qorvo (as a holder of Series F Preferred Shares or Ordinary Shares issued or issuable upon conversion of the Series F Preferred Shares) shall be entitled to a proportionate amount of such Aggregate Value remaining after deduction of all Liquidation Preferences in accordance with the Articles of Association, but no funds shall actually be transferred in respect of the amount to which Qorvo is entitled. Also for the avoidance of doubt and in accordance with <u>clause 8</u>, all Contingent Securities shall, solely for purposes of the calculations in this <u>clause 7.6(h)</u>, be deemed to be exercised and Ordinary Shares or Preferred Shares, as applicable, issued in respect of such Contingent Securities prior to the allocation of the remaining Aggregate Value in accordance with this clause 7.6(h).

For the sake of clarity, if the Articles of Association in the form set forth on <u>Schedule 12</u> are amended or replaced following the date of this Deed (subject to compliance with the consent right of Qorvo in <u>paragraph 2</u> of <u>Part A</u> of <u>Schedule 6</u>), the Aggregate Consideration allocations will instead follow the distribution methodology set forth therein.

7.7 The total aggregate amounts to be allocated to the Shares held by the Selling Shareholders and any amounts allocated to Contingent Securities pursuant to <u>clause 7.6</u> and the Articles of Association shall be the "Aggregate Consideration" for the purposes of this Deed. For the avoidance of doubt, amounts allocated to the Series F Preferred Shares (and Ordinary Shares on conversion or deemed conversion of such Series F Preferred Shares) held by Qorvo shall not form part of the Aggregate Consideration.

- 7.8 The Aggregate Consideration shall be satisfied entirely by the payment of cash consideration by Qorvo in accordance with the terms of this Deed, provided that the Aggregate Consideration payable in respect of the Contingent Securities shall be satisfied in accordance with <u>clause 8</u> and may take the form of securities issued by Qorvo.
- 7.9 Payment of that part of the Aggregate Consideration which comprises the Estimated Consideration shall be satisfied on the Completion Date by:
 - (a) in respect of the Major Investors and the Committed Shareholders and any Other Shareholder or any other person who holds Shares (including any Executive Optionholder) who has executed and delivered a Form of Acceptance payment (with respect to the Shares held by such Major Investor, Committed Shareholder, Other Shareholder or other such person) by Qorvo to the nominated bank account of the Payments Administrator of an amount equal to the entitlement of such Major Investor, Committed Shareholder, Other Shareholder or other such person to share in that part of the Aggregate Consideration which comprises the Estimated Consideration in accordance with <u>clause 7.6</u> and the Articles of Association, but subject to the Escrow Account provisions in <u>clause 7.5</u>;
 - (b) in respect of any Contingent Shareholders (including the Executive Optionholders who continue to hold Contingent Securities) with respect to the Contingent Securities, in accordance with <u>clause 8</u>, but subject to the Escrow Account provisions in <u>clause 7.5</u>; and
 - (c) in respect of any Other Shareholder or any other person holding Shares who has not signed a Deed of Adherence to this Deed or executed and delivered a Form of Acceptance (the "**Defaulting Shareholders**"), and whose Shares shall be transferred to Qorvo in accordance with the requirements of the Squeeze-Out Acquisition or the Drag-Along Acquisition (as the case may be) payment by Qorvo of an amount equal to the entitlement of such persons to share in that part of the Aggregate Consideration which comprises the Estimated Consideration in accordance with <u>clause 7.6</u> and the Articles of Association (the "**Default Consideration**") in accordance with either <u>clause 10</u> or <u>clause 11</u>.
- 7.10 For the avoidance of doubt and notwithstanding anything in this Deed to the contrary, any payments to the holders of Ltd Options or Dutch Options or other compensatory payments shall not be sent to the Payments Administrator, but shall instead be sent to, and processed by, Company's or the Subsidiaries' payroll processor, for further distribution to the applicable recipients.
- 7.11 During the Offer Period, the Offer may be accepted by any Selling Shareholder or any other person who holds Shares by execution and delivery of a Form of Acceptance to the Company and Qorvo. The Major Investors and the Committed Shareholders (and any Executive Optionholder who exercises any Contingent Security on or after the date of this Deed and is issued any Ordinary Shares) irrevocably undertake to accept the Offer and to execute and deliver a Form of Acceptance to Qorvo and the Company promptly upon receipt of the Offer Document.

- 7.12 At the conclusion of the Offer Period:
 - (a) each of the Major Investors and the Committed Shareholders, and any Other Shareholder or any other person who holds Shares who has executed and delivered a Form of Acceptance to the Company and Qorvo (including any Executive Optionholder, if applicable), shall transfer the Shares held by such person to Qorvo in accordance with clause 9, subject to any delay in such transfer required to simultaneously complete the transfer of all Shares to Qorvo in accordance with clause 7.12(c). In the event that Qorvo elects to implement the Squeeze-Out Acquisition, there shall be no delay in the completion of the transfer of such Shares unless Qorvo determines that such delay would in any event permit Qorvo to exercise its rights to effect the Squeeze-Out Acquisition pursuant to the Companies Act 2006; and
 - (b) each of the Contingent Securities shall be transferred, exchanged, lapsed or terminated in accordance with <u>clause 8</u>;
 - (c) if:
 - (i) one hundred per cent (100%) of the Shares of the Company are to be transferred to Qorvo in accordance with <u>clause 7.12(a)</u> the Completion of the Acquisition shall proceed in accordance with <u>clause 9</u>;
 - (ii) the number of Ordinary Shares or any class of Preferred Shares to be transferred to Qorvo in accordance with <u>clause 7.12(a)</u> constitutes at least [*****] of the entire issued Ordinary Shares and each class and series of Preferred Shares (as the case may be and in each case excluding those held by Qorvo) as at the conclusion of the Offer Period each of the Major Investors, the Executive Optionholders and the Company shall, and the Committed Shareholders shall use reasonable endeavours to, offer all necessary assistance to enable Qorvo to complete a Squeeze-Out Acquisition of the remaining Shares in accordance with <u>clause 10</u>;
 - (iii) the number of Shares to be transferred to Qorvo in accordance with <u>clause 7.12(a)</u> constitutes less than [*****] of any of the entire issued Ordinary Shares or any class or series of Preferred Shares (as the case may be and in each case excluding those held by Qorvo) as at the conclusion of the Offer Period each of the Major Investors, the Executive Optionholders and the Company shall, and the Committed Shareholders shall use reasonable endeavours to, render all necessary assistance to Qorvo to enable Qorvo to complete a Drag-Along Acquisition of the remaining Shares in accordance with <u>clause 11</u>; or
 - (iv) for any reason, any Contingent Securities shall not be capable of transfer, exchange or termination in accordance with <u>clause 8</u> – each of the Major Investors, the Executive Optionholders and the Company shall, and the Committed Shareholders shall use reasonable endeavours to, render all necessary assistance to Qorvo to enable Qorvo to complete a Drag-Along Acquisition of such Contingent Securities in accordance with <u>clause 11</u>.

7.13 The parties agree that the Offer shall be made in accordance with the exemptions to section 21 of the Financial Services and Markets Act 2000 (UK) and in accordance with the requirements of either article 62 or articles 63 and 64 of the Financial Services and Markets Act 2000 (Financial Promotion) Order 2005, and, accordingly, the parties agree to undertake all necessary actions to comply with such requirements.

8. TREATMENT OF THE CONTINGENT SECURITIES

Ltd Options

- 8.1 On Completion of the Acquisition, any Ltd Options or Dutch Options held by any Contingent Shareholder shall be dealt with in accordance with this <u>clause 8</u> and the Company shall procure that all Ltd Options and Dutch Options shall be dealt with in accordance with this <u>clause 8</u> and that the Administrators of the relevant Employee Share Scheme shall ensure that all Ltd Options and Dutch Options shall be dealt with in accordance with this <u>clause 8</u> and that the Administrators of the relevant Employee Share Scheme shall ensure that all Ltd Options and Dutch Options shall be dealt with in accordance with this <u>clause 8</u>. Any treatment of the Ltd Options provided for under this <u>clause 8</u> that gives rise to a taxable event for the Contingent Shareholder shall be settled according to this <u>clause 8</u>, less applicable withholding and tax liabilities due upon such settlement.
- 8.2 Pursuant to Section 15(a) of the Employee Share Scheme governing the Ltd Options, prior to or as soon as reasonably practicable following the execution of this Deed, but in no event later than ten (10) days following the execution of this Deed so long as Qorvo has provided its approval thereof pursuant to Section 8.2(c) hereof, the Administrator shall amend the Employee Share Scheme and adopt any other resolution or clarification required to:
 - (a) in respect of any Ltd Options which have vested or which are otherwise exercisable for Ordinary Shares on Completion of the Acquisition, provide for the purchase by Qorvo of each such Ltd Option held by a Contingent Shareholder for an amount of cash (subject to applicable tax withholding) equal to the amount that could have been obtained by the Contingent Shareholder, net of the exercise price of such Ltd Option, had the exercise of such Ltd Option and the issuance of any Ordinary Shares in respect of such Ltd Option been consummated immediately prior to the Acquisition and such Ordinary Shares been sold to Qorvo pursuant to the Acquisition, such amount being equal to:
 - (i) the proportion of the Aggregate Consideration due with respect to an Ordinary Share pursuant to <u>clause 7.6</u> multiplied by the number of Ordinary Shares issuable upon exercise of such Ltd Option; less
 - (ii) the aggregate exercise price attributable to such Ltd Option; and
 - (b) in respect of any Ltd Options which have not vested or which are not otherwise exercisable for Ordinary Shares on Completion of the Acquisition ("**Unvested Ltd Options**"), provide that each such Unvested Ltd Option held by a Contingent Shareholder be exchanged or substituted for a similar option, right or award covering the common stock of Parent, on the same terms and conditions (including any vesting or forfeiture provisions or repurchase rights) as were applicable under such Unvested Ltd

Option as of immediately prior to the Completion, with appropriate adjustments as to the number and kind of stock and the exercise price attributable to such similar option, right or award, as follows in respect of each such Unvested Ltd Option held by a Contingent Shareholder:

- (i) the number of shares of common stock of Parent underlying the option, right or award to be issued to the Contingent Shareholder shall be equal to the result of the following calculation (rounded down to the nearest whole number):
 - (A) the total amount of the Aggregate Consideration allocated to such Contingent Shareholder in respect of such Unvested Ltd Option held by such Contingent Shareholder; *divided by*
 - (B) the Parent Stock Completion Value; and
- (ii) the exercise price payable in respect of the exercise of such option, right or award shall be equal to the result of the following calculation (rounded up to the nearest whole cent):
 - (A) the aggregate exercise price which would have been payable by such Contingent Shareholder on exercise of such Unvested Ltd Option held by such Contingent Shareholder which has not vested or which is not otherwise exercisable for Ordinary Shares on Completion of the Acquisition; *divided by*
 - (B) the total number of shares of common stock of Parent underlying the option, right or award to be issued to the Contingent Shareholder as calculated pursuant to <u>clause 8.2(b)(i)</u>;

provided, that in the case of any Unvested Ltd Option to which Section 421 of the Code applies as of the Completion (taking into account the effect of any accelerated vesting thereof, if applicable) by reason of its qualification under Section 422 of the Code, the exercise price, the number of shares of common stock of Parent subject to such option, right or award and the terms and conditions of exercise of such option, right or award shall be determined in a manner consistent with the requirements of Section 424(a) of the Code to maintain the qualification of such options under Section 422 of the Code; provided further, that in the case of any Unvested Ltd Option to which Section 422 of the Code does not apply as of the Completion, the exercise price, the number of shares of common stock of Parent subject to such option, right or award and the terms and conditions of exercise of such option shall be determined in a manner stock of Parent subject to such option, right or award and the terms and conditions of exercise of such option shall be determined in a manner stock of Parent subject to such option, right or award and the terms and conditions of exercise of such option shall be determined in a manner consistent with the requirements of Section 409A of the Code.

- (c) Any amendment or resolution adopted by the Company in connection with this <u>clause 8.2</u> shall be subject to approval by Qorvo. Each of the Executive Optionholders and Committed Shareholders who are holders of Ltd Options hereby agree to and approve the Amendment #1 to the Employee Share Scheme governing the Ltd Options to each of such holder's option grant agreements under the Employee Share Scheme governing the Ltd Options, as set forth on <u>Schedule 18</u> to this Deed.
- 8.3 On Completion and following determination of the treatment applicable to all Ltd Options in accordance with <u>clause 8.2</u>, Qorvo shall:
 - (a) pay to the Company for distribution pursuant to its payroll processor, such sum of cash as is equal to the aggregate amounts due to the holders of Ltd Options that have been dealt with in accordance with <u>clause 8.2(a)</u>, provided that for payments attributable to Ltd Options held by Warrantors which have vested or which are otherwise exercisable for Ordinary Shares on Completion of the Acquisition, an amount of [*****]% of such sum payable to the Warrantors shall be paid into the Escrow Account and dealt with in accordance with <u>clause 7.5</u> in part satisfaction of the obligations of the Executive Optionholders under <u>clause 12</u> and <u>clause 13</u>; and, provided, further, that such payments pursuant to this <u>clause 8.3(a)</u> may be made in separate payments upon Completion and upon determination of the Revenue Payment and any Balancing Payment; and
 - (b) issue to the relevant Contingent Shareholders such options, rights or awards covering the common stock of Parent as is required in respect of Ltd Options that have been dealt with in accordance with <u>clause 8.2(b)</u>, which issuance may be following the determination of the Revenue Payment and Balancing Payment.
- 8.3.1 After receipt of the sum of cash pursuant to <u>clause 8.3(a)</u> for the benefit of the holders of Ltd Options that have been dealt with in accordance with <u>clause 8.2(a)</u>, the Company shall pay such amounts directly to the holders of Ltd Options through payroll, withholding in all cases, any amounts necessary to cover any wage tax and social security contributions payable by the Company to the tax authorities in respect of such Ltd Options in accordance with the relevant Employee Share Scheme governing these Ltd Options.

Dutch Options

- 8.4 On Completion of the Acquisition, any Dutch Options held by any Contingent Shareholder shall be dealt with in accordance with this <u>clause 8</u> and the Company shall procure that all Dutch Options shall be dealt with in accordance with this <u>clause 8</u> and that the Administrators of the relevant Employee Share Scheme shall ensure that all Dutch Options shall be dealt with in accordance with this <u>clause 8</u>.
- 8.5 Prior to or as soon as reasonably practicable following the execution of this Deed, but in no event later than ten (10) days following the execution of this Deed so long as Qorvo has provided its approval thereof, the Administrator shall amend the Employee Share Scheme and adopt any other resolution or clarification required to:

- (a) in respect of any Dutch Options which have vested or which are otherwise exercisable for Ordinary Shares on Completion of the Acquisition – provide for the purchase by Qorvo of each such Dutch Option held by a Contingent Shareholder for an amount of cash equal to the amount that could have been obtained by the Contingent Shareholder had the exercise of such Dutch Option and the issuance of any Ordinary Share in respect of such Dutch Option been consummated immediately prior to the Acquisition and such Ordinary Shares been sold to Qorvo pursuant to the Acquisition, such amount being equal to:
 - (i) the proportion of the Aggregate Consideration due with respect to an Ordinary Share pursuant to <u>clause 7.6</u> multiplied by the number of Ordinary Shares issuable upon exercise of such Dutch Option; less
 - (ii) the aggregate exercise price attributable to such Dutch Option; and
- (b) in respect of any Dutch Options which have not vested or which are not otherwise exercisable for Ordinary Shares on Completion of the Acquisition ("**Unvested Dutch Options**") – provide that each such Unvested Dutch Option held by a Contingent Shareholder be exchanged or substituted for a similar option, right or award covering the common stock of Parent, on the same terms and conditions (including any vesting or forfeiture provisions or repurchase rights) as were applicable under such Unvested Dutch Option as of immediately prior to the Completion, with appropriate adjustments as to the number and kind of stocks and the exercise prices attributable to such similar option, right or award, as follows in respect of each such Unvested Dutch Option held by a Contingent Shareholder:
 - (i) the number of shares of common stock of Parent underlying the option, right or award to be issued to the Contingent Shareholder shall be equal to the result of the following calculation (rounded down to the nearest whole number):
 - (A) the total amount of the Aggregate Consideration allocated to such Contingent Shareholder in respect of such Unvested Dutch Option held by such Contingent Shareholder; *divided by*
 - (B) the Parent Stock Completion Value; and
 - (ii) the exercise price payable in respect of the exercise of such option, right or award shall be equal to the result of the following calculation (rounded up to the nearest whole cent):
 - (A) the aggregate exercise price which would have been payable by such Contingent Shareholder on exercise of such Unvested Dutch Option held by such Contingent Shareholder which has not vested or which is not otherwise exercisable for Ordinary Shares on Completion of the Acquisition; *divided by*

- (B) the total number of shares of common stock of Parent underlying the option, right or award to be issued to the Contingent Shareholder as calculated pursuant to <u>clause 8.5(b)(i)</u>.
- (c) Any amendment or resolution adopted by the Company in connection with this <u>clause 8.5</u> shall be subject to approval by Qorvo.
- 8.6 On Completion and following determination of the treatment applicable to all Dutch Options in accordance with <u>clause 8.5</u>, Qorvo shall:
 - (a) pay to the Dutch Subsidiary for distribution to the holders of Dutch Options such sum of cash as is equal to the aggregate amounts due to the holders of Dutch Options that have been dealt with in accordance with <u>clause 8.5(a)</u>, provided that for payments attributable to the Dutch Options held by Warrantors which have vested or which are otherwise exercisable for Ordinary Shares on Completion of the Acquisition an amount of 13.75% of such sum payable to the Warrantors shall be paid into the Escrow Account and dealt with in accordance with <u>clause 7.5</u> in part satisfaction of the obligations of the Executive Optionholders under <u>clause 12</u> and <u>clause 13</u>; and, provided, further, that such payments pursuant to this <u>clause 8.3</u> may be made in separate payments upon Completion and upon determination of the Revenue Payment and any Balancing Payment; and
 - (b) issue to the relevant Contingent Shareholders such options, rights or awards covering the securities or stock of Qorvo as is required in respect of Dutch Options that have been dealt with in accordance with <u>clause 8.5(b)</u>, which issuance may be following the determination of the Revenue Payment and Balancing Payment.
- 8.6.1 After receipt by the Dutch Subsidiary of the sum of cash pursuant to <u>clause 8.6(a)</u> for the benefit of the holders of Dutch Options that have been dealt with in accordance with <u>clause 8.5(a)</u>, the Company shall be obliged to procure that the Dutch Subsidiary pays such amounts directly to the holders of Dutch Options through payroll, in all cases after deduction of any Dutch wage tax and social security contributions payable by the Dutch Subsidiary to the Dutch tax authorities in connection with the sale of such Dutch Options in accordance with the relevant Employee Share Scheme governing the Dutch Options.

[*****] Warrants

- 8.7 [*****] shall be given notice of the Offer in accordance with <u>clause 4.1</u>.
- 8.8 In the event that [*****] shall exercise either of the [*****] Warrants in accordance with the [*****] Warrant Instrument, the Company shall issue to [*****] the Series E Preferred Shares and/or the Series B-1 Preferred Shares, as applicable, thereby subscribed by [*****].
- 8.9 Upon issuance of the Series E Preferred Shares and/or the Series B-1 Preferred Shares, as applicable, to [*****], the Offer Document shall be sent to [*****] by the Company.

- 8.10 In the event that [*****] shall execute and deliver to the Company and Qorvo a valid Form of Acceptance, the Preferred Shares held by [*****] shall be transferred to Qorvo in accordance with <u>clause 9</u> on Completion of the Acquisition.
- 8.11 In the event that [*****] fails to execute and deliver to the Company any Qorvo a valid Form of Acceptance during the Offer Period, the Preferred Shares held by [*****] shall be made subject to a Squeeze-Out Acquisition in accordance with clause 10.
- 8.12 The amount of Aggregate Consideration payable in respect of any Preferred Shares held by [*****] shall be equivalent to the amount allocated to such Preferred Shares in accordance with <u>clause 7.6</u>.

General provisions relating to the Contingent Securities

- 8.13 Each of the Company, and, to the extent they hold Contingent Securities, and in each case, only as to the Contingent Securities held by them, the Major Investors, the Executive Optionholders and the Committed Shareholders, warrants to Qorvo that, upon satisfaction of the rights and obligations of Qorvo and the holders of the Contingent Securities pursuant to this <u>clause 8</u>, at Completion, all Contingent Securities of the Company shall have been either:
 - (a) transferred to Qorvo for cash;
 - (b) exchanged by Qorvo for similar options, rights or awards covering the securities or stock of Qorvo;
 - (c) converted into Shares transferable to Qorvo pursuant to the Offer, the Squeeze-Out Acquisition or the Drag-Along Acquisition; or
 - (d) lapsed, terminated or otherwise have become incapable of being exercised for Shares in the capital of the Company.
- 8.14 Qorvo shall not be obliged to complete any action under this <u>clause 8</u> in respect of the Contingent Securities unless the sale and purchase of all the Shares and Contingent Securities is completed simultaneously, including, for the avoidance of doubt, the sale and purchase of any Shares to be purchased as part of any Squeeze-Out Acquisition or Drag-Along Acquisition. The parties to this Deed intend and agree that: (a) neither the delivery by Qorvo of an Option Notice or Option Confirmation Notice nor any other action taken by Qorvo pursuant to this Deed prior to the Completion shall constitute an "Acquisition" pursuant to any Employee Share Scheme or affect any Ltd Option or Dutch Option; (b) the treatment of the Ltd Options and Dutch Options as set forth in this <u>clause 8</u> shall not take place unless and until the Completion occurs.
- 8.15 To the maximum extent permitted by Applicable Law and the Employee Share Schemes, the Company covenants and undertakes to Qorvo that it will not, and it will not permit, the rights of any Contingent Shareholder under any Contingent Securities to be accelerated during the Option Period or otherwise varied to enable any Contingent Shareholder to exercise such Contingent

Securities other than in accordance with the terms and conditions applicable to such Contingent Securities as at the date of this Deed.

- 8.16 The Company undertakes that:
 - (a) on and from the date of this Deed until the Completion of the Acquisition or the termination of this Deed (as the case may be), it shall not grant or issue any form of Contingent Security other than Contingent Securities permitted in <u>paragraph 9</u> of <u>Part A</u> of <u>Schedule 6</u> and Contingent Securities convertible or exercisable for up to a maximum of [*****] Ordinary Shares issued pursuant to the Employee Share Schemes as in force at the date of this Deed (and as amended in connection with <u>clause 8.2</u> and <u>clause 8.5</u>), provided, however, that any Contingent Securities granted or issued pursuant to the Employee Share Schemes that expire or terminate unexercised or any Ordinary Shares repurchased by the Company shall not be counted toward such maximum number unless and until such shares are re-granted as new grants or awards pursuant to the terms of the Employee Share Schemes; and
 - (b) on and from the date of the Offer until the Completion of the Acquisition or the termination of this Deed (as the case may be), it shall not grant or issue any form of Contingent Security.
- 8.17 Following the date hereof, within 15 days of the end of each of the Company's fiscal quarters, the Company shall notify Qorvo of the issuance of any Contingent Securities made in such quarter, and shall deliver to Qorvo an update of <u>Schedule 3</u> setting out full details of all Contingent Shareholders.
- 8.18 Qorvo shall procure that Parent shall reserve for issuance a number of shares of common stock of Parent at least equal to the number of shares of common stock of Parent that will be subject to the options, rights and awards to be issued pursuant to this <u>clause 8</u>. Qorvo shall procure that, as soon as practicable, but not later than five Business Days, following the Completion, Parent shall file a registration statement on Form S-8 (or any successor form, or if Form S-8 is not available, other appropriate forms) with respect to offers and sales of the shares of common stock of Parent subject to such options, rights and awards and shall use its reasonable best efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as such options, rights and awards remain outstanding and the offers and sales of common stock of Parent thereunder are required to be registered. In connection with the transactions contemplated by <u>clause 8</u>, at the Completion, Qorvo may procure that Parent (if Parent determines that it desires to do so) assume any or all of the Employee Share Schemes governing the Ltd Options and the Dutch Options, provided that, if Parent elects to assume any such Employee Share Schemes, then, except as expressly provided in this clause 8 with respect to options, rights and awards and common stock of Parent to be issued in exchange for Ltd Options and Dutch Options, notwithstanding any provision of such Employee Share Schemes, Parent shall not be entitled to grant any stock awards, whether or not permissible under applicable Law, using the share reserves of such Employee Share Schemes and, as of the Completion, all such then-ungranted shares that were reserved for issuance under such Employee Share Schemes shall thereafter be deemed not to be

reserved for issuance and no such shares shall be issued or issuable under such Employee Share Scheme. To the extent any options, rights or awards relating to common stock of Parent are issued as provided for in this <u>clause 8</u> and such issuance is under such Employee Share Schemes, (A) stock covered by such awards shall be shares of common stock of Parent as provided in this <u>clause 8</u>; (B) subject to the proviso in the immediately preceding sentence, all references in such Employee Share Scheme to a number of Ordinary Shares shall be deemed amended to refer instead to a number of shares of common stock of Parent or a committee thereof shall succeed to the authority and responsibility of the Board of Directors of the Company or any committee thereof or any other administrator with respect to the administration of such Employee Share Scheme. Qorvo shall be liable for any failure by Parent to take any of the actions it is obligated to take under this <u>clause 8.12</u>.

9. COMPLETION OF THE TRANSFER OF SHARES

- 9.1 The Major Investors and the Committed Shareholders (and any other Selling Shareholder holding the Shares) shall sell and Qorvo shall purchase the Shares with all rights attached or accruing to them at Completion.
- 9.2 Each Selling Shareholder who is a party to this Deed or who executes and delivers a Form of Acceptance during the Offer Period warrants to Qorvo that: (a) such Selling Shareholder is the legal and beneficial owner of the Shares and Contingent Securities held by such Selling Shareholder, free and clear of any Liens, (b) such Selling Shareholder has the right, authority and power to sell, assign and transfer the Shares to Qorvo, and (c) upon the Completion, Qorvo shall acquire good, valid and marketable title to the Shares, free and clear of any Liens other than Liens created by Qorvo.
- 9.3 Qorvo shall be entitled to exercise all rights attached or accruing to the Shares including, without limitation, the right to receive all dividends, distributions or any return of capital declared, paid or made by the Company on or after the Completion of the Acquisition.
- 9.4 With effect from Completion of the Acquisition, the Major Investors and the Committed Shareholders (and any other Selling Shareholders who are party to this Deed or who execute and deliver a Form of Acceptance during the Offer Period) waive all rights of pre-emption over any of the Shares conferred upon them by the Articles of Association or in any other way and undertake to take all steps necessary to ensure that any rights of pre-emption over any of the Shares are waived.
- 9.5 Completion of the sale and purchase of the Shares shall take place at the offices of Qorvo's Solicitors on:
 - (a) the date which is the second Business Day after the expiration of the Offer Period, if <u>clause 7.12(c)(i)</u> applies;
 - (b) the date on which the completion of the Squeeze-Out Acquisition occurs, if <u>clause 7.12(c)(ii)</u> applies, provided that if Qorvo so determines completion of the sale and purchase of

the Shares shall take place on such earlier date as may be necessary to enable Qorvo to effect the Squeeze-Out Acquisition; or

(c) the date on which the completion of the Drag-Along Acquisition occurs, if <u>clause 7.12(c)(iii)</u> or <u>clause 7.12(c)(iv)</u> applies,

(such date, the "Completion Date").

- 9.6 On the Completion Date, the Major Investors and the Committed Shareholders (and any other Selling Shareholders who are party to this Deed or who execute and deliver a Form of Acceptance during the Offer Period) shall procure the delivery to Qorvo of:
 - (a) a duly executed transfer in respect of the Shares to be transferred by such Selling Shareholder in favour of Qorvo (or its nominee(s));
 - (b) to the extent there are such certificates in existence, the share certificate(s) representing the Shares held by such Selling Shareholder and, to the extent that any such certificates have been lost or destroyed, an indemnity in respect thereof;
 - (c) such waivers or consents as may be necessary to enable Qorvo (or its nominee(s)) to become the registered holder(s) of the relevant Shares;
 - (d) powers of attorney executed by or on behalf of the Major Investors and Committed Shareholders (and any other Selling Shareholders who are party to this Deed or who execute and deliver a Form of Acceptance during the Offer Period) to vote relevant Shares, pending registration of the transfer of those Shares in the registers of the Company, in accordance with the instructions of Qorvo, in the form acceptable to Qorvo; and
 - (e) such other documents as may be necessary to enable Qorvo (or its nominee(s)) to obtain good legal and beneficial title to the relevant Shares.
- 9.7 Qorvo shall not be obliged to complete the sale and purchase of any of the relevant Shares unless the sale and purchase of all the Shares and Contingent Securities is completed simultaneously, including, for the avoidance of doubt, the sale and purchase of any Shares and Contingent Securities to be purchased as part of any Squeeze-Out Acquisition or Drag-Along Acquisition.
- 9.8 At Completion:
 - (a) the Major Investors shall do all those things required of them in <u>Part A</u> of <u>Schedule 7</u>;
 - (b) the Committed Shareholders shall do all those things required of them in <u>Part A</u> of <u>Schedule 7</u>;
 - (c) the Executive Optionholders shall do all those things required of them in <u>Part A</u> of <u>Schedule 7</u>;

- (d) the Company shall do all those things required of it in <u>Part A</u> of <u>Schedule 7</u>; and
- (e) Qorvo shall do all those things required of it in <u>Part A</u> of <u>Schedule 7</u>.
- 9.9 Qorvo is not obliged to proceed with the Completion of the Acquisition unless (a) the Major Investors, the Committed Shareholders, the Executive Optionholders and the Company comply with all of their obligations under this <u>clause 9</u> and those set forth on <u>Part A</u> of <u>Schedule 7</u> and (b) the conditions set forth on <u>Part B</u> of <u>Schedule 7</u> are complied with during the Offer Period and on or prior to the Completion Date.
- 9.10 If the obligations of any of the Major Investors, the Committed Shareholders, the Executive Optionholders or the Company under this <u>clause 9</u> or <u>Schedule 7</u> are not complied with on the Completion Date, Qorvo may:
 - (a) if the failure for the obligations and conditions required by this <u>clause 9</u> or <u>Schedule 7</u> are curable by the Company, the Major Investors, the Committed Shareholders or the Executive Optionholders, defer Completion (so that the provisions of this <u>clause 9</u> shall apply to Completion as so deferred); provided that, so long as the Company, the Major Investors, the Committed Shareholders and the Executive Optionholders are using their reasonable endeavours and working diligently to cure the failure during such period, such deferral may be for a period of no more than 90 days;
 - (b) proceed to Completion as far as practicable (without limiting its rights under this Deed); or
 - (c) terminate this Deed by notice in writing to the Company (with a copy to the Securityholder Agent).
- 9.11 If this Deed is terminated in accordance with <u>clause 9.10(c)</u>, all obligations of Qorvo under this Deed shall end except for those expressly stated to continue without limit in time, but (for the avoidance of doubt) all rights and liabilities of the parties which have accrued before termination shall continue to exist and Qorvo shall be entitled to pursue any cause of action or claim it may have against the Company or against a Major Investor, Committed Shareholder or an Executive Optionholder (but in the event of a claim against a Major Investor, Committed Shareholder or an Executive Optionholder, such claim may only be brought against the breaching party and not against any other Major Investor, Committed Shareholder or an Executive Optionholder), including seeking damages (including all costs and expenses incurred by Qorvo in connection with this Deed and the Offer).
- 9.12 Subject to the terms of this Deed, any provision of this Deed and any other documents referred to in it which is capable of being performed after but which has not been performed at or before Completion and all Warranties and covenants and other undertakings contained in or entered into pursuant to this Deed shall remain in full force and effect notwithstanding Completion.
- 9.13 If at any time Qorvo or the Company becomes aware of a fact or circumstance that might prevent any obligations or conditions specified in <u>Part B</u> of <u>Schedule 7</u> being satisfied, such party shall

promptly inform the other parties and the Securityholder Agent. If at any time the Major Investors, the Executive Optionholders or any Committed Shareholder knows of a fact or circumstance that would prevent any obligations or conditions specified in <u>Part B</u> of <u>Schedule 7</u> being satisfied, such party shall promptly inform Qorvo and the Company.

- 9.14 Each of the Company, the Major Investors, the Committed Shareholders and the Executive Optionholders undertakes to use all reasonable endeavours to procure (to the extent within its control) the fulfilment of the conditions set out in <u>Part B</u> of <u>Schedule 7</u> and shall keep the other parties reasonably advised of the progress towards satisfaction of the same.
- 9.15 At any time prior to the conclusion of the Offer Period or the Completion of the Acquisition (as the case may be), Qorvo may waive any of the conditions set out in <u>Schedule 7</u> by notice to the Company, provided however, that Qorvo shall not waive any conditions which are specifically required for the transactions contemplated hereunder by Applicable Law.

10. SQUEEZE-OUT ACQUISITION

- 10.1 If this <u>clause 10</u> applies at the conclusion of the Offer pursuant to <u>clause 7.12(c)(ii)</u>, on or after the conclusion of the Offer Period and at any time up to [*****] after the conclusion of the Offer Period, Qorvo may compulsorily acquire the Shares of any Other Shareholder or any other person who holds Shares who has not executed and delivered to Qorvo and the Company a Form of Acceptance pursuant to the rights of Qorvo under section 979 of the Companies Act (the "Squeeze-Out Acquisition").
- 10.2 To facilitate the Squeeze-Out Acquisition, Qorvo may as soon as reasonably practicable after the conclusion of the Offer Period, but in any event within [*****] days thereof:
 - (a) serve by recorded delivery a Form 980(1) Statutory Notice on any Other Shareholder who has not executed and delivered to Qorvo and the Company a Form of Acceptance in respect of the Shares held by such Other Shareholder;
 - (b) send to the Company by recorded delivery a copy of the first Form 980(1) Statutory Notice served in respect of any Shares under <u>clause 10.2(a)</u>, together with a Form 980 Statutory Declaration sworn by a director of Qorvo containing all necessary particulars under the Companies Act, including stating that the conditions for giving the declaration have been satisfied.
- 10.3 [*****] weeks after service of the notices required in accordance with <u>clause 10.2</u>, Qorvo shall:
 - (a) send to the Company by recorded delivery a copy of all Form 980(1) Statutory Notices served by Qorvo in respect of any Shares together with stock transfer forms for the transfer of the Shares subject to the Squeeze-Out Acquisition, duly executed on behalf of the Other Shareholder and duly stamped with ad valorem stamp duty; and
 - (b) pay to the Company all consideration due to the Other Shareholders and any other person in respect of such Shares in accordance with the terms of the Offer.

- 10.4 Upon receipt of all cash and documents required to be delivered by Qorvo pursuant to <u>clause 10.3</u>, the Company shall cause Qorvo to be entered into the register as the holder of all Shares to be transferred pursuant to the Squeeze-Out Acquisition and the remaining transfers of Shares shall be completed in accordance with this Deed.
- 10.5 The Company must hold all consideration received from Qorvo under <u>clause 10.3</u> on trust for the Other Shareholders and must hold such money in a separate, interest-bearing bank account. The Company must, as soon as reasonably practicable, transfer such proceeds to the Other Shareholders and otherwise deal with such proceeds in accordance with the requirements of section 982 of the Companies Act.

11. DRAG-ALONG ACQUISITION

- 11.1 Each of the Major Investors, the Executive Optionholders and the Committed Shareholders and the Company warrants to, and agrees with, Qorvo that the Offer shall constitute a Qualifying Offer under the Articles of Association and each of the Major Investors, the Executive Optionholders and the Committed Shareholders undertakes to vote all Shares held by the Major Investors, the Executive Optionholders and the Committed Shareholder, as may be necessary to ensure that the Offer shall constitute a Qualifying Offer under the Articles of Association. Additionally, each of the Major Investors and the Executive Optionholders shall, and the Committed Shareholders shall use reasonable endeavours to, take such acts and execute all such documents as may be necessary to ensure that the Offer shall constitute a Qualifying Offer under the Articles of Association.
- 11.2 If this <u>clause 11</u> applies at the conclusion of the Offer pursuant to <u>clause 7.12(c)(iii)</u> or <u>clause 7.12(c)(iv)</u> and following the conversion of Contingent Securities into Shares in accordance with <u>clause 8</u>, as the Major Investors shall have accepted the Offer pursuant to the undertakings given by them in <u>clause 4.1(g)(iii)</u> and elsewhere in this Deed, the Major Investors warrant and agree that they shall be entitled under the Articles of Association to serve a Drag-Along Notice on any Other Shareholders who have not executed and delivered a Form of Acceptance to Qorvo and the Company, and that upon service of such Drag-Along Notice such Other Shareholders shall become bound to accept the Offer (as a Qualifying Offer) and to transfer their Shares to Qorvo (or its nominee) with full title guarantee on the date specified in the Drag-Along Notice.
- 11.3 Accordingly, on conclusion of the Offer Period, if this <u>clause 11</u> applies pursuant to <u>clause 7.12(c)(iii)</u> or <u>clause 7.12(c)(iv)</u>, and following the receipt of a written request from Qorvo, which must be delivered within [*****] days following the conclusion of the Offer Period, each of the Major Investors irrevocably undertakes to Qorvo that within [*****] days of receipt of such notice the Major Investors shall serve a Drag-Along Notice on each Other Shareholder who has not executed and delivered a Form of Acceptance to Qorvo and the Company.
- 11.4 If any Other Shareholder served with such Drag-Along Notice shall not, within [*****] Business Days of being requested or deemed to have been requested by the Drag-Along Notice to do so, execute transfers in respect of all the Shares held by him and deliver to Qorvo the transfers and the certificate for such Shares (or a suitable indemnity in place of it), then, pursuant to the rights

granted to the Major Investors under the Articles of Association, the Major Investors hereby undertake to nominate and appoint a person identified by Qorvo as agent and attorney of such Other Shareholder (the "**Defaulting Drag-Along Shareholder**") with authority to:

- (a) execute and deliver such transfer on his behalf;
- (b) against receipt of such person of the consideration receivable by the Defaulting Drag-Along Shareholder under the Offer (as a Qualifying Offer), deliver such transfer to Qorvo; and
- (c) to hold that consideration in trust for the Defaulting Drag-Along Shareholder.
- 11.5 The Company, through all of its Directors (and each of the Major Investors and the Committed Shareholders, to the extent that they have appointed, or have a right to appoint, any Director), shall procure that the Directors shall forthwith register the transfer of the Shares of the Defaulting Drag-Along Shareholder to Qorvo and the transfer of any and all other Shares held by all other Selling Shareholders shall be completed on the date of such transfer in accordance with this Deed.
- 11.6 The Company warrants to Qorvo that any transfer completed pursuant to this <u>clause 11</u> (the "**Drag-Along Acquisition**") shall be valid and enforceable.
- 11.7 Each Major Investor irrevocably acknowledges and agrees that, if any Major Investor defaults on its obligations to effect the transfer pursuant to this <u>clause 11</u> in accordance with the terms of this <u>clause 11</u> and this Deed, such obligations shall be completed by a Qorvo Officer on behalf of such defaulting Major Investor pursuant to the security power of attorney granted by each Major Investor in accordance with <u>clause 30</u> in favour of any Qorvo Officer to secure the performance of its obligations under this Deed.
- 11.8 Without limiting any other provision of this Deed, each Major Investor warrants to Qorvo and agrees that it shall not effect any Disposal of any of its Shares at any time prior to termination of this Deed if such Disposal would result in such Major Investor no longer having the right (together with the other Major Investor) to give the Drag-Along Notice and effect the Drag-Along Acquisition pursuant to the Articles.

12. WARRANTIES

12.1 Subject to <u>clause 13.2</u>, each of the Warrantors warrants to Qorvo that each of the Warranties set out in <u>Schedule 8</u> is accurate and not misleading in all respects at the date of this Deed (except for such Warranties that speak specifically as of a particular date, which shall be accurate and not misleading as of such date). Each of the Warrantors also warrants to Qorvo that (a) each of the Warranties set out in <u>Schedule 8</u> which are qualified as to materiality, shall be accurate and not misleading in all respects, and (b) each of the Warranties set out in <u>Schedule 8</u> which are not qualified as to materiality, shall be accurate and not misleading in all material respects, in each case at the Completion Date as if repeated immediately before Completion by reference to the facts and circumstances subsisting at that date on the basis that any reference in the Warranties,

whether express or implied, to the date of this Deed is substituted by a reference to the Completion Date (except for such Warranties that speak specifically as of a particular date, which shall be accurate and not misleading as of such date).

- 12.2 The Warrantors undertake to disclose in writing to Qorvo if they know that a material breach of the Warranties has occurred, and in each case, shall do so promptly following such time that it comes to its notice both before or at the time of Completion.
- 12.3 The Warrantors undertake (in the absence of fraud) that if any claim is made against them in connection with the Acquisition of the Shares or Contingent Securities by Qorvo, not to make any claim against any member of the Group or any director, employee, agent or adviser of any member of the Group on whom it may have relied before agreeing to any terms of this Deed or authorising any statement in the Disclosure Letter.
- 12.4 Each of the Warranties shall be construed as a separate and independent warranty and (except where expressly provided to the contrary) shall not be limited or restricted by reference to or inference from the terms of any other Warranty.
- 12.5 Between the execution of this Deed and Completion of the Acquisition, the Company shall notify Qorvo promptly if it becomes aware of a fact or circumstance which constitutes, or which is reasonably likely to constitute, a breach (whether repudiatory or remedial in nature or not) of the Warranties or which would, or which would be reasonably likely to, cause a Warranty to be untrue, inaccurate or misleading if given in respect of facts or circumstances as at Completion.
- 12.6 The remedies of Qorvo in respect of any breach of any of the Warranties shall continue to subsist notwithstanding Completion.

13. REMEDIES AND LIMITATIONS ON LIABILITY

- 13.1 Subject to the terms and conditions of this <u>clause 13</u>, following Completion, the Warrantors severally, and not jointly, shall save, defend, indemnify and hold harmless Qorvo, its Affiliates, and their respective Representatives, successors and assigns (collectively, the "**Indemnitees**") from and against, and shall compensate and reimburse each of the Indemnitees for the Indemnifiable Portion of any and all Losses, asserted against, incurred, sustained or suffered by any of the foregoing as a result of, arising out of or relating to:
 - (a) any breach of the Warranties, as set forth in <u>clause 12.1</u> and <u>Schedule 8</u>;
 - (b) any breach by the Company of the warranties provided in <u>clause 8.13</u> and <u>clause 11</u>;
 - (c) any breach by a Warrantor of the warranty provided by such Warrantor in <u>clause 8.13</u> or <u>clause 9.2</u>, but only to the extent of the Shares and Contingent Securities held by such Warrantor;
 - (d) any breach by a Major Investor of the warranties provided in <u>clause 11.2</u> and <u>clause 11.8</u>;

- (e) any breach by the Company of any covenants made by the Company under this Deed, or any breach by the Warrantors of any covenants made by the Warrantors under this Deed;
- (f) any breach of the covenant made by the Warrantors set forth in <u>Schedule 13</u> (Tax Covenant); and
- (g) the appointment of and services provided by the authorised person under <u>clause 16.5</u>.

13.1.1 Additionally, each Warrantor shall (severally) save, defend, indemnify and hold harmless the Indemnitees from and against, and shall compensate and reimburse each of the Indemnitees for any and all Losses asserted against, incurred, sustained or suffered by any of the Indemnitees as a result of, arising out of or relating to a breach by such Warrantor of the warranty provided by such Warrantor in <u>clause 29</u>.

13.1.2 Notwithstanding anything in this Agreement to the contrary, solely for the purposes of the determination of Losses, the warranties of the Company and the Warrantors in this Deed that are qualified by materiality, Material Adverse Effect or knowledge shall be deemed to be made without such materiality, Material Adverse Effect and knowledge qualifiers. Additionally, no Losses may be claimed under <u>clause 13.1</u> or <u>Schedule 13</u> by an Indemnitee to the extent such Losses are included in the calculation of any adjustment to the Aggregate Value pursuant to <u>clause 7.3</u>.

- 13.2 No Indemnitee shall be entitled to claim that any fact, matter or circumstance causes any of the Warranties (other than the Fundamental Warranties) to be breached or renders any of the Warranties (other than the Fundamental Warranties) misleading as at the date of this Deed to the extent it has been fairly disclosed to Qorvo with sufficient detail to allow Qorvo to identify the nature and scope of the matters, facts and circumstances disclosed in the Disclosure Letter delivered by the Warrantors on the date of this Deed in the absence of any fraud or deliberate concealment on the part of the Warrantors or their respective agents or advisers and only those matters so disclosed in the Disclosure Letter delivered on the date of this Deed shall qualify the Warranties (but not, for the avoidance of doubt, the Fundamental Warranties) as at the date of this Deed.
- 13.3 No information or notice given by any of the Warrantors or any other party to this Deed which is provided or delivered on or after the date of this Deed or other than expressly in this Deed or the Disclosure Letter, shall limit or qualify the Warranties (or the Fundamental Warranties) made on and as of the date of this Deed in any manner whatsoever.
- 13.4 The Warrantors may update the Disclosure Letter during the Diligence Period provided that the Disclosure Letter must be delivered at least [*****] Business Days prior to the conclusion of the Diligence Period. Any such updated or supplemental Disclosure Letter shall be referred to as the "**Final Disclosure Letter**." No Indemnitee shall be entitled to claim that any fact, matter or circumstance causes any of the Warranties (other than the Fundamental Warranties (excepting Section 2.2 (Capital Structure))) to be breached or renders any of the Warranties (other than the Fundamental Warranties (excepting Section 2.2 (Capital Structure))) misleading as at Completion of the Acquisition to the extent it has been fully, fairly and specifically disclosed in the Final Disclosure Letter delivered by the Warrantors at least [****] Business Days prior to

conclusion of the Diligence Period in the absence of any fraud or deliberate concealment on the part of the Warrantors or their respective agents or advisers and only those matters so disclosed in the Final Disclosure Letter delivered at least [*****] Business Days prior to conclusion of the Diligence Period shall qualify the Warranties (but not, for the avoidance of doubt, the Fundamental Warranties other than Section 2.2 (Capital Structure)) as at Completion of the Acquisition.

- 13.5 No liability shall attach to the Warrantors in respect of claims under the Warranties or the Tax Covenant, as the case may be, if and to the extent that the limitations referred to in this <u>clause 13</u> apply, in the absence of any fraud or deliberate concealment on the part of the Warrantors or their respective agents or advisers.
- 13.6 The warranties (including the Warranties) of the Company and the Warrantors contained in this Deed and in the Offer Document (as given by each Warrantor pursuant to such Warrantor's Form of Acceptance) and each Deed of Release shall survive the Completion until the date that is 18 months after the Completion Date; provided, that the warranties set forth in Section 1 (Organisation, Standing and Power), Section 2 (Capital Structure), Section 3 (Due Authorisation) and Section 23 (Brokers) of the Warranties (collectively referred to herein as the "Fundamental Warranties"), and any warranty in the case of fraud or deliberate concealment, shall survive indefinitely; and provided, that the Tax Warranties shall survive as provided in Schedule 13.

13.7 Claim Procedures.

(a) <u>Claims</u>. None of the Warrantors shall have any liability under this Deed to any Indemnitee unless notice (a "Claim Notice") of an actual or threatened claim, or of discovery of any facts or circumstances that Qorvo or another Indemnitee reasonably believes may result in a claim (an "Indemnification Claim"), hereunder is provided to the Securityholder Agent, in writing by Qorvo (on behalf of itself or any other Indemnitee), and with respect to the warranties, prior to the expiration of the survival periods set forth in <u>clause 13.6</u> for such warranty. Any Claim Notice shall describe the Indemnification Claim in reasonable detail and shall indicate the estimated amount of Losses if then estimable (any such amount as delivered in a Claim Notice or later determined by an Indemnitee, the "Claim Amount"). Additionally, each Indemnification Claim shall be made in accordance with this <u>clause 13</u>, the Escrow Agreement and, if applicable, <u>Schedule 13</u>. Following the delivery of any Claim Notice, the Securityholder Agent or its representatives and agents shall be provided with such information regarding such Claim Notice as they may reasonably require for the purposes of investigating or resolving any disputes or responding to any matters or inquiries raised in such Claim Notice and shall be afforded reasonable access, upon notice and at reasonable times during business hours and without disruption to the Company's or Qorvo's business, to employees of the Company or Qorvo in connection with such investigation and resolution.

- (b) Objection to Claims. If the Securityholder Agent wishes to object to any Indemnification Claim or portion thereof or the amount of such Claim set forth in a Claim Notice, the Securityholder Agent must deliver a written objection to Qorvo and the Escrow Agent, within [*****] Business Days after receipt by the Securityholder Agent of such Claim Notice expressing such objection and explaining in reasonable detail and in good faith the basis therefor. Following receipt by Qorvo of the Securityholder Agent's written objection, if any, Qorvo and the Securityholder Agent will promptly, and within [*****] Business Days of the date Qorvo receives the Securityholder Agent's written objection, meet to discuss the rights of the respective parties with respect to each Indemnification Claim that is the subject of such written objection. If the Securityholder Agent and Oorvo (on behalf of itself or any other Indemnitee) should agree as to the amount of the Losses for an Indemnification Claim, (i) a memorandum setting forth such agreement on the amount of Losses for such Indemnification Claim will be prepared and executed by Qorvo (on behalf of any other Indemnitee, if applicable) and the Securityholder Agent and (ii) as promptly as practicable, and in any event within [*****] Business Days following the execution of such memorandum by both parties, the Warrantors shall pay, in accordance with clause 13.13, the aggregate U.S. Dollar amount of the Losses payable to Qorvo or any other Indemnitee, as finally determined and set forth in such memorandum and the Warrantors shall be deemed to have irrevocably waived any further right to object to such Indemnification Claim and to have agreed that Losses so agreed to by the Securityholder Agent and Qorvo are indemnifiable hereunder. In the event that the Securityholder Agent and Qorvo do not prepare and sign such a memorandum or such memorandum does not address in full the written objections timely delivered within [*****] Business Days of receipt by Qorvo from the Securityholder Agent of the written objection, then Qorvo may at any time commence an Action to resolve such dispute and enforce its rights with respect thereto, including any applicable payment from the Escrow Account.
- (c) <u>No Objections to Claim</u>. If Qorvo does not receive an objection from the Securityholder Agent with respect to any Indemnification Claim set forth in a Claim Notice by the end of the [*****] Business Day period referred to in <u>subclause (b)</u> above, (i) the Warrantors and the Securityholder Agent shall be deemed to have irrevocably waived any right to object to such Indemnification Claim and to have agreed that Losses in the amount of the applicable Claim Amount are indemnifiable hereunder and (ii) as promptly as practicable and within [*****] Business Days following expiration of such period, subject to <u>clause 13.13</u>, below, the Warrantors shall pay such Claim Amount to Qorvo in accordance with <u>clauses 13.13</u>.
- 13.8 Following the giving of a Claim Notice under <u>clause 13.7:</u>
 - (a) where the Indemnification Claim described in a Claim Notice:
 - (i) involves a claim against Qorvo and/or any other Indemnitee; and
 - (ii) is either (A) contingent or (B) otherwise not capable of being quantified at the time of such notice,

then the Warrantors shall cease to be liable for such Indemnification Claim if legal proceedings in respect of such Indemnification Claim have not been commenced within [*****] months of such Indemnification Claim ceasing to be contingent or becoming capable of being quantified; or

- (b) where the Indemnification Claim described in a Claim Notice is not contingent and is quantifiable, then the Warrantors shall cease to be liable for such Indemnification Claim if legal proceedings in respect of such Indemnification Claim have not been commenced within [*****] months of service of delivery of the Claim Notice to the Securityholder Agent.
- 13.9 Notwithstanding anything to the contrary contained in this Deed, and except as provided in <u>clause 13.10</u> below:
 - (a) the Warrantors shall not be liable for any Indemnification Claim arising out of or relating to a claim made pursuant to <u>clause 13.1(a)</u> unless and until the aggregate amount of which may be recovered from the Warrantors under such clause equals or exceeds USD \$[*****], in which case the Warrantors shall be liable for the full amount of such Indemnification Claim from the first dollar thereof; provided, however, that the foregoing clause shall not apply to any Indemnification Claim arising out of or relating to the inaccuracy or breach of any Fundamental Warranty, or to any warranty (including the Warranties), in the event of fraud or deliberate concealment;
 - (b) the maximum aggregate amount which may be recovered from a Warrantor arising out of or relating to any Indemnification Claim made pursuant to <u>clause 13.1(a)</u> shall be an amount equal to such Warrantor's pro rata portion of the Escrow Account; provided, however, that the foregoing clause shall not apply to any Indemnification Claim arising out of or relating to the inaccuracy or breach of any Fundamental Warranty, or to any warranty (including the Warranties), in the event of fraud or deliberate concealment; provided, however, that the aggregate amount to be paid by the Warrantors for claims involving breach of a warranty (including the Warranties) in the event of fraud or deliberate concealment, shall not exceed the portion of the Aggregate Value actually distributed to the Warrantors; provided, further, that the aggregate amount to be paid by a particular Warrantor for a claim involving breach of a warranty (including the Warranties) in the event of fraud or deliberate concealment; amount to be paid by a particular Warrantor for a claim warranty (including the Warrantor; and
 - (c) the maximum aggregate amount which may be recovered from a Warrantor arising out of or relating to any Indemnification Claim made pursuant to <u>clause 13.1(b)</u> through (g) and <u>clause 13.1.1</u> or in connection with a breach of any Fundamental Warranty, shall be an amount equal to that portion of the Aggregate Value actually distributed to such Warrantor (including such Warrantor's pro rata portion of the Escrow Account); provided, however, that the foregoing clause shall not apply, and liability shall not be capped, as to a Warrantor in the event of fraud or deliberate concealment by such Warrantor.

- 13.10 Other than matters fully, fairly and specifically disclosed in the Disclosure Letter, no other information of which Qorvo or any of its Affiliates or Representatives may have knowledge (whether actual, constructive or imputed) prevents or limits an Indemnification Claim made by Qorvo for breach of any of the Warranties or for any other Indemnification Claim arising out of or relating to any breach of any warranty, undertaking or covenant contained in this Deed.
- 13.11 If in respect of or in connection with any breach of any of the Warranties or any facts or matters warranted not being true and being misleading any amount payable by the Warrantors or from the Escrow Account pursuant to this Deed (whether under this <u>clause 13</u> or otherwise) (a) is subject to Tax in the hands of the recipient or the person beneficially entitled to such payment (the "**Recipient**"), or (b) results in a Tax benefit to the Recipient as a result of such Loss (which benefit is received in the same year as, or a prior year to, the payment of the indemnification amount under this <u>clause 13</u>), the Warrantors shall, in the event of <u>clause 13.11(a)</u> pay, or such recipient shall be entitled to receive, as applicable, additional amounts so as to ensure that the net amount received by the Recipient on an after-tax basis is equal to the full amount payable by Warrantors or from the Escrow Account under this Deed, and in the event of <u>clause 13.11(b)</u>, reduce the amount payable by the Warrantors or from the Escrow Account under this deed by such benefit amount.

13.12 Third Party Claims.

- (a) If any Third Party provides written notice to Qorvo or any of its Affiliates with respect to any Third Party Claims, which Qorvo believes in good faith, may give rise to an Indemnification Claim by Qorvo against the Escrow Account, then Qorvo shall give notice to the Securityholder Agent within [*****] days after Qorvo has received any such Third Party Claim setting forth such material information with respect to the Third Party Claim as is reasonably available to Qorvo, including the estimated amount, if reasonably practicable, of the Loss that has been or may be sustained, and copies of all material written evidence; *provided, however*, that no delay or failure on the part of Qorvo in notifying the Securityholder Agent shall relieve the Securityholder Agent and the Warrantors from their obligations hereunder unless the Securityholder Agent and the Warrantors are thereby materially prejudiced (and then solely to the extent of such material prejudice).
- (b) The Securityholder Agent (on behalf of the Warrantors) shall have the right to participate in, or, by providing written notice delivered to Qorvo within [*****] days after receiving Qorvo's notice of such claim, assume the defence thereof by the Securityholder Agent's own counsel (which counsel shall be reasonably acceptable to Qorvo), at the sole expense of the Warrantors, independent of the then-remaining Escrow Account, and Qorvo and its Affiliates shall cooperate in good faith in such defence; provided that, if the Securityholder Agent (on behalf of the Warrantors) assumes any such defence: (A) the Warrantors shall be obligated to indemnify and hold harmless Qorvo with respect to any and all Losses in connection with such Third Party Claim, (B) the limitations on the Warrantors' liability set forth in this Deed shall not apply with respect to such Third Party Claim and (C) the Securityholder Agent (on behalf of the Warrantors) shall acknowledge

such indemnification obligation in writing to Qorvo; provided, however, that the Securityholder Agent (on behalf of the Warrantors) shall not have the right to assume the defence of such claim if:

- (i) Qorvo has reasonably determined that Losses which may be incurred as a result of the Third Party Claim may exceed, either individually or when aggregated with all other Third Party Claims, the total dollar value of the Escrow Account;
- (ii) the Third Party Claim may involve an injunction or other equitable relief;
- (iii) the Third Party Claim would impose criminal liability or damages or any other claim other than civil monetary damages at law;
- (iv) the Third Party Claim alleges infringement or misappropriation of, or challenges the validity of, Intellectual Property rights; or
- (v) the Third Party Claim involves a customer or supplier of Qorvo, Parent the Company or any of its Subsidiaries.
- (c) If the Securityholder Agent (on behalf of the Warrantors) assumes any such defence, the Securityholder Agent shall conduct the defence of the Third Party Claim actively and diligently. Neither the Securityholder Agent nor the Warrantors shall compromise or settle such Third Party Claim or consent to entry of any judgment in respect thereof without the prior written consent of Qorvo, which consent shall not be unreasonably withheld or delayed.
- (d) In the event that the Securityholder Agent (on behalf of the Warrantors) assumes the defence of the Third Party Claim in accordance with <u>clause 13.12(b)</u>, Qorvo may retain separate counsel and participate in the defence of the Third Party Claim, but the fees and expenses of such counsel shall be at the expense of Qorvo, unless Qorvo shall reasonably determine that (i) there is a conflict of interest between or among Qorvo on the one hand, and the Securityholder Agent and the Warrantors on the other hand, or (ii) the Securityholder Agent or the Warrantors have failed to vigorously prosecute or defend, with respect to such Third Party Claim, in which case the reasonable fees and expenses of such counsel will be paid out of the Escrow Account. Qorvo will cooperate in the defence of the Third Party Claim and will provide full access to documents, assets, properties, books and records reasonably requested by the Warrantors or Securityholder Agent and material to the claim and will make available all officers, directors and employees reasonably requested by the Warrantors or Securityholder Agent for investigation, depositions and trial.
- (e) If the Securityholder Agent fails or elects not to assume the defence of Qorvo or Parent against such Third Party Claim, which the Securityholder Agent (on behalf of the Warrantors) had the right to assume under <u>clause 13.12(b)</u>, Qorvo shall have the right to undertake the defence and Qorvo shall not compromise or settle such Third Party Claim or consent to entry of any judgment in respect thereof without the prior written consent of

the Warrantors, which consent shall not be unreasonably <u>withheld</u> or delayed. If the Securityholder Agent (on behalf of the Warrantors) is not entitled to assume the defence of Qorvo against such Third Party Claim pursuant to <u>clause 13.12(b)</u>, Qorvo shall have the right to undertake the defence, consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim in any manner it may deem appropriate (and Qorvo need not consult with, or obtain any consent from, the Warrantors in connection therewith); provided, however, that, except with the written consent of the Securityholder Agent on behalf of the Warrantors, no settlement of any such claim or consent to entry of any judgment with respect to such Third Party Claim shall alone be determinative of the validity of the Third Party Claim against the Escrow Account or personally against the Warrantors. In each case, Qorvo shall conduct the defence of the Third Party Claim actively and diligently, and the Warrantors and Securityholder Agent will cooperate with Qorvo and provide full access to documents, assets, properties, books and records in its possession reasonably requested by Qorvo for investigation, depositions and trial.

13.13 Payment of Indemnification Claims; Escrow Account:

(a) The Escrow Agent shall not distribute any amounts from the Escrow Account until a Loss is finally determined by the parties to this Deed pursuant to the terms of clause 13.7 or 13.12 and the Escrow Agreement. If an Indemnifiable Claim is contingent or otherwise not capable of being quantified at the time the Claim Notice is sent, Qorvo shall as soon as reasonably practicable, but in any event, prior to the end of the Escrow Period, provide the Escrow Agent with its reasonable good faith estimate of the Loss relating to such Indemnification Claim, and such amount shall be retained by the Escrow Agent and not distributed until the amount of the Loss is finally determined hereunder. Notwithstanding the foregoing, so long as Qorvo provides its reasonable good faith estimate as soon as reasonably practicable shall relieve the Securityholder Agent and the Warrantors from their obligations under this Deed or the Escrow Agent's obligation to retain such amount as provided in this clause 13.13, unless the Securityholder Agent and the Warrantors are thereby materially prejudiced (and then solely to the extent of such material prejudice).

(b) Once the amount of a Loss is finally determined by the parties to this Deed pursuant to the terms of clause 13.7 or 13.12 and the Escrow Agreement, the Warrantors shall be responsible only for the Indemnifiable Portion of such Loss, which shall be satisfied from the Escrow Account, to the extent of the amount then held in the Escrow Account, before seeking to recover any amounts directly from the Warrantors. To the extent that the amount of the Escrow Account (including the interest or other income accrued thereon) (i) has been distributed, pro rata, to the Warrantors following the date that is 18 months from the Completion Date, (ii) has been exhausted or (iii) if there are pending unresolved claims against the Warrantors, would be exhausted if one or more pending claims against the Warrantors are resolved in favour of Qorvo, then subject to the limitations set forth in

this <u>clause 13</u>, and only for claims made pursuant to <u>clause 13.1(b)</u>, <u>clause 13.1(c)</u>, <u>clause 13.1(d)</u>, <u>clause 13.1(e)</u>, <u>clause 13.1(g)</u> or <u>clause 13.1.1</u> such Indemnification Claim may be made directly against the Warrantors, severally and not jointly, in accordance with their pro rata share, as amongst each other, and such amounts shall be paid promptly (but in no event later than [*****] Business Days after the applicable payment obligation accrues) by the applicable Warrantors by wire transfer of U.S. Dollars in immediately available funds to such account or accounts as may be designated in writing by Qorvo. If any amount is not paid in accordance with the foregoing, Purchaser may seek enforcement of the obligation with respect to the relevant Indemnification Claim in any court available therefor or in accordance with the terms of <u>clause 44</u>. Any amounts paid to Qorvo in respect of any Indemnification Claim asserted on behalf of an Indemnitee other than Qorvo will be received by Qorvo on behalf of such other Indemnitee.

- (c) Upon the termination of the Escrow Account pursuant to the terms of the Escrow Agreement, the Escrow Agent shall pay any amounts remaining in the Escrow Account and not subject to pending Indemnification Claims to the Payments Administrator for further distribution to the Warrantors. Payment of such amounts to the Warrantors shall be made in accordance with the terms of the Escrow Agreement.
- 13.14 Payments by the Warrantors pursuant to <u>clause 13.1</u> or <u>Schedule 13</u> in respect of any Loss shall be limited to the amount of any liability or damage that remains after deducting therefrom any insurance proceeds and any indemnity, contribution or other similar payment actually received by Qorvo or its Affiliates in respect of any such claim, less any related costs and expenses (including the aggregate cost of pursuing any related insurance claims and any related increases in insurance premiums or other chargebacks). Qorvo shall use its reasonable endeavours to seek to recover insurance proceeds in connection with making a claim under this <u>clause 13</u> or <u>Schedule 13</u>.
- 13.15 Subject to <u>clause 34.4</u>, the parties (other than the Securityholder Agent, with respect to its rights and indemnities under <u>clause 46</u>) acknowledge and agree that, following Completion, their sole and exclusive remedy with respect to any and all claims (other than claims arising from fraud or deliberate concealment on the part of a party hereto in connection with the transactions contemplated by this Deed) for any breach of any warranty, covenant, agreement or obligation set forth herein or otherwise relating to the subject matter of this Deed, shall be pursuant to the indemnification provisions set forth in <u>Schedule 13</u> and this <u>clause 13</u>. Nothing in this <u>clause 13.15</u> or in other provision of this Deed shall limit Qorvo's or Company's right or ability to pursue any causes of action or claims either party may have prior to Completion, and, accordingly, prior to Completion, Qorvo and the Company shall each have the right and ability to pursue any cause of action or claim it may have, including seeking any remedy described in <u>clause 34.4</u> or seeking damages (including all costs and expenses incurred by such party in connection with this Deed and the Offer) arising from any breach of any warranty, covenant, agreement or obligation set forth in this Deed or otherwise relating to the subject matter of this Deed.

14. TAX COVENANT

On Completion, Schedule 13 (Tax Covenant) shall become effective.

15. NON-SOLICITATION

- 15.1 In this <u>clause 15</u>, "**Third Party Negotiations**" means any discussions or negotiations between the Company, the Major Investors, the Executive Optionholders, the Committed Shareholders or any of their respective officers, agents, senior management, employees, advisers or other Representatives acting on their behalf and anyone other than Qorvo (or a company which is a subsidiary of Qorvo) or any of its officers, agents employees, advisers, or other Representatives (the "**Third Party**") relating to a corporate activity (the "**Corporate Activity**"), being:
 - (a) the disposal (whether by way of sale, offer, transfer or otherwise) of part of, or the whole of, the issued share capital (or any interest in such share capital) of the Company, other than sales, offers or transfers specifically permitted by the terms of this Deed (including without limitation Section 19 hereof); or
 - (b) the disposal (whether by way of sale, offer, transfer or otherwise) of all, or any part of, the business or (other than in the ordinary course of trading) the assets of the Company.
- 15.2 The obligations of the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders under this <u>clause 15</u> are given severally and not jointly and any reference to the Major Investors, the Executive Optionholders or the Committed Shareholders includes any one or more of them.
- 15.3 In consideration of Qorvo entering into this Deed, the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders grant Qorvo a period of exclusivity on and from the date of this Deed and throughout the Option Period and until the earlier of (i) Completion of the Acquisition, and (ii) termination of this Deed (as the case may be, the "**Exclusivity Period**"), for Qorvo to conduct the due diligence and to decide whether or not to proceed with the Offer and the Acquisition.
- 15.4 Immediately on signing this Deed, each of the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall terminate, or procure the termination of, any Third Party Negotiations currently taking place involving such party's officers, agents, senior management, employees, advisers or other Representatives and shall communicate to the relevant Third Party that the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders will refuse to entertain any Corporate Activity being discussed as part of such Third Party Negotiations.
- 15.5 During the Exclusivity Period, the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders undertake that they shall not, and, solely with respect to the Company, the Major Investors and the Executive Optionholders, to the extent reasonably possible, that they shall procure that no Selling Shareholder shall, directly or indirectly:

- (a) induce, solicit, procure or otherwise encourage a Corporate Activity from a Third Party;
- (b) enter into, re-start, solicit, initiate or otherwise participate in any Third Party Negotiations;
- (c) seek or encourage any approach that might be reasonably expected to lead to Third Party Negotiations;
- (d) enter into any heads of terms, agreement, arrangement or understanding (whether or not legally binding) with any Third Party relating to a Corporate Activity, whether pursuant to any Third Party Negotiations or otherwise; or
- (e) supply or otherwise disclose any information about the Group to a Third Party that wishes to enter into Third Party Negotiations to the extent the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders have knowledge of such desire.
- 15.6 The Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall undertake to procure that their officers, senior management, employees, agents, advisers and other Representatives acting on their behalf comply with the undertakings in this <u>clause 15</u> as if they were the Major Investors, the Executive Optionholders or the Committed Shareholders (as the case may be).
- 15.7 Each of the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall promptly notify Qorvo in writing if, during the Exclusivity Period, they receive an indication from any Third Party (including any Third Party with whom a Selling Shareholder or the Company was having Third Party Negotiations prior to the date of this Deed) that such Third Party may wish to have discussions with a view to a possible Corporate Activity.

16. RECOMMENDATION BY THE COMPANY DIRECTORS

- 16.1 Each of the Major Investors, the Executive Optionholders and the Committed Shareholders irrevocably undertake to Qorvo to procure (including by the exercise of any of their rights under the Articles of Association and the Amended & Restated Investment Agreement and by the exercise of any rights attaching to any of their Shares or Contingent Securities) that each Director shall, subject always to the statutory duties of the Directors, recommend the Offer to each and every person who owns Shares or Contingent Securities of the Company.
- 16.2 Subject always to the statutory duties of the Directors, in furtherance of the undertaking given in <u>clause 16.1</u>, each of the Directors shall be required to:
 - (a) execute a unanimous written resolution of the Directors on the date of this Deed recommending the Offer to Shareholders as a contingent offer that is fair and reasonable on its contingent terms;

- (b) on exercise of the Option by Qorvo, execute any other document required to be signed by the Directors in relation to the Offer in accordance with Applicable Law.
- 16.3 Each of the Major Investors, the Executive Optionholders and the Committed Shareholders acknowledges, agrees and undertakes for the benefit of the Directors and Qorvo that:
 - (a) in approving the entering into of this Deed by the Company, the Directors have satisfied their statutory duties;
 - (b) in executing the unanimous written resolution of the Directors on the date of this Deed recommending the Offer to Shareholders as a contingent offer that is fair and reasonable on its contingent terms, the Directors have satisfied their statutory duties;
 - (c) any subsequent recommendation of the Offer by the Directors to the Shareholders in accordance with <u>clause 16.1</u> shall be in accordance with the statutory duties of the Directors; and
 - (d) no Major Investor, Executive Optionholder or Committed Shareholder shall have, and each Major Investor, Executive Optionholder and Committed Shareholder hereby irrevocably waives, any right to claim against the Directors for any breach of statutory duty in connection with the entry into this Deed by the Company, the unanimous written resolution of the Directors recommending the Offer to Shareholders on the date of this Deed or any subsequent recommendation of the Offer by the Directors to the Shareholders.
- In the event that any Director or Qorvo determines, within [*****] Business Days of the Option Notice being served by 16.4 Qorvo, that, for any reason, there is any possibility that any Other Shareholder or any other person who holds Shares or Contingent Securities may claim that any subsequent recommendation of the Offer by the Directors to the Shareholders in accordance with clause 16.1 may be in breach of the statutory duties of the Directors, the Directors and Qorvo shall each, at their discretion, be entitled to seek, or in the case of Oorvo require the Directors to seek, the consent, approval and direction of the Shareholders to the Directors to recommend the Offer to the Shareholders as fair and reasonable and in the best interests of the Company and the Shareholders, such consent, approval and direction to be submitted to the Shareholders of the Company as a special resolution of the Company. Each of the Major Investors, Executive Optionholders and the Committed Shareholders irrevocably undertakes to the Company and to Qorvo to vote in favour of such special resolution, and to do all such things and execute all such documents so as to ensure that such consent, approval and direction shall pass as a special resolution of the Company as soon as reasonably practicable following service of the Option Notice to enable Qorvo to make the Offer on the basis of a unanimous recommendation by the Directors (including, for the avoidance of doubt, any consent to short notice of a meeting of the Shareholders of the Company). Each of the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders acknowledges and agrees that it is the intention of the parties that any special resolution of the Shareholders of the Company passed pursuant to this clause 16.4 shall cure any breach by the Directors of their statutory duties pursuant to any action taken in connection with this Deed, the Offer or the Acquisition, shall absolve each

Director of any liability for the same, and shall serve as a ratification and approval of any and all actions undertaken by the Directors in connection with this Deed, the Offer or the Acquisition as being in accordance with the statutory duties of the Directors.

- 16.5 In the event that the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders cannot procure the unanimous recommendation of the Offer by the Directors (subject to such Directors' statutory duties) on the exercise of the Option by Qorvo:
 - (a) Qorvo shall be entitled, as necessary to comply with Applicable Law, to appoint a financial intermediary as an authorised person under the Financial Services and Markets Act 2000 to make the Offer on behalf of Qorvo; and
 - (b) in the event that Qorvo is forced to appoint an authorised person under <u>clause 16.5(a)</u> in order to make the Offer and complete the Acquisition due to the failure by the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders to procure the unanimous recommendation of the Offer by the Directors, the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders, shall be required to notify Qorvo of such failure as soon as reasonably practicable after such failure and in any event within [*****] Business Days of such failure. Following receipt of such notice, the requirement under <u>clause 4.1(g)</u> for the Offer to be made by the Company within [*****] Business Days of receipt of the Offer Confirmation Notice shall be temporarily suspended and Qorvo shall be granted a period of up to [*****] Business Days from the date of the notice delivered under this <u>clause 16.5(b)</u> to arrange for the appointment of a financial intermediary as an authorised person under the Financial Services and Markets Act 2000, and, upon finalisation of such appointment, the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall provide all assistance required to enable Qorvo to proceed with the Offer in accordance with the remaining terms and conditions of this Deed.

17. ACCESS TO INFORMATION AND ACCOUNTS

- 17.1 The Company shall provide Qorvo with complete access (including for inspection and copying) at all reasonable times to the Representatives, properties, offices, plants and other facilities, books and records of the Company and each of its Subsidiaries, and shall furnish Qorvo with such financial, operating and other data and information as Qorvo may reasonably request save in respect of the Company's intellectual property and Projects not in the public domain, save in respect of the Company's intellectual property and Projects not in the public domain, to which data, information and materials Qorvo shall be provided access from the beginning of the Diligence Period through the Completion of the Acquisition; provided, that Qorvo shall be provided with access to data and information regarding the Company's intellectual property and Projects not in the public domain from the beginning of the Option Period through Completion of the Acquisition.
- 17.2 All records of the Company shall be retained for a period of at least six (6) years from the end of the year to which such record relates or such longer period as may be required by law.

- 17.3 The Company shall provide to Qorvo audited accounts of the Group for each Accounting Period promptly following their approval by the Directors.
- 17.4 All confidential and proprietary information provided to Qorvo by the Company shall be "Confidential Information" for the purposes of this Deed and the obligations of <u>clause 39</u> shall apply to Qorvo in respect of such information.

18. RESTRICTIONS ON DEALING WITH SHARES

- 18.1 No Major Investor, Executive Optionholders or Committed Shareholder shall make a Disposal, and the Company, the Major Investors, the Executive Optionholders and the Committed Shareholders shall procure that any Disposal by any Other Shareholder or Contingent Shareholder shall be prohibited, except a Disposal of a Share or Contingent Security which is expressly approved in writing by Qorvo; provided that:
 - (a) a Selling Shareholder (other than the Major Investors) or Contingent Shareholder may transfer any Share or Contingent Security in accordance with <u>clause 19</u>; and
 - (b) this restriction shall not prevent the transmission of any Share or Contingent Security by operation of law on testate to any beneficiary on the death of any Selling Shareholder or Contingent Shareholder.
- 18.2 Each of the Major Investors, the Executive Optionholders and the Committed Shareholders undertakes to exercise all rights of such persons under the Articles of Association, the Amended & Restated Investment Agreement or otherwise to prevent the Disposal of any Share or Contingent Security or any legal or beneficial interest in a Share or Contingent Security by any Selling Shareholder or Contingent Shareholder in contravention of <u>clause 18.1</u>.
- 18.3 No transfer of Shares or Contingent Securities permitted by this Deed or otherwise purported to be made shall be valid unless the transferee has executed a Deed of Adherence to this Deed as a Committed Shareholder (or Executive Optionholder, whichever applies). Any purported transfer that does not comply with this <u>clause 18.3</u> shall be void ab initio and the Company shall not recognise any transfer of any Shares or Contingent Securities in contravention of this <u>clause 18.3</u>.

19. PERMITTED TRANSFERS

- 19.1 Subject to the Articles of Association and the conditions set forth therein, a Selling Shareholder (other than the Major Investors) or Contingent Shareholder may at any time transfer any of the Shares or Contingent Securities held by it in the following ways and to the following persons (each a "**Permitted Transferee**"):
 - (a) a Selling Shareholder or Contingent Shareholder may transfer any of his Shares or Contingent Securities to the trustees of his Family Trust;

- (b) the trustees of a Family Trust of a Selling Shareholder or Contingent Shareholder may transfer any Share or Contingent Security held by them in that capacity to the new trustees of that Family Trust;
- (c) Shares or Contingent Securities may be transferred by a Selling Shareholder or Contingent Shareholder to a person who is to hold such Shares or Contingent Securities as his nominee but any transfer by such nominee shall be subject to the same restrictions as though it were a transfer by the Selling Shareholder or Contingent Shareholder itself;
- (d) Shares or Contingent Securities may be transferred by a nominee to the beneficial owner of such Shares or Contingent Securities or to another nominee of the same beneficial owner;
- (e) Shares or Contingent Securities held by or on behalf of any Selling Shareholder (where such Selling Shareholder constitutes a fund managed professionally for investment purposes by persons managing, or advising in respect of investments of, various other such investment funds) may be transferred to:
 - (i) any persons managing, or advising in respect of, the investment of the funds held by the Selling Shareholder or Affiliates of the Selling Shareholder or within the same wholly-owned group as any person managing, or advising in respect of, the investment of funds held by the Selling Shareholder or Affiliates of the Selling Shareholder or to a nominee or trustee for such persons;
 - (ii) a nominee or trustee for the funds held by the Selling Shareholder (or such fund itself where the transferor is its manager or investment adviser) and any Shares or Contingent Securities held by a nominee or trustee for a fund may be transferred to that fund or to another nominee or trustee for such fund but such transfer by such nominee or trustee shall be subject to the same restrictions as applied when the Shares or Contingent Securities were transferred to such nominee or trustee;
 - (iii) another fund which is Controlled, managed or advised by the same manager or adviser or by another member of the same wholly-owned group as such manager or adviser or to a nominee or trustee for such a fund; or
 - (iv) any partner of, or investor, shareholder or unitholder, in such fund;
- (f) Shares or Contingent Securities held by or on behalf of any Selling Shareholder which is a corporate entity may be transferred to such Selling Shareholder's Affiliates; and
- (g) any person to whom Shares or Contingent Securities have been transferred by any person pursuant to this clause 19.1 may transfer all or any Shares or Contingent Securities back to the original transferor or to any other person to whom the original transferor, if it still held such Shares or Contingent Securities, would have been able to transfer them under this clause 19.1;

provided, that, in case of any of the foregoing, any transferee to whom Shares or Contingent Securities are transferred shall, as a condition to and prior to such transfer, execute a Deed of Adherence to accede to this Deed as a Warrantor, Major Investor, Executive Optionholder, Committed Shareholder, Contingent Shareholder and/or Other Shareholder, as applicable, in the same capacity or capacities as the transferor.

19.2 The transferor and the transferee of any Share or Contingent Security transferred under <u>clause 19.1</u> shall each provide to the Company at their own expense any information and evidence reasonably requested in writing for the purpose of determining whether the transfer to the proposed transferee complies with the terms of this <u>clause 19.2</u>.

20. CONSENT TO TRANSFER FOR THE PURPOSES OF THE ARTICLES

This Deed constitutes the irrevocable written consent of each of the Major Investors, the Executive Optionholders and each Committed Shareholder for the purposes of the Articles of Association and the Amended & Restated Investment Agreement to any transfer of Shares or Contingent Securities which is permitted, required or made in accordance with the terms of this Deed, including, without limitation, all transfers to Qorvo.

21. GENERAL UNDERTAKINGS

- 21.1 Each Major Investor, each Executive Optionholder and each Committed Shareholder undertakes to Qorvo that it will:
 - (a) exercise its voting rights and other rights as a member of the Company in order (insofar as it is able to do so through the exercise of such rights) to give full effect to the terms of this Deed;
 - (b) to the extent permitted by Applicable Law, procure that any Director appointed by it from time to time shall exercise such Director's: (i) voting rights; (ii) rights to call meetings of the Board; (iii) rights to bring matters to a vote of the Board; and (iv) other powers and authorities, in each case, in order (insofar as they are able to do so through the exercise of such rights, powers and authorities) to give full effect to the terms of this Deed and the rights and obligations of the parties as set out in this Deed, including causing the Company to satisfy each of its obligations under this Deed; and
 - (c) at all times act in good faith towards Qorvo and use all reasonable endeavours to ensure that the terms of this Deed are observed.
- 21.2 Qorvo undertakes that it will at all times act in good faith towards each Major Investor, each Executive Optionholder, each Committed Shareholder and the Company and use all reasonable endeavours to ensure that the terms of this Deed are observed; provided, however, nothing in this <u>clause 21.2</u> or any other provision of this Deed shall be deemed to impose any obligation on Qorvo to exercise the Option, and any determination to exercise or not exercise the Option shall be in Qorvo's sole and absolute discretion.

- 21.3 On or following the Completion Date, Qorvo shall provide each Major Investor, Committed Shareholder and Executive Optionholder a list of the names of the Key Employees and senior employees of the Company or any member of the Group (the "**Designated Individuals**"), and each Major Investor, Committed Shareholder and Executive Optionholder undertakes and covenants with Qorvo that following receipt of such notice, such party shall not, whether directly or indirectly, on such party's own behalf or on behalf of or in conjunction with any other person, firm, company or other entity, for the period of [*****] months following the Completion Date, solicit, interfere with, or endeavour to cause any of the Designated Individuals to leave his or her employment or induce or attempt to induce any such employee to terminate or breach his or her employment agreement; provided, however, that nothing in this section shall prohibit the use of a general solicitation in a publication or by other means not targeted at the employees of the Company or any member of the Group.
- 21.4 Qorvo agrees that all rights to indemnification, advancement of expenses and exculpation by the Company now existing in favour of each person who is now, or has been at any time prior to the date hereof or who becomes prior to the Completion Date, an officer of director of the Company (a "D&O Indemnified Party"), as provided in the Articles of Association, or pursuant to any other contracts, agreements or deeds with such parties, shall survive and remain in full force and effect in accordance with their terms.
- 21.5 For six years after the Completion Date, to the fullest extent permitted under Applicable Law, the Company shall, and Qorvo shall procure that the Company shall, indemnify, defend and hold harmless each D&O Indemnified Party against all Losses arising in whole or in part out of actions or omissions in their capacity as such occurring at or prior to the Completion Date (including in connection with the transactions contemplated by this Deed), and shall reimburse each D&O Indemnified Party for any legal or other expenses reasonably incurred by such D&O Indemnified Party in connection with investigating or defending any such Losses as such expenses are incurred, subject to the Company's receipt of an undertaking by such D&O Indemnified Party to repay such legal and other fees and expenses paid in advance if it is ultimately determined in a final and non-appealable judgment of a court of competent jurisdiction that such D&O Indemnified Party is not entitled to be indemnified under Applicable Law; provided, however, that the Company will not be liable for any settlement effected without the Company's prior written consent (which consent shall not be unreasonably withheld or delayed).
- 21.6 Prior to Completion, the Company shall obtain and fully pay for "tail" insurance policies with a claims period of at least six (6) years from the Completion Date with at least the same coverage and amount and containing terms and conditions that are not less advantageous to the directors and officers of the Company as the Company's existing policies with respect to claims arising out of or relating to events which occurred before or at the Completion Date (including in connection with the transactions contemplated by this Deed) (the "D&O Tail Policy"). During the term of the D&O Tail Policy, Company shall not, and Qorvo shall procure that the Company does not, take any action following the Completion Date to cause the D&O Tail Policy to be cancelled or any provision therein to be amended or waived; provided, that such policy shall be paid in full by the Company prior to the Completion of the Acquisition and none of the Company (upon or

following Completion of the Acquisition), Qorvo, Parent or any Affiliate of any of the foregoing shall be obligated to pay any premiums or other amounts in respect of such D&O Tail Policy.

- 21.7 The obligations of Company and Qorvo under <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u> shall survive Completion and shall not be terminated or modified in such a manner as to adversely affect any D&O Indemnified Party to whom <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u> apply without the consent of such affected D&O Indemnified Party (it being expressly agreed that the D&O Indemnified Parties to whom <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u> apply shall be third-party beneficiaries of such clauses, each of whom may enforce the provisions of <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u>.
- 21.8 In the event the Company (i) consolidates with any other person and shall not be the continuing or surviving corporation or entity in such consolidation or (ii) transfers all or substantially all of its properties, shares and assets to any person, then, and in either such case, Qorvo shall ensure that proper provision be made so that the successors and assigns of the Company shall assume all of the obligations set forth in <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u>. The agreements and covenants contained herein shall not be deemed to be exclusive of any other rights to which any indemnified party is entitled, whether pursuant to Applicable Law, contract or otherwise. Nothing in this Deed is intended to, shall be construed to or shall release, waive or impair any rights to directors' and officers' insurance claims under any policy that is or has been in existence with respect to the Company or its officers, directors and employees, it being understood and agreed that the indemnification provided for in <u>clause 21.4</u>, <u>clause 21.5</u> and <u>clause 21.6</u> is not prior to, or in substitution for, any such claims under any such policies.
- 21.9 Following the termination of this Deed, Qorvo and each Committed Shareholder agree to take all actions necessary to amend the Articles of Association to delete (a) any rights specifically provided to Qorvo in connection with this Deed, including any exclusive consent rights, and (b) any obligations of the Company and the other Shareholders added to the Articles of Association specifically in connection with this Deed, provided, however, that no changes shall be made with respect to the Qorvo Observer (as defined therein). In addition, by signing this Deed, Qorvo agrees that immediately upon the termination of this Deed any rights specifically provided to Qorvo in the Articles of Association in connection with this Deed, including any exclusive consent rights, shall automatically terminate, and any obligations of the Company and the other Shareholders added to the Articles of Association specifically in connection specifically in connection with this Deed, including any exclusive consent rights, shall automatically terminate, and any obligations of the Company and the other Shareholders added to the Articles of Association specifically in connection with this Deed, shall automatically terminate; provided, however, that all rights and obligations pertaining to the Qorvo Observer shall continue.

22. UNDERTAKINGS BY THE COMPANY

To the extent permitted by Applicable Law, the Company undertakes with Qorvo that it will comply with each of the provisions of this Deed. Each undertaking by the Company in respect of each provision of this Deed shall be construed as a separate undertaking and if any of the undertakings is unlawful or unenforceable the remaining undertakings shall continue to bind the Company.

23. RESTRICTIVE COVENANTS FOR RESTRICTED SHAREHOLDERS

- 23.1 Subject to Completion of the Acquisition, as further consideration for Qorvo undertaking the Acquisition of the Company on the terms set out in this Deed and for good and valuable consideration received by the Restricted Shareholders, and in recognition of the fact that Qorvo will, following Completion of the Acquisition, have an interest as the owner of the Company in securing the protection of the goodwill, trade, Confidential Information and employee base of the Group and as consideration for being given the opportunity to participate in the Offer on the terms and conditions of this Deed and for being given the opportunity to transfer the Shares and Contingent Securities held by them for valuable consideration in accordance with this Deed in an amount determined by reference to the Aggregate Value agreed under this Deed, each Restricted Shareholder gives the undertakings, covenants and obligations set out in this <u>clause 23</u> to Qorvo and to the Company.
- 23.2 Each Restricted Shareholder will not while he holds Shares or Contingent Securities make any investment in any other company save for shares or other capital amounting to less than 5% of the total issued share capital of any company which is listed on a regulated market.
- 23.3 Each Restricted Shareholder undertakes and covenants with Qorvo that he shall not, whether directly or indirectly, on his own behalf or on behalf of or in conjunction with any other person, firm, company or other entity:
 - (a) for the Restricted Period, solicit or entice away or endeavour to solicit or entice away from the Company or any member of the Group any person, firm, company or other entity who is, or was, in the [*****] month period immediately prior to the Completion Date, a client of the Company or any member of the Group with whom the Restricted Shareholder had business dealings during the course of his Employment in that [*****] month period. Nothing in this <u>clause 23.3(a)</u> shall prohibit the seeking or doing of business not in direct or indirect competition with the Business of the Company or any member of the Group;
 - (b) for the Restricted Period, solicit or entice away or endeavour to solicit or entice away from the Company or any member of the Group any person, firm, company or other entity who is, or was, in the [*****] month period immediately prior to the Completion Date, a prospective client of the Company or any member of the Group. For the purposes of this <u>clause 23.3(b)</u> the term "prospective client" shall mean any person, firm, company or other entity which was, in the [*****] month period immediately prior to the Completion Date, being actively solicited or responded positively to canvassing by the Company or any member of the Group and with which solicitation the Restricted Shareholder was personally involved during the course of his Employment in that [*****] month period. Nothing in this <u>clause 23.3(b)</u> shall prohibit the seeking or doing of business not in direct or indirect competition with the Business of the Company or any member of the Group;
 - (c) for the Restricted Period, solicit or entice away or endeavour to solicit or entice away any individual who is employed or engaged by the Company or any member of the Group as a director or in a managerial, executive or technical capacity; and

- (d) for the Restricted Period, carry on, set up, be employed, engaged or interested in a business anywhere in the United Kingdom, the Netherlands, the United States of America or China, or any other country in which the Company or any member of the Group conducts Business at the time of the Completion, which is to be in competition with the Business of the Company or any member of the Group as at the Completion Date with which the Restricted Shareholder was actively involved during the [*****] month period immediately prior to the Completion Date. It is agreed that in the event that any such company ceases to be in competition with the Company and/or any member of the Group this <u>clause 23.3(d)</u> shall, with effect from that date, cease to apply in respect of such company. The provisions of this <u>clause 23.3(d)</u> shall not, at any time following the Completion Date, prevent the Restricted Shareholder from holding shares or other capital amounting to less than 5% of the total issued share capital of any company which is listed on a regulated market and, in addition, shall not prohibit the seeking or doing of business not in direct or indirect competition with the Business of the Company or any member of the Group.
- 23.4 Each Restricted Shareholder undertakes and covenants that at no time after the termination of his Employment shall he directly or indirectly represent himself as being interested in or employed by or in any way connected with any member of the Group, other than as a former director or employee of a member of the Group and (where applicable) as a minority shareholder or option holder or former minority shareholder or option holder of the Company.
- 23.5 By written notice to the Restricted Shareholder, Qorvo may by one or more stages, vary the terms of any part of this clause 23 as it may apply to the Restricted Shareholder by reducing (but not increasing) the period during and/or the activities to which the restrictions are to apply.
- 23.6 Each Restricted Shareholder agrees that, having regard to all the circumstances, the restrictions in this <u>clause 23</u> are no greater than is reasonable and necessary for the protection of Qorvo's interests in the Business and the goodwill of the Company following Completion. The parties agree that:
 - (a) each restriction shall be read and construed independently of the other restrictions so that if one or more are found to be void or unenforceable as an unreasonable restraint of trade or for any other reason the remaining restrictions shall not be affected; and
 - (b) if any restriction is found to be void but would be valid and enforceable if some part of it were deleted, the restriction shall apply with the deletions that are necessary to make it valid and enforceable.

24. DISCLAIMER OF CORPORATE OPPORTUNITY

24.1 To the maximum extent permitted from time to time under the laws of England, the Company renounces any interest or expectancy of the Company in, or in being offered an opportunity to participate in, business opportunities that are from time to time presented to Qorvo or its officers, directors or shareholders (including Qorvo's appointed observer of the Company's Board of Directors). No amendment or repeal of this <u>clause 24.1</u> shall apply to or have any effect on the

liability or alleged liability of any officer, director or shareholder of Qorvo for or with respect to any opportunities of which such officer, director or stockholder becomes aware prior to such amendment or repeal.

- 24.2 Except as Qorvo may otherwise agree in writing after the date hereof:
 - (a) Qorvo will have the right:
 - to directly or indirectly engage in any business (including, without limitation, any business activities or lines of business that are the same as or similar to those pursued by, or competitive with, the Company or any member of the Group) or invest, own or deal in securities of any other person so engaged in any business;
 - (ii) to directly or indirectly do business with any client or customer of the Company or any member of the Group; and
 - (iii) not to present potential transactions, matters or business opportunities to the Company or any member of the Group, and to pursue, directly or indirectly, any such opportunity for itself, and to direct any such opportunity to another person;
 - (b) Qorvo and Qorvo's appointed observer of the Company's Board of Directors will have no duty (contractual or otherwise) to communicate or present any corporate opportunities to the Company or any of its Affiliates or to refrain from any actions specified in <u>clause 24.2(a)(i)</u> of this Deed, and the Company, on its own behalf and on behalf of its Affiliates, each hereby renounces and waives any right to require Qorvo to act in a manner inconsistent with the provisions of this <u>clause 24.2(b)</u>.
 - (c) Qorvo will not be liable to the Company or any of its Affiliates for breach of any duty (contractual or otherwise) by reason of any activities or omissions of the types referred to in this <u>clause 24.2</u> or of any such person's participation therein.

25. CONFLICT WITH ARTICLES; INVESTMENT AGREEMENT

- 25.1 To the extent permitted by law, in the event of any ambiguity or discrepancy between the provisions of this Deed and the Articles of Association, the provisions of this Deed shall prevail as between Qorvo, the Major Investors, the Executive Optionholders and the Committed Shareholders for so long as this Deed remains in force. Each of the Major Investors, the Executive Optionholders and the Committed Shareholders shall exercise all voting and other rights and powers available to it so as to give effect to the provisions of this Deed and, if necessary, to procure (as far as it is able to do so) any required amendment to the Articles of Association.
- 25.2 In the event of any ambiguity or discrepancy between the provisions of this Deed and the provisions of the Amended & Restated Investment Agreement, the provisions of this Deed shall prevail as between the Major Investors, the Committed Shareholders, the Company, the Executive Optionholders and Qorvo for so long as this Deed remains in force. Each of the Major

Investors, the Executive Optionholders and the Committed Shareholders shall agree to waive or vary any rights or obligations under the Amended & Restated Investment Agreement so as to give full effect to the provisions of this Deed.

25.3 On Completion of the Acquisition, each of the Major Investors, the Executive Optionholders and the Committed Shareholders and the Company agrees that the Amended & Restated Investment Agreement shall terminate and have no further force or effect.

26. PRE-EMPTION RIGHTS

Each of the Major Investors, the Executive Optionholders and the Committed Shareholders hereby waives, undertakes to waive and to procure the waiver of, any pre-emption rights if required to give the maximum possible effect to the provisions of this Deed.

27. TERMINATION OF THIS DEED

- 27.1 This Deed shall terminate immediately (except for those provisions expressly stated to continue without limit in time and without prejudice to any rights or liabilities arising under this Deed) if:
 - (a) any right of termination arises for Qorvo under any other provision of this Deed and Qorvo elects to terminate this Deed;
 - (b) Qorvo and the Company mutually agree to terminate this Deed;
 - (c) Qorvo has not delivered an Option Notice prior to the end of the Option Period or the written notice required by clause 4.1(f) by 12:01 a.m. on the date which is [*****] Business Days after the conclusion of the Diligence Period;
 - (d) Following a request by the Company for Additional Funding pursuant to <u>clause 4.2</u>, Qorvo fails to wire the additional funds to the Company prior to [*****] in accordance with the terms of <u>clause 4.2</u>;
 - (e) any court or other Governmental Entity having jurisdiction over a party hereto shall have issued an order, decree or ruling or taken any other action (which order, decree, ruling or other action the parties shall have used their reasonable endeavours to resist, resolve or lift, as applicable, permanently enjoining, permanently restraining or otherwise permanently prohibiting the transactions contemplated by this Deed, and such order, decree, ruling or other action shall have become final and nonappealable; or
 - (f) Qorvo elects by notice in writing to terminate this Deed following a failure of any other party to comply with any of its material covenants or agreements contained in this Deed required to be complied with prior to the date of such termination, which failure to comply has not been cured within [*****] Business Days following receipt by such other party of written notice of such failure to comply.
- 27.2 In addition to any other rights of termination may have under this Deed, Qorvo may terminate this Deed (except for those provisions expressly stated to continue without limit in time and

without prejudice to any rights or liabilities arising under this Deed) if any one or more material inaccuracies or misrepresentations in, or breaches in any material respect of, the warranties made by the Company in this Deed has not been cured within [*****] Business Days following receipt by the Company of written notice from Qorvo of the breach.

27.3 In the event of termination of this Deed, the rights and liabilities accruing to any party prior to termination of this Deed continue and are unaffected, and any provision of this Deed which is expressed to continue following termination shall continue in full force and effect and the parties shall remain bound by such provision.

28. ASSIGNMENT

Other than as expressly provided elsewhere in this Deed, no party may assign, transfer or novate this Deed or any rights under this Deed to any other person without the consent of the other parties. Qorvo shall have the right to assign this Deed to any Affiliate of Qorvo. This Deed will be binding upon, inure to the benefit of, and be enforceable by, the parties and their respective successors and assigns.

29. GENERAL WARRANTIES

- 29.1 Each party to this Deed (other than the Company and the Securityholder Agent) warrants and represents to the other parties that:
 - (a) either:
 - (i) it is a company incorporated or partnership formed and validly existing under the laws of its respective jurisdiction of incorporation or formation;
 - (ii) in respect of each individual, he or she is a natural person who is ordinarily resident in the jurisdiction indicated in <u>Schedule 2</u> (as the case may be);
 - (b) it has the necessary power and authority to enter into and perform this Deed;
 - (c) the execution, delivery and performance by it of this Deed will not result in a material breach of:
 - (i) (if applicable) any provision of the Articles of Association or equivalent constitutional documents of it; or
 - (ii) to its knowledge, any order, judgment or decree of any court or regulatory authority by which it is bound;
 - (d) assuming due execution and delivery by each of the other parties to this Deed, this Deed constitutes, the legal, valid and binding obligations of such party, enforceable against such party in accordance with its terms;

- (e) to its knowledge, it is not, or will not be, required to give any notice to or make any filing with or obtain any permit, consent, waiver or other authorisation from any regulatory authority in connection with the execution, delivery and performance of this Deed;
- (f) there is no claim, action, proceeding, arbitration, investigation, or hearing pending or, so far as it is aware, threatened, by or before any court or regulatory authority or private arbitration tribunal against it which would prevent the entry into and performance of this Deed; and
- (g) no order has been made, petition presented or resolution passed for its winding up, and no administrator or any receiver or manager has been appointed by any person in respect of it or all or any of its assets, and, so far as it is aware, no steps have been taken to initiate any such appointment and no voluntary arrangement has been proposed, and it has not become subject to any analogous proceedings, appointments or arrangements under the laws of any applicable jurisdiction.

30. FURTHER ASSURANCES; SECURITY POWER OF ATTORNEY

- 30.1 Each party shall, from time to time during the term of this Deed, use reasonable endeavours to, and at its own cost do, execute and deliver or procure to be done, executed and delivered all such further acts, documents and things required by, and in a form reasonably satisfactory to, the requesting party to give full effect to this Deed and its rights, powers and remedies under this Deed.
- 30.2 Security power of attorney
 - (a) Pursuant to the Powers of Attorney Act 1971 (UK), each of the Major Investors, Committed Shareholders and Executive Optionholders hereby irrevocably and unconditionally, to the fullest extent permissible under applicable law and by way of security for the performance of its obligations under this Deed, appoints any Qorvo Officer as its attorney to execute and do in its name or otherwise and on its behalf all documents, acts and things which the attorney shall in its absolute discretion consider necessary or desirable in order to implement the obligations of that Major Investor, Committed Shareholder or Executive Optionholder under this Deed with respect to:
 - (i) the execution and delivery of a Form of Acceptance in respect of the Offer immediately upon receipt of the Offer Document and in any case as soon as practicable during the Offer Period;

- (ii) the transfer of all the Shares held by such Major Investor, Committed Shareholder or Executive Optionholder (as the case may be); and
- (iii) the issuance of a Drag-Along Notice by the Major Investors sufficient to enable Qorvo to complete the Drag-Along Acquisition.
- (b) Each Major Investor, Committed Shareholder and Executive Optionholder undertakes to ratify whatever any Qorvo Officer as its attorney shall lawfully do or cause to be done in accordance with its power of attorney and to indemnify and keep such attorney indemnified from all claims, costs, expenses, damages and losses which the attorney may suffer as a result of the lawful exercise by him of the powers conferred on him under this power of attorney.
- (c) This power of attorney shall remain in force in relation to each Major Investor, Committed Shareholder and Executive Optionholder until this Deed is terminated in respect of the rights and obligations of that Major Investor, Committed Shareholder or Executive Optionholder (as the case may be) in accordance with the terms of this Deed. This power of attorney is coupled with an interest.
- 30.3 Pursuant to the security power of attorney given by each Major Investor, Committed Shareholder and Executive Optionholder under this <u>clause 30</u>, if any Major Investor, Committed Shareholder or Executive Optionholder defaults on its obligation to execute and deliver a Form of Acceptance to Qorvo and the Company pursuant to this Deed, the attorney shall execute and deliver on behalf of the Major Investor, Committed Shareholder or Executive Optionholder a Form of Acceptance not less than five (5) days prior to the conclusion of the Offer Period and the attorney shall, on behalf of each Major Investor, Committed Shareholder, do all such acts and/or execute all such documents in a form satisfactory to Qorvo as Qorvo may reasonably require to give effect to the acceptance of the Offer by each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder in accordance with the obligations of each Major Investor, Committed Shareholder and Executive Optionholder under this Deed.
- 30.4 Pursuant to the security power of attorney given by each Major Investor under this <u>clause 30</u>, if any Major Investor defaults on its obligation to issue a Drag-Along Notice sufficient to enable Qorvo to complete the Drag-Along Acquisition, the attorney shall execute and deliver on behalf of such Major Investor such Drag-Along Notice and shall, on behalf of such Major Investor, do all such acts and/or execute all such documents in a form satisfactory to Qorvo as Qorvo may reasonably require to enable Qorvo to complete the Drag-Along Acquisition pursuant to the terms of this Deed.
- 30.5 On completion of any transfer of Shares under the security power of attorney given by each Major Investor, Committed Shareholder and Executive Optionholder under this <u>clause 30</u>:
 - (a) the attorney shall deliver on behalf of the Major Investor, Committed Shareholder or Executive Optionholder to Qorvo a duly executed transfer in favour of Qorvo together with the certificate representing the relevant Shares (to the extent that such certificate is not held by the Company) and a power of attorney in a form and in favour of a person

nominated by Qorvo, so as to enable Qorvo, pending registration, to exercise all rights of ownership in relation to the Shares transferred to it including, without limitation, the voting rights;

- (b) Qorvo shall pay the consideration due in respect of the relevant Shares to the Payments Administrator (in accordance with this <u>clause 30.5</u>) by wire transfer on Completion; and
- (c) the attorney shall, on behalf of each Major Investor, Committed Shareholder or Executive Optionholder, do all such acts and/or execute all such other documents in a form satisfactory to Qorvo as Qorvo may reasonably require to give effect to the transfer of Shares to it.
- 30.6 If a transfer of Shares is executed on behalf of any Major Investor, Committed Shareholder or Executive Optionholder under this <u>clause 30</u>:
 - (a) the Company may receive the purchase money in trust for that Major Investor, Committed Shareholder or Executive Optionholder and the receipt of the Company for the purchase of money shall be a good discharge for Qorvo, who shall not be bound to see to the application of the purchase money;
 - (b) the Company shall, subject to the instrument of transfer being duly stamped, cause Qorvo to be registered as the holder of the relevant Shares; and
 - (c) once registration has taken place in purported exercise of the power contained in this <u>clause 30</u>, the validity of the proceedings shall not be questioned by any person and the Major Investor, Committed Shareholder or Executive Optionholder shall not have any right, and shall renounce any right arising, to claim damages or other compensation or right of action against the Company or any other person in respect of the transfer of the Shares.

31. LIMITATION OF RELATIONSHIP

- 31.1 Nothing in this Deed:
 - (a) is intended to or shall operate to create a partnership of any of the parties or, except as explicitly provided in this Deed, to authorise any party to obligate or bind any other party in any manner whatsoever;
 - (b) will be construed as making one party an agent (except as explicitly provided in this Deed), partner, principal or employer of another party; or
 - (c) is intended to or shall operate to create any term of employment.
- 31.2 Except as explicitly provided in this Deed (and except for the Securityholder Agent's appointment to act on behalf of the Selling Shareholders after Completion), no party will have any authority to bind any other and will not pledge the credit of the other party nor represent itself as being the other party's, partner, employee, agent or Representative and will not hold itself out to any third party as such nor as having any power or authority to incur any obligation of any nature, express or implied, on behalf of the other party.

32. SET-OFF AND WITHHOLDINGS

- 32.1 No party will be entitled to assert any credit, set-off or counterclaim against any other party in order to justify withholding payment of any amount owed in connection with this Deed in whole or in part.
- 32.2 Any amount payable by any party to another party shall be made in full and free from any deduction or withholding whatsoever, except as required by law.

33. NOTICES

- 33.1 Except where expressly stated otherwise, a notice under this Deed shall only be effective if it is in writing. Facsimiles and email are permitted.
- 33.2 Notices under this Deed shall be sent to a party at its address or number and for the attention of the individual set out below:

Party and Title of Individual	Address	Facsimile No.	Email Address
Qorvo	7628 Thorndike Road Greensboro, NC 27409-9421 Attn: Todd Gillenwater	[****]	[****]
	With a copy (which shall not constitute notice) to:		
	Qorvo, Inc. 7628 Thorndike Road Greensboro, NC 27409-9421 Attn: Jeff Howland Fax: [*****] EMail: [*****]		
Company	2960 North 1 st Street, San Jose, California 95134, Attn: CEO		[****]
[****	[****]		[****]
[****]	[****]	[****]	[****]
Shareholder Representative Services LLC (as the Securityholder Agent)	1614 15th Street, Suite 200, Denver, Colorado, 80202, Attention: Managing Director	[****]	[****]
Each Committed Shareholder, Executive Optionholder, Restricted Shareholder and Warrantor	As provided on the signature pages hereto.		

Provided that a party may change its notice details on giving notice to the other parties of the change in accordance with this clause. That notice shall only be effective on the date falling five (5) Business Days after the notification has been received or such later date as may be specified in the notice.

33.3 Any notice given under this Deed shall, in the absence of earlier receipt, be deemed to have been duly given as follows:

(a) if delivered personally, on delivery;

- (b) if sent by first-class post, two (2) Business Days after the date of posting;
- (c) if sent utilising a next-day service by a recognised next-day courier on the (1st) Business Day following the date of posting;
- (d) if sent by airmail, six (6) Business Days after the date of posting;
- (e) if sent by facsimile, at the expiration of forty-eight (48) hours after the time it was sent; and
- (f) if sent by email, at the time at which it was sent.
- 33.4 Any notice given under this Deed outside Working Hours in the place to which it is addressed shall be deemed not to have been given until the start of the next period of Working Hours in such place.
- 33.5 No notice given under this Deed may be withdrawn or revoked except by notice given in accordance with this <u>clause 33</u>.

34. REMEDIES AND WAIVERS

- 34.1 No delay or omission by any party to this Deed in exercising any right, power or remedy provided by law or under this Deed shall:
 - (a) affect that right, power or remedy; or
 - (b) operate as a waiver of it.
- 34.2 The single or partial exercise of any right, power or remedy provided by law or under this Deed shall not, unless otherwise expressly stated, preclude any other or further exercise of it or the exercise of any other right, power or remedy.
- 34.3 The rights, powers and remedies provided in this Deed are cumulative and not exclusive of any rights, powers and remedies provided by law, unless the contrary is expressly stated in this Deed.
- 34.4 The parties agree that irreparable damage would occur in the event that any of the provisions of this Deed were not performed in accordance with their specific terms or were otherwise breached. Accordingly, each of the parties shall be entitled to seek, and the other parties shall not object to such party seeking, specific performance of the terms hereof, including an injunction or injunctions to prevent breaches of this Deed and to enforce specifically the terms and provisions of this Deed, this being in addition to any other remedy to which such party is entitled at law or in equity. Each of the parties hereby further waives (a) any defence in any action for specific performance that a remedy at law would be adequate and (b) any requirement under any law to post security as a prerequisite to obtaining equitable relief.

35. INVALIDITY AND SEVERANCE

- 35.1 If at any time any provision of this Deed is or becomes illegal, invalid or unenforceable in any respect under the law of any jurisdiction, that shall not affect or impair:
 - (a) the legality, validity or enforceability in that jurisdiction of any other provision of this Deed; or
 - (b) the legality, validity or enforceability under the law of any other jurisdiction of that or any other provision of this Deed.
- 35.2 If any invalid, unenforceable or illegal provision would be valid, enforceable and legal if some part of it were deleted, the provision shall apply with whatever deletion is necessary to give effect to the commercial intention of the parties.

36. LANGUAGE

- 36.1 Each notice, demand, request, statement, instrument, certificate or other communication under or in connection with this Deed shall be:
 - (a) in English; or
 - (b) if not in English, accompanied by an English translation made by a translator, and certified by an officer of the party giving the notice to be accurate.
- 36.2 The receiving party shall be entitled to assume the accuracy of and rely upon any English translation of any document provided pursuant to <u>clause 36.1(b)</u>.

37. ENTIRE AGREEMENT

- 37.1 This Deed (and other documents referred to in this Deed) constitutes the whole and only agreement between the parties relating to the subject matter of this Deed (except for the Securityholder Agent's engagement letter among the Securityholder Agent, the Company and the Selling Shareholders).
- 37.2 Each party acknowledges that in entering into this Deed it is not relying on any pre-contractual statement which is not set out in this Deed.
- 37.3 Except in the case of fraud, no party shall have any right of action against any other party to this Deed arising out of or in connection with any pre-contractual statement except to the extent that it is repeated in this Deed.
- 37.4 For the purposes of this <u>clause 37</u>, "**pre-contractual statement**" means any draft, agreement, undertaking, representation, warranty, promise, assurance or arrangement of any nature whatsoever, whether or not in writing, relating to the subject matter of this Deed made or given by any person at any time prior to the date of this Deed.

38. AMENDMENT AND VARIATION

- 38.1 This Deed may be amended or waived in any manner whatsoever by agreement in writing between (x) prior to Completion, the Company and Qorvo and (y) after Completion, the Securityholder Agent and Qorvo, provided that any amendment prior to Completion which may adversely affect:
 - (a) the rights or benefits or obligations of a Major Investor, Committed Shareholder or Executive Optionholder pursuant to this Deed or the Articles of Association; or
 - (b) the enjoyment of any such rights or benefits by a Major Investor, Committed Shareholder or Executive Optionholder,

must be approved by a majority (by number) of the adversely affected Major Investors, Committed Shareholders or Executive Optionholders (as the case may be).

39. CONFIDENTIALITY

- 39.1 Each party shall treat as confidential all information which is disclosed by or obtained through any other party pursuant to, or in connection with, this Deed or the performance of this Deed (whether orally or in writing and whether or not such information is expressly stated to be confidential), or which otherwise comes or has come into the possession of that party, which relates to:
 - (a) the provisions of this Deed;
 - (b) the negotiations relating to this Deed;
 - (c) the subject matter of this Deed;
 - (d) the Offer;
 - (e) the Option;
 - (f) the Offer Document;
 - (g) the Acquisition; or
 - (h) any other party (but only with respect to information about such other party relating to this Deed),

other than information which is in the public domain (otherwise than as a result of any breach by any party of its obligations under this Deed) (such information, the "**Confidential Information**"). Confidential Information shall include any information about the Company or any member of the Group that a party may possess or come into possession of in such party's role as a shareholder, optionholder, director or employee of the Company or any member of the Group. Notwithstanding anything in this Deed (including <u>clause 39.2</u>) to the contrary, following

Completion, the Securityholder Agent shall be permitted to disclose information to the Selling Shareholders who have a need to know such information, provided that such persons either (A) agree to observe the terms of this <u>clause 39</u> or (B) are bound by obligations of confidentiality to the Securityholder Agent of at least as high a standard as those imposed on the Securityholder Agent under this <u>clause 39</u>.

- 39.2 Each party shall:
 - (a) not disclose any such Confidential Information to any person other than any of its directors or employees (or employees of Affiliates who are involved in the monitoring of the investment of such party in the Company) who needs to know such Confidential Information in order to discharge his duties;
 - (b) undertake to the Company and to Qorvo not to use any Confidential Information in any manner in conflict with the Company or in competition with the Business of the Company; and
 - (c) procure that any person to whom any such Confidential Information is disclosed by it complies with the restrictions contained in this <u>clause 39</u> as if such person were a party to this Deed.
- 39.3 Notwithstanding the other provisions of this <u>clause 39</u>, any party may disclose any such Confidential Information:
 - (a) to the extent required by law or for the purpose of any judicial proceedings;
 - (b) to the extent required by existing contractual obligations;
 - (c) to its professional advisers, auditors and bankers provided they have a duty to keep such Confidential Information confidential;
 - (d) to the extent the Confidential Information has come into the public domain through no fault of that party;
 - (e) to the extent the party to which such Confidential Information relates has given prior written consent to the disclosure; and
 - (f) to the extent required for the implementation and enforcement of the rights and obligations of the parties under this Deed and any documents referred to therein;

provided, that if a party is required to disclose Confidential Information by law, rule or regulation, existing contractual obligations or for the purpose of any judicial proceedings, such party will, to the extent permitted by law or the requesting entity, notify the disclosing party as promptly as reasonably practicable so that the disclosing party may seek a protective order or other appropriate remedy and/or waive compliance with this Deed.

- 39.4 Each party shall be entitled to refer to the fact that they have entered into this Deed without being in breach of this clause 39.
- 39.5 The restrictions contained in this <u>clause 39</u> shall continue to apply after the termination of this Deed without limit in time; provided, that, notwithstanding anything to the contrary in this Deed, following Completion, Qorvo shall not be restricted in any way with respect to Confidential Information of the Company.

40. ANNOUNCEMENTS

- 40.1 No public announcement concerning this Deed shall be made by any party without the prior written approval of Qorvo and the Company, such approval not to be unreasonably withheld or delayed.
- 40.2 Notwithstanding <u>clause 40.1</u>, any party may, whenever practicable after consultation with the other parties, make an announcement concerning this Deed if required by:
 - (a) applicable law; or
 - (b) any securities exchange or regulatory or governmental body to which that party is subject, wherever situated, whether or not the requirement has the force of law.

41. COSTS AND EXPENSES

Except as otherwise specified elsewhere in this Deed, each party will bear its own costs associated with the performance of this Deed.

42. COUNTERPARTS

- 42.1 This Deed may be executed in any number of counterparts, and by the parties on separate counterparts, but shall not be effective until Qorvo, the Company and each Major Investor has executed at least one counterpart.
- 42.2 Each counterpart shall constitute an original of this Deed, but all the counterparts shall together constitute but one and the same instrument.

43. CHOICE OF GOVERNING LAW

This Deed is to be governed by and construed in accordance with the laws of England. Any matter, claim or dispute arising out of or in connection with this Deed, whether contractual or non-contractual, is to be governed by and determined in accordance with the laws of England.

44. ARBITRATION

44.1 Any controversy or claim arising out of or relating to this Deed, or the breach thereof, ("**Proceedings**") shall be settled by arbitration administered by the London Court of International Arbitration in accordance with its Arbitration Rules (which shall be deemed to be

incorporated by reference to this clause), and judgment on the award rendered by the arbitrator(s) may be entered in any court having jurisdiction thereof.

- 44.2 Each party shall be entitled to appoint one arbitrator. The arbitrators appointed by the parties must then appoint a third arbitrator, who must serve as the presiding arbitrator.
- 44.3 The location of arbitration must be London, United Kingdom.
- 44.4 The arbitration proceedings must be conducted in English.
- 44.5 The decision of the arbitrators must be in writing and will be final and binding on the parties. A party shall only be permitted to appeal the decision of the arbitrators to the courts of England and Wales in the case of manifest error in the decision.
- 44.6 Each party must bear its own costs and charges and must pay one-half of the costs and expenses of such arbitration, subject always to the final award of the arbitrators as to costs.
- 44.7 Each of the parties hereby acknowledges and agrees that its failure to participate in arbitration proceedings in any respect, or to comply with any request, order or direction of the arbitrators, will not preclude the arbitrators proceeding with such arbitration and/or making a valid final award.
- 44.8 Each party waives (and agrees not to raise) any objection, on the ground of forum non conveniens or on any other ground, to the taking of Proceedings in accordance with this <u>clause 44</u>. Each party also agrees that a judgment against it in Proceedings shall be conclusive and binding upon it and may be enforced in any other jurisdiction.

45. AGENT FOR SERVICE

- 45.1 Each of the Major Investors, the Executive Optionholders and the Committed Shareholders hereby irrevocably appoints Capita Trust Company Limited (Service of Process) 4th Floor, 40 Dukes Place, London EC3A 7NH to be its agent for the receipt of Service Documents ("**Capita**"). It agrees that any claim form, application notice, order, judgment or any other document relating to any Proceedings (a "**Service Document**") may be effectively served on it in connection with Proceedings in England and Wales by service on its agent effected in any manner permitted by the Civil Procedure Rules.
- 45.2 If at any time Capita refuses or is unable to act as an agent under <u>clause 45.1</u>, each affected Major Investor, Executive Optionholder and Shareholder agrees to appoint as its replacement agent a law firm regulated by the Solicitors' Regulation Authority of international standing having an address for service in England and Wales, such law firm to be selected by the Company Directors (in good faith) to act on behalf of each affected Major Investor, Executive Optionholder and Shareholder as an agent for service. The Company shall notify the other parties of the name and address of the replacement agent and shall pay any costs arising on the appointment. The provisions of this <u>clause 45</u> applying to service on an agent apply equally to service on a replacement agent.

- 45.3 A copy of any Service Document served on an agent shall also be sent by post to each Major Investor, Executive Optionholder, Committed Shareholder and the Securityholder Agent. Failure or delay in so doing shall not prejudice the effectiveness of service of the Service Document.
- 45.4 Qorvo hereby irrevocably appoints Ms. Clare Fowley, RF Micro Devices UK, Ltd., 1015 Arlington Business Park Theale, Berkshire, Reading RG7 4SA UK to be its agent for the receipt of Service Documents ("Agent"). It agrees that any Service Document may be effectively served on it in connection with Proceedings in England and Wales by service on its agent effected in any manner permitted by the Civil Procedure Rules.
- 45.5 If at any time Agent refuses or is unable to act as an agent under <u>clause 45.4</u>, Qorvo agrees to appoint as its replacement agent a law firm regulated by the Solicitors' Regulation Authority of international standing having an address for service in England and Wales, such law firm to be selected to act on behalf of Qorvo as an agent for service. Qorvo shall notify the other parties of the name and address of the replacement agent and shall pay any costs arising on the appointment. The provisions of this <u>clause 45</u> applying to service on an agent apply equally to service on a replacement agent.
- 45.6 A copy of any Service Document served on an agent shall also be sent by post to Qorvo. Failure or delay in so doing shall not prejudice the effectiveness of service of the Service Document.

46. SECURITYHOLDER AGENT

- 46.1 Each Selling Shareholder signatory hereto or otherwise bound by the terms hereof shall be deemed to have approved the designation of, and hereby designates, Shareholder Representative Services LLC as the Securityholder Agent under the terms set forth herein. For the avoidance of doubt and notwithstanding anything herein or elsewhere to the contrary, in no event shall the Securityholder Agent be required to have any responsibilities in connection with its engagement hereunder until after the Completion has occurred, other than with respect to its obligations under <u>clause 39</u>.
- 46.2 If the Securityholder Agent shall resign or be removed by the Selling Shareholders, the Selling Shareholders shall (by consent of those Persons entitled to at least a majority of the Aggregate Value to be distributed to the Selling Shareholders), within ten (10) days after such resignation or removal, appoint a successor to the Securityholder Agent, which successor shall be reasonably acceptable to Qorvo and shall not be an employee of Qorvo or any Group Company at such time. Any such successor shall succeed the former Securityholder Agent as the Securityholder Agent hereunder.
- 46.3 The Securityholder Agent will incur no liability of any kind with respect to any action or omission by the Securityholder Agent in connection with its services pursuant to this Deed or the agreements ancillary hereto, except in the event of liability directly resulting from the Securityholder Agent's gross negligence or wilful misconduct. The Selling Shareholders will severally, and not jointly (based upon each Selling Shareholder's fully-diluted ownership interest in the Company relative to the aggregate fully-diluted ownership interests of all Selling

Shareholders party to or bound by the terms of this Deed; provided, that in all cases the Securityholder Agent's indemnity coverage from the Selling Shareholders bound hereby shall sum to 100%), indemnify, defend and hold harmless the Securityholder Agent from and against any and all losses, liabilities, damages, claims, penalties, fines, forfeitures, actions, fees, costs and expenses (including the fees and expenses of counsel and experts and their staffs and all expense of document location, duplication and shipment) (collectively, "Agent Losses") arising out of or in connection with the Securityholder Agent's execution and performance of this Deed and the agreements ancillary hereto, in each case as such Agent Loss is suffered or incurred; provided, that in the event that any such Agent Loss is finally adjudicated to have been directly caused by the gross negligence or wilful misconduct of the Securityholder Agent, the Securityholder Agent will reimburse the Selling Shareholders the amount of such indemnified Agent Loss to the extent attributable to such gross negligence or wilful misconduct. If not paid directly to the Securityholder Agent by the Selling Shareholders, any such Agent Losses may be recovered by the Securityholder Agent from (a) the Expense Fund Amount and (b) the amounts in the Escrow Account at such time as remaining amounts would otherwise be distributable to the Selling Shareholders; provided, that while this section allows the Securityholder Agent to be paid from the Expense Fund Amount and the funds in the Escrow Account, this does not relieve the Selling Shareholders from their obligation to promptly pay such Agent Losses as they are suffered or incurred, nor does it prevent the Securityholder Agent from seeking any remedies available to it at law or otherwise. In no event will the Securityholder Agent be required to advance its own funds on behalf of the Selling Shareholders or otherwise. The Selling Shareholders acknowledge and agree that the foregoing indemnities will survive the resignation or removal of the Securityholder Agent or the termination of this Deed. For the avoidance of doubt, no limitations on liability set forth in any other clause of this Deed are intended to apply to the Securityholder Agent's indemnities hereunder.

- 46.4 The Expense Fund Amount will be used for the purposes of paying directly, or reimbursing the Securityholder Agent for, any third party expenses pursuant to this Deed and the agreements ancillary hereto. The Selling Shareholders will not receive any interest or earnings on the Expense Fund Amount and irrevocably transfer and assign to the Securityholder Agent any ownership right that they may otherwise have had in any such interest or earnings. The Securityholder Agent will not be liable for any loss of principal of the Expense Fund Amount other than as a result of its gross negligence or wilful misconduct. The Securityholder Agent will hold these funds separate from its corporate funds, will not use these funds for its operating expenses or any other corporate purposes and will not voluntarily make these funds available to its creditors in the event of bankruptcy. As soon as practicable following the completion of the Securityholder Agent's responsibilities, the Securityholder Agent shall disburse the balance of the Expense Fund Amount to the Payments Administrator for further distribution to the Selling Shareholders. For tax purposes, the Expense Fund Amount shall be treated as having been received and voluntarily set aside by the Selling Shareholders at the time of Completion. The parties agree that the Securityholder Agent is not acting as a withholding agent or in any similar capacity in connection with the Expense Fund Amount.
- 46.5 Notwithstanding that the Company has been represented by the Sellers' Solicitors in the preparation, negotiation and execution of this Deed, the Company agrees that after Completion

Sellers' Solicitors may represent the Securityholder Agent, the Selling Shareholders and/or their affiliates in matters related hereto, including without limitation in respect of any indemnification claims pursuant hereto. The Company hereby acknowledges, on behalf of itself and its affiliates, that it has had an opportunity to ask for and has obtained information relevant to such representation, including disclosure of the reasonably foreseeable adverse consequences of such representation, and it hereby waives any conflict arising out of such future representation.

47. THIRD PARTY RIGHTS

- 47.1 A person who is not a party to this Deed shall not have any rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of this Deed.
- 47.2 The rights of the parties to terminate, rescind or agree to any variation, waiver or settlement under this Deed is not subject to the consent of any person that is not a party to this Deed.

The Schedules to this Deed now following form part of this Deed.

SCHEDULE 1

DEFINITIONS AND INTERPRETATION

1. Definitions

In this Deed and the Schedules:

"Accounting Period"	means the period between 1 April in each year and 31 March in the next year;	
"Accounting Policies"	means the accounting policies set out in Schedule 14 (Accounting Policies);	
"Accounts"	means the audited consolidated financial statements of the Company and its Subsidiaries, prepared in accordance with Applicable Law, comprising, in each case, the balance sheet, the profit and loss statement, the notes to the accounts, the directors' and auditors' reports and cashflow statement for each of the three (3) consecutive accounting reference periods the last of which ended on the Accounts Date, copies of which are attached to the Disclosure Letter;	
"Accounts Date"	means the date of the most recent Accounts (for purposes of the execution of this Deed, 31 March 2014);	
"Acquisition"	has the meaning given in <u>clause 4.1</u> and means the acquisition by Qorvo of all of the Shares and the Contingent Securities pursuant to this Deed;	
"Action"	means any claim, action, suit, inquiry, proceeding, audit or investigation by or before any Governmental Entity, or any other arbitration, mediation or similar proceeding;	
"Actual Net Debt Amount"	means the amount calculated in accordance with $\underline{Part (B)(1)}$ of <u>Schedule 14</u> (Accounting Policies) and in an amount as finally agreed to on the Actual Net Debt Statement;	
"Actual Net Debt Statement"	means a statement setting out the Actual Net Debt Amount as at the Effective Time, as agreed or determined in accordance with <u>Part (B)(2)</u> of <u>Schedule 14</u> (Accounting Policies);	
"Actual Working Capital Amount"	means the amount calculated in accordance with <u>Part (C)</u> of <u>Schedule 14</u> (Accounting Policies) and in an amount as finally agreed to on the Actual Working Capital Statement;	
"Actual Working Capital Statement"	means a statement setting out the Actual Working Capital Amount as at the Effective Time, as agreed or determined in accordance with <u>Schedule 11</u> (Effective Time Statements);	
"Additional Funding"	has the meaning given in <u>clause 4.2;</u>	
"Additional Funding Notice"	has the meaning given in <u>clause 4.2;</u>	

"Affiliate"	means, in relation to any person, any entity over which that person has Control, any Controller of that person or any other entity which is controlled by the same person as Controls the first entity;
"Affiliated Person"	has the meaning given in paragraph 20.1 of Schedule 8;
"Aggregate Consideration"	has the meaning given in <u>clause 7.7;</u>
"Aggregate Exercise Price"	means (a) the aggregate of: (i) the exercise price for each vested Contingent Security which is purchased by Qorvo pursuant to <u>clause 8.2(a)</u> or <u>8.5(a)</u> , multiplied by (ii) the number of Shares issuable upon exercise of such Contingent Security, <u>plus</u> (b) the lower of (i) the aggregate of (A) the exercise price for each unvested Contingent Security which is purchased by Qorvo pursuant to <u>clause 8.2(a)</u> or <u>clause 8.5(a)</u> , multiplied by (B) the number of Shares issuable upon exercise of such Contingent Security and (ii) [*****].
"Aggregate Value"	has the meaning given in <u>clause 7.2;</u>
"Amended & Restated Investment Agreement"	means the amended and restated investment agreement entered into on the date of this Deed as a condition precedent to the Subscription Agreement, between, inter alia, the Major Investors, Qorvo and the Company;
"Annualised Legacy Product Revenue"	has the meaning given in <u>clause 7.2(e);</u>
"Applicable Laws"	means any of the following, to the extent that it applies to this Deed, the Offer or the Acquisition in any relevant jurisdiction in which any Selling Shareholder resides or within which any Group Company is incorporated or operates: (a) any statute, regulation, by-law, legislation, ordinance or subordinate legislation in force from time to time; (b) the common law and the law of equity; (c) any applicable civil or criminal code; (d) any binding court order, judgment or decree; (e) any applicable industry code, policy or standard enforceable by law; and (f) any applicable direction, policy, requirement, rule or order that is given by a regulatory authority which is enforceable by law;
"Articles of Association"	means the articles of association of the Company in the form set out in <u>Schedule 12</u> or, if the articles of association of the Company are amended or replaced, the articles of association so amended or replaced;
"Associate"	means in relation to each party who is an individual: (a) the spouse, civil partner, widow or child or grandchild or remoter issue (including any step or adopted child) of the party; or (b) the trustees of any trust (whether arising under settlement, declaration, trust will or on intestacy) established by or on behalf of the party under which the party and/or any person identified in the foregoing <u>clause (a)</u> are the only persons capable of being a beneficiary or beneficiaries thereof;

"Balancing Amount"	means an amount equal to the aggregate of:
	(a) the Working Capital Shortfall (expressed as a negative number) or the Working Capital Excess; and
	(b) the Net Debt Shortfall or the Net Debt Excess (expressed as a negative number);
"Board of Directors"	means the board of directors of the Company consisting of all the Directors so appointed from time to time or any duly appointed committee thereof;
"Books and Records"	means all books of account, financial and accounting records, correspondence, documents, files, memoranda and other papers (in electronic, paper or other form) relating to the Company and the Subsidiaries and all computer disks or tapes or other machine legible programs relating to the Company and the Subsidiaries;
"Business"	means the research, development, design, manufacture, marketing and sale of [*****];
"Business Day"	means a day (other than a Saturday or a Sunday) on which banks are open for business in both London and California;
"California Subsidiary"	means Cavendish Kinetics Inc., a California corporation incorporated as a wholly-owned subsidiary of the Company;
"Cash"	means:
"Cash"	means: (a) all cash or cash equivalents held by or for any Group Company; and
"Cash"	
"Cash"	(a) all cash or cash equivalents held by or for any Group Company; and
"Cash" "Change of Control", "Control", "Controller" and "Controlled"	(a) all cash or cash equivalents held by or for any Group Company; and(b) all balances of any Group Company with any bank or other financial institution,
"Change of Control", "Control", "Controller" and	(a) all cash or cash equivalents held by or for any Group Company; and(b) all balances of any Group Company with any bank or other financial institution, but excluding any Trapped Cash;
"Change of Control", "Control", "Controller" and "Controlled"	 (a) all cash or cash equivalents held by or for any Group Company; and (b) all balances of any Group Company with any bank or other financial institution, but excluding any Trapped Cash; each has the meaning given in paragraph 2 of this Schedule 1;
"Change of Control", "Control", "Controller" and "Controlled" "Claim Amount"	 (a) all cash or cash equivalents held by or for any Group Company; and (b) all balances of any Group Company with any bank or other financial institution, but excluding any Trapped Cash; each has the meaning given in paragraph 2 of this Schedule 1; has the meaning given in clause 13.7(a);
"Change of Control", "Control", "Controller" and "Controlled" "Claim Amount" "Claim Notice"	 (a) all cash or cash equivalents held by or for any Group Company; and (b) all balances of any Group Company with any bank or other financial institution, but excluding any Trapped Cash; each has the meaning given in paragraph 2 of this Schedule 1; has the meaning given in clause 13.7(a); has the meaning given in clause 13.7(a);

"[****] Warrants"	means the warrants constituted by the [*****] Warrant Instruments;
"Committed Shareholder"	means each of the persons listed in <u>Part A</u> of <u>Schedule 2</u> (Schedule of Shareholders), and (a) any Other Shareholder or (b) any Contingent Shareholder who exercises any Contingent Security following the date of this Deed and who receives Ordinary Shares or Preferred Shares on exercise of such Contingent Security, in each case in (a) and (b) above where such person executes a Deed of Adherence to accede to this Deed as a Committed Shareholder, shall also be considered to be a Committed Shareholder for all purposes under this Deed on and from the date of such Deed of Adherence, and <u>Part A</u> of <u>Schedule 2</u> (Schedule of Shareholders) shall be updated as necessary to add any such additional Committed Shareholders;
"Companies Act"	means the Companies Act 2006 (UK) and any regulations or subordinate legislation enacted thereunder;
"Completion"	means the completion of the sale and purchase of all the Shares and Contingent Securities pursuant to this Deed;
"Completion Date"	has the meaning given in <u>clause 9.5;</u>
"Condition"	has the meaning given in <u>clause 2.1;</u>
"Confidential Information"	has the meaning given in <u>clause 39.1;</u>
"Contingent Securities"	means any and all outstanding warrants, options or other Convertible Securities or other instruments convertible into Shares which may be issued by the Company from time to time, and includes all such Contingent Securities held at any time by the Contingent Shareholders as identified in <u>Schedule 3</u> (including the Dutch Options, the Ltd Options and the [*****] Warrants);
"Contingent Shareholder"	means each of the persons listed in <u>Schedule 3</u> and any other person who holds Contingent Securities from time to time;
"Convertible Securities"	means any bonds, debentures, notes or other evidences of indebtedness, and any warrants, shares or any other securities (other than Preferred Shares) convertible into, exercisable for, or exchangeable for Ordinary Shares;
"Corporate Activity"	has the meaning given in <u>clause 15.1;</u>
"[****]"	means [*****], a corporation incorporated in [*****] with its principal place of business at [*****];
"[*****] Confidentiality Agreement"	means that certain confidentiality and non-disclosure agreement anticipated to be entered into by and among Qorvo, the Company and [*****], in a form mutually agreed by such parties;
"D&O Indemnified Party"	has the meaning given in <u>clause 21.4;</u>

"D&O Tail Policy"	has the meaning given in <u>clause 21.6;</u>
"Data Room"	means the data room created and maintained by the Company for the purposes of facilitating the due diligence of the Company in connection with the Series F Financing and this Deed, as such data room may be updated in accordance with <u>clause 6.2</u> ;
"Deed of Adherence"	means a deed in substantially the form set out in Schedule 10;
"Default Consideration"	has the meaning given in <u>clause 7.9(c);</u>
"Defaulting Drag-Along Shareholder"	has the meaning given in <u>clause 11.4;</u>
"Defaulting Shareholders"	has the meaning given in <u>clause 7.9(c)</u> ;
"Designated Individuals"	has the meaning given in <u>clause 21.3;</u>
"Diligence Period"	has the meaning given in <u>clause 4.1(e);</u>
"Directors"	means the directors of the Company holding office and duly appointed in accordance with the Articles of Association from time to time;
"Disclosure Letter"	means the disclosure letter dated on the date of this Deed together with all attachments thereto addressed by the Warrantors to Qorvo which discloses exceptions to the Warranties;
"Disposal"	in relation to a Share or Contingent Security, includes, without limitation:
	(a) sale, assignment or transfer;
	(b) creating or permitting to subsist any Lien;
	(c) creating any trust or conferring any interest;
	(d) any agreement, arrangement or understanding in respect of votes or the right to receive dividends;
	(e) the renunciation or assignment of any right to receive dividends; and
	(f) any agreement to do any of the above, except an agreement to transfer Shares or Contingent Security which is conditional on compliance with the terms of this Deed;
"Drag-Along Acquisition"	has the meaning given in <u>clause 11.6;</u>
"Drag-Along Notice"	has the meaning given in the Articles of Association;

"Dutch Options"	means any options or other rights convertible into Ordinary Shares which are issued pursuant to either the Cavendish Kinetics BV 2003 Share Option Plan or the Cavendish Kinetics BV 2007 Share Option Plan;
"Dutch Subsidiary"	means Cavendish Kinetics BV, a wholly-owned subsidiary of the Company, incorporated as a private limited liability company in the Netherlands and having its registered office at Hugo de Grootlaan 35, 5223LB's-Hertogenbosh, The Netherlands;
"Effective Time"	means 12:01 a.m. on the Completion Date;
"Effective Time Statements"	means the Actual Net Debt Statement and the Actual Working Capital Statement;
"Employee"	means an employee of any Group Company;
"Employee Share Scheme"	means the Cavendish Kinetics BV 2003 Share Option Plan, the Cavendish Kinetics BV 2007 Share Option Plan, the Cavendish Kinetics Limited 2007 Equity Incentive Plan and any other share ownership scheme or plan established by the Company or any Subsidiary with the consent of Qorvo for the benefit of Employees and/or former Employees;
"Employment"	means the employment and, if applicable, director relationship and consulting relationship of the relevant Restricted Shareholder with the Company or any member of the Group;
"Equity Consideration"	has the meaning given in <u>clause 7.2(a);</u>
"ERISA"	means the Employee Retirement Income Security Act of 1974 of the United States, as amended;
"Escrow Account"	has the meaning given in <u>clause 7.5;</u>
"Escrow Agent"	means MUFG Union Bank, N.A., a national banking association;
"Escrow Agreement"	means that certain Escrow Agreement to be entered into at or prior to Completion by and among Qorvo, the Company, the Securityholder Agent and the Escrow Agent in substantially the form attached hereto as Schedule 19;
"Estimated Aggregate Dutiable Consideration"	means Qorvo's good faith estimate of the amount of the Aggregate Consideration which shall be paid in accordance with the terms of this Deed for the transfer to Qorvo of all the Shares held by the Selling Shareholders, which estimate shall be consistent with the principles of allocation of the Aggregate Value as set forth in this Deed (including <u>clause 7.6</u>); provided, for the sake of clarity, no amount of Aggregate Consideration attributable to Contingent Securities shall be included in Estimated Aggregate Dutiable Consideration;
"Estimated Consideration"	means the Equity Consideration less the Expense Fund Amount less the Estimated Net Debt Amount, less the Share of Stamp Duty;

"Estimated Net Debt Amount"	means an amount notified in writing by Qorvo to the Company on or after the conclusion of the Diligence Period, being a reasonable and good faith estimate of:
	(a) the amount of External Debt as at the Effective Time; less
	(b) the amount of Cash as at the Effective Time,
	as set out in the Estimated Net Debt Statement. In determining the Estimated Net Debt Amount, Qorvo shall consult with the Company to determine a reasonable and good faith estimate;
"Estimated Net Debt Statement"	means a statement setting out the constituent elements of the Estimated Net Debt Amount (including the constituent elements of the estimated External Debt) in the agreed form signed on behalf of the Company and Qorvo on the Completion Date;
"Exclusivity Period"	has the meaning given in <u>clause 15.3;</u>
"Executive Optionholders"	has the meaning given in the list of parties to this Deed;
"Expense Fund Amount"	means an amount of \$[*****].
"Expert"	has the meaning given in paragraph 9 of Part A of Schedule 11;
"Extended Option Period"	has the meaning given in <u>clause 4.2;</u>

"External Debt"

means:

(a) all borrowings and other indebtedness of any Group Company by way of any overdraft, acceptance, credit or similar facility including under the Finance Documents;

(b) all break fees, prepayment fees and other costs, expenses or penalties related to or arising as a result of the termination or prepayment of any facilities referred to in <u>paragraph (a)</u> above;

(c) the total redemption amount of all shareholder loans (including all break fees, prepayment fees and other costs, expenses or penalties related to or arising as a result of the termination or prepayment);

(d) all other arrangements the purpose of which is to borrow money;

(e) all performance bonds, guarantees, letters of credit, sureties, financial indemnities or other assurances against the financial loss of any person other than a Group Company (at the face value of such bond or other item);

(f) all recourse factoring or recourse discounting of receivables;

(g) all interest accrued on any or all of the borrowings detailed above;

(h) the mark to market effect (positive or negative) of all interest rate, foreign exchange and other derivative instruments, and any amounts payable on the termination of such arrangements if they are to be terminated on Completion;

(i) all finance lease obligations;

(j) all outstanding fees payable to the Sellers' Solicitors, the Sellers' Accountants and any other advisers engaged on behalf of the Selling Shareholders in connection with the transaction contemplated by this Deed;

(k) all bonuses payable by any Group Company to the extent they arise as a result of the sale of the shares in any Group Company (including any related National Insurance Contributions and any other social security contributions or Tax payable by any Group Company);

(l) the amount of the pre Effective Time Tax liabilities of all Group Companies to the extent not paid prior to Completion (computed as at the Effective Time as if the Effective Time was the end of an accounting period for the Group but disregarding (i) any provision or reserve for deferred Tax and (ii) the availability of any relief which is dependent on expenditure by the Group after the Effective Time);

(m) all National Insurance Contributions and any other social security contributions or Tax payable by a Group Company in respect of (i) the issue of shares or securities, (ii) the issue or exercise of share options or securities options or (iii) the redemption, purchase or repayment of any securities of any Group Company; and

(n) an amount equal to any Leakage that occurs or has occurred between the Effective Time and Completion;

"Family Trust"	has the meaning given in the Articles of Association;
"Final Disclosure Letter"	has the meaning given in <u>clause 13.4;</u>
"Finance Documents"	means that certain Loan and Security Agreement between the California Subsidiary and [*****] (as lender), dated [*****];
"Form of Acceptance"	means the document to be delivered by all Selling Shareholders accepting the Offer, in substantially the form set out in <u>Schedule 5</u> ;
"Form 980 Statutory Declaration"	means the statutory declaration required to be sent under Section 980 of the Companies Act in connection with the Squeeze-Out Acquisition;
"Form 980(1) Statutory Notice"	means the statutory notice required to be sent under Section 980(1) of the Companies Act in connection with the Squeeze-Out Acquisition;
"Fundamental Warranties"	has the meaning given in <u>clause 13.6;</u>
"Governmental Entity"	means federal, national, supranational, state, provincial, local or similar government, governmental, regulatory or administrative authority, branch, agency or commission or any court, tribunal or judicial body (including any grand jury);
"Group"	means the Company and each Subsidiary at the relevant time;
"Group Company"	means the Company or any Subsidiary and any other Subsidiary Undertaking of the Company from time to time;
"HSR Act"	means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended;
"Indemnifiable Portion"	means that portion of a Loss, as finally determined in accordance with the terms of this Deed, that is subject to indemnification by the Warrantors, and which is calculated by multiplying the final Loss amount (as determined pursuant to the provisions of this Deed) by the sum of the Warrantors' Pro Rata Shares. For the sake of clarity, if the final Loss amount as determined in accordance with the terms of this Deed is \$[*****], and the sum of the Warrantors' Pro Rata Shares is [*****]%, then the Indemnifiable Portion would be \$[*****], and Qorvo and its affiliates would be entitled to recover \$[*****] from the Warrantors in accordance with their pro rata share, as amongst each other;
"Indemnification Claim"	has the meaning given in <u>clause 13.7(a);</u>
"Indemnitees"	has the meaning given in <u>clause 13.1;</u>
"Indirect Tax"	means value added, turnover, sales, use, distribution or corresponding Tax;
"Initial Option Period"	has the meaning given in <u>clause 4.1(a);</u>

"Interim Accounts"	means the most recent consolidated financial statements of the Company and its Subsidiaries, prepared in accordance with Applicable Law, comprising, in each case, the balance sheet, the profit and loss statement, and cashflow statement for (i) the fiscal year ended March 31, 2015 and (ii) for the two-month period ended May 31, 2015, copies of which are attached to the Disclosure Letter;
"IRS"	means the Internal Revenue Service of the United States;
"Key Employee"	means each employee of the Group who is determined by Qorvo to be a Key Employee, as may be notified by Qorvo to the Company from and after the date on which Qorvo delivers the Option Notice;
"Leakage"	means:
	(i) any dividend or distribution paid or made by any Group Company to the Selling Shareholder;
	 (ii) any payments (including management fees) made to, assets transferred to or liabilities assumed, indemnified or incurred for the benefit of any Selling Shareholder or any of their related persons or Affiliates (excluding the Group Companies) by any Group Company, other than: (i) pursuant to commercial agreements in the ordinary course of or in connection with that Group Company's business; or (ii) in respect of their employment (if applicable);
	(c) any payment made by any Group Company to any Selling Shareholder in respect of any share capital or other securities of any Group Company being redeemed, purchased or repaid, or any other return of capital; or
	(d) the waiver by any Group Company of any amount owed to that Group Company by any Selling Shareholder or any of their related persons or Affiliates.
"Legacy Product Revenue"	has the meaning given in <u>Schedule 15;</u>
"Liens"	means mortgages, liens, encumbrances, charges, claims, pledges, security interests or impositions;
"Liquidation Event"	has the meaning given in the Articles of Association;
"Liquidation Preference"	has the meaning given in the Articles of Association;
"Losses"	means any and all losses, damages, liabilities, deficiencies, claims, diminution of value, interest, awards, judgments, penalties, costs and expenses (including actual attorneys' fees, costs and other actual out-of-pocket expenses incurred in investigating, preparing, actioning or defending the foregoing); provided, however, that other than in connection with Third Party Claims, Losses shall not include indirect or consequential loss or damage, or loss of profit or special loss (save where such loss of profits or special loss is a direct loss);

"Ltd Options"	means any options or rights, other than Dutch Options, convertible into Ordinary Shares issued in accordance with the Cavendish Kinetics Limited 2007 Equity Incentive Plan;
"Major Investors"	has the meaning given in the list of parties to this Deed;
"Material Adverse Effect"	means any change or effect that is or would reasonably be expected (as far as can be foreseen at the time) to be materially adverse to the business, operations, properties, assets, liabilities, earnings or results of operations, or the condition (financial or otherwise) of the Company and its Subsidiaries, taken as a whole, provided, however, that Material Adverse Effect shall not be deemed to include any of the foregoing arising out of or by virtue of:
	(a) any change or effect in the Company's industry or in the economy or financial markets generally to the extent such changes do not have a disproportionate effect on the Company and its Subsidiaries relative to other participants in the Company's industry;
	(b) natural disaster, commencement or material escalation of war, armed hostilities or other international or national crisis or security event directly or indirectly involving the United States after the date of this Deed, to the extent that such events do not have a disproportionate effect on the Company and its Subsidiaries relative to other participants in the Company's industry;
	(c) any change in Applicable Law or UK GAAP, to the extent that such events do not have a disproportionate effect on the Company and its Subsidiaries relative to other participants in the Company's industry;
	(d) any failure to meet internal forecasts or financial projections, in and of itself, though the facts and circumstances giving rise to or contributing to such failure may be taken into account in determining whether there has been a Material Adverse Effect; and
	(e) any change or effect as a direct result of the requirements of this Deed or the announcement or pendency of the transactions contemplated by this Deed;
"Material Agreement"	means any agreement between any person and any member of the Group with a contracted amount in excess of USD \$[*****] (or its equivalent) paid or received annually;
"Material Group Asset"	means any asset(s) held by any member of the Group with a market value in excess of USD \$[*****] (or its equivalent);
"National Insurance Contributions"	means any amounts of national insurance or social security contributions which any person is or may be or become bound to make to any person and which is or purports to be in the nature of taxation;
"Net Debt Excess"	means the amount (if any) by which the Actual Net Debt Amount exceeds the Estimated Net Debt Amount;
"Net Debt Shortfall"	means the amount (if any) by which the Actual Net Debt Amount is less than the Estimated Net Debt Amount;

"Offer"	has the meaning given in <u>clause 4.1</u> and means the offer to be made by Qorvo to acquire the Shares and the Contingent Securities from the Selling Shareholder pursuant to this Deed;
	Shares and the Contingent Securities from the Sening Shareholder pursuant to this Deed,
"Offer Confirmation Notice"	has the meaning given in clause $4.1(f)$;
"Offer Document"	means the document to be delivered to all Selling Shareholders in connection with the Offer, in substantially the form set out in <u>Schedule 4</u> , with such revisions and additions as may be deemed appropriate or required under any applicable securities laws and regulations or any other Applicable Laws;
"Offer Period"	has the meaning given in <u>clause 4.1(h);</u>
"Option"	has the meaning given in <u>clause 4.1;</u>
"Option Notice"	has the meaning given in <u>clause 4.1(c)</u> ;
"Option Period"	has the meaning given in <u>clause 4.1(a)</u> and includes the Initial Option Period and any Extended Option Period;
"Ordinary Shares"	has the meaning given in the Articles of Association;
"Other Shareholder(s)"	means those persons listed in <u>Part B</u> of <u>Schedule 2</u> and any other person who holds Shares in the Company from time to time, excluding the Major Investors, the Committed Shareholders and the Executive Optionholders; and <u>Part B</u> of <u>Schedule 2</u> (Schedule of Shareholders) shall be updated as necessary to add any such additional Other Shareholders;
"Overpayment Amount"	has the meaning given in <u>clause 7.3(c);</u>
"Parent"	means Qorvo, Inc., a Delaware corporation;
"Parent Stock Completion Value"	means the volume weighted average of the daily closing prices per share of the common stock of Parent on NASDAQ (or other primary market or exchange on which such stock is then trading) for the five (5) consecutive trading days ending on and including the trading day immediately preceding the date of Completion, as reported (by Bloomberg Financial Markets) in composite transactions for NASDAQ (or such other market or exchange);
"Payments Administrator"	means Acquiom Clearinghouse LLC, a Delaware limited liability company, which will be engaged to process certain payments hereunder pursuant to the Payments Agreement in connection with the Completion;
"Payments Agreement"	means that certain Acquiom Payments Administration Agreement to be entered into at or prior to Completion by and among the Payments Administrator, Qorvo and the Securityholder Agent in substantially the form attached hereto as Schedule 20;

"Payment Statement"	means a statement setting out the total consideration payable under <u>clause 7</u> and the Balancing Amount payable by Qorvo or the Selling Shareholders under <u>clause 7.3</u> , as agreed and determined in accordance with <u>Schedule 11</u> (Effective Time Statements);
"Permitted Transferee"	has the meaning given in <u>clause 19</u> as the context may require;
"Preferred Shares"	has the meaning given in the Articles of Association;

"**Preproduction Release of the** has the meaning given in <u>Schedule 16;</u> **Specified Product**"

"Pro Rata Share"	means, with respect to any Warrantor, such Warrantor's ownership interest in the Company as of immediately prior to the Completion, determined by dividing (a) the number of Shares owned by such Warrantor as of immediately prior to the Completion, plus the number of Shares issuable upon exercise of all Contingent Securities held by such Warrantor as of immediately prior to the Completion by (b) the sum of (i) the aggregate number of Shares issued immediately prior to the Completion on an as-converted to Ordinary Shares basis and (ii) the aggregate number of Shares issuable upon the exercise in full of all Contingent Securities issued immediately prior to the Completion;
"Proceedings"	has the meaning given in <u>clause 44.1;</u>
"Products"	means any products sold or distributed by the Company or any of its Subsidiaries;
"Qualifying Offer"	has the meaning given in the Articles of Association;
"Qorvo Competitor"	means, at any given time, any company or other person that is listed or described as a competitor of Qorvo or Parent in any annual report or quarterly report filed by Parent with the United States Securities and Exchange Commission within the 24 months immediately preceding such time, and any successor or acquiror of any such company or person; provided, however, that any company or other person that holds Shares or Contingent Securities as of the date of this Deed, shall not be considered a "Qorvo Competitor" for purposes of this Deed with respect only to sales or issuances of Shares or Contingent Securities based on preemptive or pro rata participation rights existing between such Person and the Company as of the date hereof and disclosed to Qorvo in the Disclosure Letter;
"Qorvo Officer"	means any person who holds office from time to time as an executive officer or corporate vice president director of Qorvo or Parent;
"Qorvo's Accountants"	means KPMG LLP;
"Recipient"	has the meaning given in <u>clause 13.11;</u>
"Representatives"	means, with respect to any person, the officers, directors, principals, employees, agents, auditors, advisers, bankers and other representatives of such person;

"Required Consents"	has the meaning given in <u>clause 5.5;</u>
"Required Notices"	has the meaning given in <u>clause 5.5;</u>
"Restricted Period"	means a period of [*****] months following the Completion Date;
"Restricted Shareholders"	means each of the Executive Optionholders who is a party to this Deed as of the date of this Deed other than [*****]; provided, however, that if an Executive Optionholder (other than [*****]) is not an Employee, director or manager of or consultant to the Company or any of its Subsidiaries as of the date of delivery of the Option Notice, then such individual shall no longer be a Restricted Shareholder; and, provided, further, that with respect to [*****], the only provisions of clause 23 that shall apply to him are clause 23.1, clause 23.3(c) and clause 23.6.
"Revenue Payment"	has the meaning given in <u>clause 7.2(e);</u>
"Sellers' Accountants"	means PricewaterhouseCoopers LLP;
"Sellers' Disagreement Notice"	has the meaning given in <u>paragraph 4(b)</u> of <u>Part A</u> of <u>Schedule 11</u> ;
"Sellers' Solicitors"	means Pillsbury Winthrop Shaw Pittman LLP;
"Selling Shareholders"	means each of the Major Investors, the Committed Shareholders, the Other Shareholders and the Contingent Shareholders (including the Executive Optionholders);

"Series A Liquidation have the meanings given in the Articles of Association; Preference," "Series A Preferred Shares," **"Series B Liquidation** Preference.' "Series B Preferred Shares," "Series B-1 Liquidation Preference," "Series B-1 Preferred Shares," "Series C Liquidation Preference," "Series C Preferred Shares," **"Series D Liquidation** Preference.³ "Series D Preferred Shares," "Series E Liquidation Preference,' "Series E Preferred Shares," "Series F Liquidation Preference" and "Series F Preferred Shares" "Series F Financing" means the financing related to the issuance and sale of Series F Preferred Shares pursuant to the Subscription Agreement and the transactions contemplated thereby. means the service and employment agreements entered into between each of the Restricted "Service Agreement" Shareholders and the Company or another member of the Group; "Service Document" has the meaning given in clause 45; "Shareholder(s)" means each of Qorvo, the Major Investors, the Committed Shareholders and the Other Shareholders and, upon the issuance to any Contingent Shareholder of any Shares one exercise of any rights attaching to Contingent Securities, such Contingent Shareholder; means Shareholder Representative Services LLC, a Colorado limited liability company, acting "Securityholder Agent" solely in its capacity as the representative and agent of the Selling Shareholders; "Share of Stamp Duty" means the Stamp Duty Percentage of the Estimated Aggregate Dutiable Consideration; "Shares" means any shares issued or to be issued in the share capital of the Company, and includes the Ordinary Shares and the Preferred Shares;

(([+++++]))	has the meaning since in [****].
"[*****]"	has the meaning given in [*****];
"Specified Product"	means [*****];
"Squeeze-Out Acquisition"	has the meaning given in <u>clause 10.1;</u>
"Stamp Duty Percentage"	means an amount equal to one-half of the percentage rate of stamp duty payable on instruments effecting the transfer of shares pursuant to the Finance Act 2003, or any successor legislation thereof, as in effect on the Completion Date.
"Subscription Agreement"	has the meaning given in the Recitals;
"Subsidiary"	means the Dutch Subsidiary, the California Subsidiary and each other subsidiary or Subsidiary Undertaking of the Company from time to time;
"Subsidiary Undertaking"	shall be construed in accordance with sections 1161 and 1162 of the Act;
"Tax" or "Taxation"	means, whether or not directly or primarily chargeable against or attributable to the Company and regardless of whether the Company or any Subsidiary has, or may have, any right of reimbursement against any other person:
	(a) any form of tax, levy, impost, duty, contribution, customs and other import duties, liability and charge in the nature of taxation and all related withholdings or deductions of any kind (including, for the avoidance of doubt, any National Insurance Contributions and social security contribution liabilities and similar or corresponding obligations) wherever and whenever payable and shall further include any amount payable as a consequence of any claim, direction order or determination of any Taxation Authority; and
	(b) all fines, penalties, charges, costs and interest included in or relating to any of the above or to any obligation in respect of any of the above;
"Tax Warranty"	means a warranty in paragraph 9 of Schedule 8 (Warranties);
"Taxation Authority"	means any government, state or municipality or any local, state, federal or other fiscal, revenue, customs or excise authority, body or official competent to impose, administer, levy, assess or collect Tax;
"Taxation Statute"	means any directive, statute, enactment, law or regulation wherever enacted or issued, coming into force or entered into providing for or imposing any Tax, or providing for the reporting, collection, assessment or administration of any Tax liability, and shall include orders, regulations, instruments, by-laws or other subordinate legislation made under the relevant statute or statutory provision and any directive, statute, enactment, law, order, regulation or provision that amends, extends, consolidates or replaces the same or that has been amended, extended, consolidated or replaced by the same;
"Third Party"	has the meaning given in <u>clause 15.1;</u>

"Third Party Claim"	means any claim, suit or cause of action made or brought by any person who is not a party to this Deed or an Affiliate of a party to this Agreement or a Representative of the foregoing
"Third Party Negotiations"	has the meaning given in <u>clause 15.1;</u>
"Transaction Documents"	means this Deed, the Subscription Agreement, and the Amended & Restated Investment Agreement;
"Trapped Cash"	means cash or cash equivalents held by any Group Company which may not be lawfully used by that Group Company to service indebtedness (if any) incurred at Completion by that Group Company and may not lawfully be paid by way of dividend, distributed or lent by that Group Company to Qorvo;
"UK GAAP"	means all Statements of Standard Accounting Practice, Financial Reporting Standards and Urgent Issues Task Force Abstracts issued by the Accounting Standards Board, and extant/mandatory for adoption as at Completion of the Acquisition;
"Unvested Ltd Options"	has the meaning given in <u>clause 8.2(b);</u>
"Unvested Dutch Options"	has the meaning given in <u>clause 8.5(b);</u>
"Warranties"	means each of the Warranties set out in Schedule 8;
"Warrantors"	means each person listed in Schedule 9;
"Working Capital Excess"	means the amount (if any) by which the Actual Working Capital Amount exceeds the Working Capital Target;
"Working Capital Shortfall"	means the amount (if any) by which the Actual Working Capital Amount is less than the Working Capital Target;
"Working Capital Target"	means an amount notified in writing by Company to Qorvo on or after the conclusion of the Diligence Period, but in any event no later than [*****] days prior to Completion, being a reasonable and good faith estimate of the working capital requirements of the Company as at Completion of the Acquisition. In determining the Working Capital Target, Qorvo shall consult with the Company to determine a reasonable and good faith estimate; and
"Working Hours"	means 9.00 a.m. to 6.00 p.m. on a Business Day.

2. Meaning of Control

2.1 For the purposes of this Deed, "**Control**" means in relation to a person the direct or indirect power of another person (whether such other person is the direct or indirect parent company of

91

the first mentioned person or otherwise) to secure that the first mentioned person's affairs are conducted in accordance with the wishes of such other person:

- (A) by means of the holding of any shares (or any equivalent securities) or the possession of any voting power;
- (B) by virtue of any powers conferred on any person by the articles of incorporation or association or any other constitutional documents of any company or other entity of any kind; or
- (C) by virtue of any contractual arrangement,

and "Controlled" and "Controller" shall be construed accordingly.

- 2.2 A "Change of Control" shall occur if:
 - (A) a person who Controls another person ceases to do so;
 - (B) a different person acquires Control of such other person;
 - (C) any person acquires Control of another person in circumstances where no person previously Controlled such other person; or
 - (D) any person acquires all or substantially all of the assets of the Company or the Group in any one transaction or series of connected transactions.
- 2.3 For the purposes of this <u>paragraph 2</u> (and in connection with the use in this Deed of the terms defined in this <u>paragraph 2</u>) all of the members of any consortium, partnership or joint venture which has any interest (direct or indirect) in any Shareholder shall be deemed to be one person.
- 2.4 For the purposes of this paragraph 2 (and in connection with the use in this Deed of the terms defined in this paragraph 2), Qorvo will not be deemed to Control the Company as a result of the Series F Financing, the entering into of this Deed or the other Transaction Documents.

SCHEDULE OF SHAREHOLDERS

SCHEDULE OF CONTINGENT SECURITIES

OFFER DOCUMENT

FORM OF ACCEPTANCE

CONDUCT OF BUSINESS

PART A: ACTS AND MATTERS REQUIRING CONSENT OF QORVO PRIOR TO EXERCISE OF THE OPTION

The acts and matters referred to in <u>clause 5</u> are as follows:

- 1. Any restructuring, refinancing, administration, scheme of arrangement or any other arrangement with the members and/or creditors of the Company.
- 2. Any amendment to the articles of association or equivalent organisational documents of any member of the Group, including the Articles of Association.
- 3. Any change to the rights attaching to any class of Shares or Contingent Securities in the Company or any member of the Group which are not set out in the Articles of Association or Employee Share Schemes (other than the amendments required pursuant to <u>clause 8.2</u> and <u>clause 8.5</u>).
- 4. Any application for the listing of Shares or other securities of the Company or any Subsidiary on any stock exchange or for permission for dealings in any Shares or other securities of the Company or any Subsidiary in any securities market.
- 5. Issue or sell any Contingent Securities to any person that is not a Committed Shareholder or Executive Optionholder, or any person who becomes a Committed Shareholder or Executive Optionholder by signing a Deed of Adherence in connection with such issuance or sale.
- 6. Any resolution to wind up the Company or any member of the Group.
- 7. The filing of a petition for winding up by the Company or any member of the Group or the making of any arrangement with its or their creditors generally or any application for an administration order or for the appointment of a receiver or administrator.
- 8. Any purchase or redemption of any share capital of the Company or any member of the Group, other than (a) from employees, officers, directors, consultants or other persons performing services for the Company or any Subsidiary pursuant to an agreement under which the Company has the right to repurchase such shares at cost (or the lesser of cost or the then fair market value) upon the occurrence of certain events, or pursuant to the exercise by the Company of any rights of first refusal with respect to such shares and (b) in accordance with the terms and conditions of this Deed.

- 9. The sale, issuance or grant of any equity of the Company or any of its Subsidiaries, including any Contingent Securities, except as set forth in <u>subclauses (a)</u> through (c) below:
 - (a) the Company may grant and issue Contingent Securities as provided in <u>clause 8.16</u> of this Deed;
 - (b) the Company may sell or issue any securities of the Company, including Convertible Securities pursuant to one or more bona fide capital raising transactions, provided, that any and all such transactions meet each of the following conditions:
 - (i) the aggregate amount raised in such transactions does not exceed \$20,000,000;
 - (ii) no such transaction is consummated prior to the one-year anniversary of the date of this Deed;
 - (iii) the pre-emptive rights of Qorvo to participate in such transaction shall not have been waived by the shareholders of the Company, and Qorvo shall have the right to acquire its pro rata portion of the securities sold in such transaction in accordance with the Articles, including Article 4.5 (unless such right has been waived by Qorvo);
 - (iv) the antidilution adjustment provisions under the Articles with respect to any Shares held by Qorvo shall not have been waived by the shareholders of the Company, and any Shares held by Qorvo shall be entitled, as applicable, to the antidilution adjustments in the Articles, including Article 5.3(e), with respect to the sale or issuance of securities in such transaction (unless such right has been waived by Qorvo);
 - (v) the rights, preferences and privileges of any securities issued shall not be senior to those of the Series F Preferred Shares;
 - (vi) any sale, issuance or grant of securities is subject to the provisions of this Deed, including <u>clause 3.2</u> of this Deed;
 - (vii) no securities are sold, issued or granted to any Qorvo Competitor;
 - (viii) no such transaction may be entered into or consummated after Qorvo has delivered an Option Notice to the Company; and

- (ix) all required consents and approvals of the Board of Directors and the shareholders of the Company, pursuant to the Articles of Association, Applicable Law or otherwise, are obtained prior to any such transaction; and.
- (c) the Company may issue any security issuable upon exercise or conversion of any Convertible Securities
 (i) outstanding as of the date of this Deed or (ii) issued or granted pursuant to <u>subclause (a)</u> or (b) above.
- 10. The repayment of capital or assets to members of the Company or the members of any Subsidiary.
- 11. The sale or acquisition of any Material Group Asset or the sale or acquisition of any material part of the Business.
- 12. The grant of any option or right of pre-emption in respect of the acquisition of any Material Group Company or Material Group Asset.
- 13. Any change to the tax residence of the Company or any member of the Group.
- 14. Any subscription for or acquisition by any other means of any interest (whether on its own behalf or as nominee) in the share capital or instruments convertible into the share capital (or the equivalent) of any other body corporate or undertaking or the incorporation of any subsidiary by any member of the Group.
- 15. Other than in the ordinary course of business, the entry by any member of the Group into any agreement or other arrangement for the sale, licensing, assignment or disposal of any rights in respect of patents, trademarks, copyrights, designs, technical information or other intellectual property.
- 16. The declaration of any dividends or distributions, other than in accordance with this Deed.
- 17. The authorisation, proposal or announcement of an intention to do any of the foregoing, or enter into any contract, agreement, commitment or arrangement to do any of the foregoing.
- 18. Without prejudice to the foregoing, any transaction or arrangement, or the entering into of any contract, agreement, commitment or arrangement which may result in any Leakage.

PART B: ACTS AND MATTERS REQUIRING CONSENT OF QORVO FOLLOWING EXERCISE OF THE OPTION

The acts and matters referred to in <u>clause 5</u> are as follows:

- 1. Each of the acts or matters listed in <u>Part A</u> of this <u>Schedule 6</u>.
- 2. The creation, consolidation, sub-division, conversion, reduction or cancellation of any share capital of the Company or any Subsidiary.
- 3. Any change in the corporate governance arrangements of the Company or any Subsidiary.
- 4. Any capital raisings of new equity capital by any member of the Group (including by way of any issue, allotment, an option, a right to subscribe or acquire or conversion of any security into any share capital in the Company or any Subsidiary) other than pursuant to any Employee Share Scheme in existence as at the date of this Deed (and as amended pursuant to <u>clause 8.2</u> and <u>clause 8.5</u>).
- 5. The entry by any member of the Group into any commercial transaction or contract, or the amendment of the terms of any commercial transaction or contract, with any Major Investor or Committed Shareholder or any person who is an Associate of any Major Investor or Committed Shareholder.
- 6. The raising of any indebtedness by the Group.
- 7. The entry by any member of the Group into any Material Agreement.
- 8. Any alteration to the Accounting Period of the Group or to the accounting policies of the Group unless such alteration is required by law or relevant accounting requirements.
- 9. Any release, modification or abrogation of any liabilities, obligations or covenants owed to the Company by any of the parties to this Deed, whether or not subject to any terms and conditions.
- 10. The initiation and subsequent conduct of any litigation by any member of the Group save for the collection of debts arising in the ordinary course of business or any interim injunction or other urgent applications in circumstances where it is not practicable to obtain prior consent.
- 11. The entry into any contract by any member of the Group other than on arm's-length terms.
- 12. Other than in the normal course of business, any loan or advance or the giving of any guarantee or indemnity or the provision of any credit by any member of the Group.

- 14. Any entry into any new loan or loan facility.
- 15. Any factoring of any of the debts of any member of the Group.
- 16. The entry by any member of the Group into any hire purchase, rental or leasing agreement, the total cost of which agreement may exceed an amount of USD \$100,000 (or its equivalent) over the life of the agreement.
- 17. The employment, dismissal or change to the terms of employment of any director, officer or Key Employee of any member of the Group.
- 18. Any increase in the total emoluments, pensions or other benefits or payments paid to or for the benefit of the past, present and future directors, officers or Key Employees of the any member of the Group, or their respective dependants.

ACTIONS ON COMPLETION

PART A: ACTIONS ON COMPLETION

At Completion, the Major Investors, the Executive Optionholders and the Committed Shareholders shall:

- 1. deliver to Qorvo a Tax Covenant in the agreed form duly executed by the Warrantors;
- 2. deliver to Qorvo a Deed of Release and Waiver in the form attached hereto as <u>Schedule 17</u>, executed by each of the Major Investors, Executive Optionholders and Committed Shareholders;

At Completion, the Company shall:

- 1. deliver to Qorvo (or by prior agreement with Qorvo make available to Qorvo (or to any person whom Qorvo may nominate) at the registered office of the relevant member of the Group):
 - (A) the statutory books (which shall be written up to but not including the Completion Date), the articles of association or other constitutional documents and common seal (if any) of each member of the Group and share certificates or other documents of title that are in the Company's control in respect of all the issued share capital of each Subsidiary which is owned directly or indirectly by the Company (together with duly executed transfers in respect of any shares in any of the Subsidiaries not held by the Company, insofar as they are not held by or to the order of a member of the Group);
 - (B) the Books and Records;
 - (C) copies (signed by the auditors) of the letters from the auditors of the Company and of each Subsidiary referred to in paragraph 2 below; and
 - (D) a DVD with the contents of the electronic Data Room.
- 2. procure the present directors, officers and secretary of each member of the Group (other than any director or secretary whom Qorvo may wish should continue in office) to resign their offices as such and to relinquish any rights which they may have under any contract of employment with any member of the Group or under any statutory provision (including any right to damages or compensation for breach of contract, loss of office, redundancy or unfair dismissal or on any other account whatsoever) and to confirm that no agreement or arrangement is outstanding under

102

which any member of the Group has or could have any obligation to any of them including in respect of remuneration or expenses, such resignations to be tendered at the board of directors meetings referred to in <u>paragraph 4</u> below;

- 3. procure the present auditors of the Company and each Subsidiary to resign their office as such to take effect as at the Completion Date, and to deposit at the registered office of the Company, or, as the case may be, the relevant Subsidiary, a letter notifying their resignation, acknowledging that they have no claim against the Company or, as the case may be, the relevant Subsidiary and containing a statement pursuant to the Companies Act that there are no circumstances connected with their ceasing to hold office which they consider should be brought to the attention of any members or creditors;
- 4. procure board of directors meetings of the Company and each Subsidiary to be held at which:
 - (A) in the case of the Company, it shall be resolved that each of the transfers relating to the Shares shall be approved for registration and (subject only to the transfer being duly stamped) each transferee registered as the holder of the Shares concerned in the register of members;
 - (B) each of the persons nominated by Qorvo shall be appointed directors and/or secretary, as Qorvo shall direct, such appointments to take effect on the Completion Date;
 - (C) the resignations of the directors and secretary referred to in paragraph 2 above shall be tendered and accepted so as to take effect at the close of the meeting;
 - (D) all existing instructions to banks shall be revoked and new instructions shall be given to such banks in such form as Qorvo may direct;
 - (E) the situation of the registered office shall be changed to such address as Qorvo may nominate and (subject to the provisions of the Companies Act) the accounting reference date shall be changed in accordance with any instructions given by Qorvo;
 - (F) in the case of the Company, service or employment agreements in the agreed form shall be approved and entered into by each Key Employee; and
 - (G) he execution by the Company (or any member of the Group) of each contract of employment with each Key Employee is approved and authorised.
- 5. procure that minutes of each of the board of directors meetings referenced in <u>paragraph 4</u>, certified as correct by the secretary of the relevant company and the resignations and

acknowledgements and service or employment agreements, referred to in this <u>Schedule 7</u> are delivered to Qorvo's Solicitors; and

6. procure that each Key Employee shall execute and deliver to the Company and Qorvo a service or employment agreement between such Key Employee and Qorvo or the Company (or any member of the Group) on terms and conditions which are approved by Qorvo.

PART B: CONDITIONS TO COMPLETION

The Offer shall at all times prior to Completion of the Acquisition be conditional on the following conditions being satisfied during the Offer Period or (in the case of certain conditions) prior to Completion of the Acquisition, or waived by Qorvo:

- 1. Qorvo not having given written notice to the Company that Qorvo does not wish to proceed with the purchase of the Shares and Contingent Securities pursuant to the Offer and this Deed. No such notice shall be given unless Qorvo becomes aware of any matter, fact or circumstance relating to the Company which Qorvo has determined in good faith, either alone or together with any other matter, fact or circumstance relating to the Company of which Qorvo is aware, has or is likely to have a Material Adverse Effect on the financial, legal or business condition of the Company or its businesses;
- 2. (a) each of the Warranties which are qualified as to materiality being true, accurate and not misleading in all respects when made and as at Completion of the Acquisition by reference to the facts and circumstances then existing (except for such Warranties that speak specifically as of a particular date, which shall be true, accurate and not misleading as of such date) and (b) each of the Warranties which are not qualified as to materiality being true, accurate and not misleading in all material respects when made and as at Completion of the Acquisition by reference to the facts and circumstances then existing (except for such Warranties that speak specifically as of a particular date, which shall be true, accurate and not misleading in all material respects when made and as at Completion of the Acquisition by reference to the facts and circumstances then existing (except for such Warranties that speak specifically as of a particular date, which shall be true, accurate and not misleading in all material respects as of such date);
- 3. the Offer having been open for not less than twenty (20) Business Days (or such longer period as may be required with respect to the Offer by the Financial Services and Markets Act 2000 (Financial Promotion) Order 2005 or Applicable Law) beginning with the day after the day on which the Offer was first communicated to the Committed Shareholders, the Major Investors, the Other Shareholders and any other person who holds Shares on the date of the Offer.
- 4. valid Forms of Acceptance being received (and not, where permitted, withdrawn) by the conclusion of the Offer Period in respect of:

- a. not less than [*****] of the entire issued Ordinary Shares and each class and series of Preferred Shares (as the case may be and in each case excluding those held by Qorvo), or such lesser percentage as Qorvo may decide, not being less than fifty per cent (50%) by nominal amount of the total Shares of the Company; or
- b. the Shares held by the Major Investors sufficient to trigger a Drag-Along Acquisition;
- 5. no Governmental Entity enacting, issuing, promulgating, enforcing or entering any Applicable Law (whether temporary, preliminary or permanent) that is then in effect and that enjoins, restrains, conditions, makes illegal or otherwise prohibits the consummation of the transactions contemplated by this Deed;
- 6. the waiting period (and any extension thereof) applicable to Completion under the HSR Act or similar foreign anti-trust Applicable Laws shall have expired or been terminated;
- 7. each Key Employee shall have executed and delivered to the Company and Qorvo a service or employment agreement between such Key Employee and Qorvo or the Company (or any member of the Group) on terms and conditions which have been approved by Qorvo;
- 8. the termination, effective immediately prior to Completion, by the Company or the applicable Subsidiary of the employment of each those Employees identified in writing by Qorvo at least 5 Business Days prior to the Completion; and
- 9. the Company shall have given the Required Notices and obtained the Required Consents.

WARRANTIES

1. Organisation, Standing and Power.

1.1 Each of the Company and its Subsidiaries is a corporation duly incorporated and/or organised, validly existing and in good standing under the Applicable Laws of the jurisdiction of its incorporation or organisation as set forth in Section 1.1 of the Disclosure Letter and has the requisite corporate power and authority to carry on its business as now being conducted. Each of the Company and its Subsidiaries is duly qualified to do business, and is in good standing in each jurisdiction where the character of its properties owned or held under lease or the nature of its activities makes such qualification necessary, except where the failure to be so qualified would not, either individually or in the aggregate, have a Material Adverse Effect. The Company has previously delivered to Qorvo accurate and complete copies of the articles of association or equivalent constitutional documents, each as currently in full force and effect, of the Company and each of its Subsidiaries. There have been no predecessor entities of the Company or any of its Subsidiaries. The minute books of each of the Company and its Subsidiaries have been made available for inspection by Qorvo prior to the date hereof and are true and materially complete. No order has been made, petition presented or resolution passed for the winding up of the Company or any of its Subsidiaries, and no administrator or any receiver or manager has been appointed by any person in respect of the Company or any of its Subsidiaries or all or any of their respective assets, and, so far as the Company is aware, no steps have been taken to initiate any such appointment and no voluntary arrangement has been proposed, and neither the Company nor any of its Subsidiaries has become subject to any analogous proceedings, appointments or arrangements under the laws of any applicable jurisdiction.

2. Capital Structure.

- 2.1 The Shares are legally and beneficially owned by the persons and in the amounts with the corresponding certificate numbers set forth on Schedule 2.1 included in the Data Room in folder "2.1 & 2.2 Authorized Capital and List of Shareholders" and delivered to Qorvo on or prior to the date of the Deed. There are no issued Ordinary Shares or Preferred Shares that are not listed on Schedule 2.1 included in the Data Room in folder "2.1 & 2.2 Authorized Capital and List of Shareholders" and delivered to Qorvo on or prior to the date of the Deed. There are no issued Ordinary Shares or Preferred Shares that are not listed on Schedule 2.1 included in the Data Room in folder "2.1 & 2.2 Authorized Capital and List of Shareholders" and delivered to Qorvo on or prior to the date of the Deed. All of the issued Shares have been duly authorised and validly allotted and issued in compliance with all Applicable Laws and are fully paid.
- 2.2 The Company has reserved [*****] Ordinary Shares for issuance to officers, directors, employees and consultants to the Company pursuant to the Employee Share Schemes duly adopted by the Board of Directors. Of such reserved shares, [*****] shares have been issued pursuant to the exercise of options, options to purchase [*****] shares have been granted and are currently outstanding, and [*****] shares remain available for issuance to officers, directors, employees and consultants pursuant to the Employee Share Schemes. Schedule 2.2 included in the Data Room in folder "2.3 Options, Warrants, Rights" (and delivered to Qorvo on or prior to the date of the Deed) contains a correct and complete list of each outstanding option to purchase Ordinary Shares issued under the Employee Share Schemes, including the holder, date of grant, term, acceleration of vesting or exercisability, if any, any restrictions on the exercise or sale of such option or the underlying shares (other than any restrictions set forth in the Employee Share Scheme), exercise price and number of Ordinary Shares subject thereto. Schedule 2.2 included in

the Data Room in folder "2.3 – Options, Warrants, Rights" (and delivered to Qorvo on or prior to the date of the Deed) contains a correct and complete list of each other interest convertible into or exchangeable for Shares, including the holder, date of grant, term, acceleration of vesting or exercisability, if any, exercise price, net exercisability and number of shares of Common Stock subject thereto. Other than the foregoing outstanding options and convertible securities and this Deed, there are no options, warrants, calls, rights or agreements to which the Company or any of its Subsidiaries is a party or by which it is bound obligating the Company to issue, deliver or sell, or cause to be issued, delivered or sold, any capital stock of or any equity interests in the Company or any of its Subsidiaries (including Shares), or obligating the Company or any of its Subsidiaries and all Shares to be issued pursuant to the exercise of convertible securities will be duly authorised and validly allotted and issued, and are fully paid in compliance with all Applicable Laws.

- 2.3 There are no outstanding contractual obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any Shares or any equity interests in the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries has any outstanding bonds, debentures, notes or other obligations the holders of which have the right to vote (or are convertible into or exercisable for securities having the right to vote) with the shareholders of the Company on any matter. Neither the Company nor any its Subsidiaries is party to or bound by (a) any agreement or commitment pursuant to which the Company or any of its Subsidiaries is or could be required to register any securities under any Applicable Laws or (b) any debt agreements or instruments which grant any rights to vote (contingent or otherwise) on matters on which shareholders of the Company may vote.
- 2.4 Except as set forth in Schedule 2.4, the Company and its Subsidiaries do not own or control and has never owned or controlled, directly or indirectly, fifty per cent (50%) or more of the stock or other equity interests of any other corporation, partnership, limited liability company, joint venture or other legal entity, either alone or through or together with any other Subsidiary.
- 2.5 Schedule 2.5 sets forth, for each Subsidiary of the Company, the amount of its authorised shares (if relevant) or other equity or ownership interests, the amount of its issued or outstanding shares or other equity or ownership interests and the record and beneficial holders of its issued or outstanding shares or other equity or ownership interests. Each of the Subsidiaries is wholly owned by the Company.
- 2.6 Other than the Transaction Documents and the [*****] Warrants, there are no stockholder agreements, voting trusts or other agreements or understandings to which the Company or any of its Subsidiaries is a party or by which it is bound relating to the voting or registration of any Shares.

3. Due Authorisation.

3.1 All corporate action on the part of the Company, its officers, directors, managers and shareholders necessary for (a) the authorisation, execution and delivery of, and the performance of all obligations of the Company under this Deed and the other Transaction Documents, (b) the filing of the Articles of Association and (c) the authorisation, allotment, issuance, reservation for issuance and delivery, as applicable, of the Series F Preferred Shares and as may be issued pursuant to the Additional Funding, being issued and sold under the Subscription Agreement, has been taken prior to entry into this Deed. The Transaction Documents have been executed and

delivered by the Company and are valid and legally binding obligations of the Company enforceable in accordance with their terms, subject, as to enforcement of remedies, to applicable bankruptcy, insolvency, moratorium, reorganisation and similar Applicable Laws affecting creditors' rights generally and to general equitable principles.

4. No Conflict; Required Filings and Consents.

- 4.1 The execution, delivery and performance by the Company of this Deed and each of the other Transaction Documents, and the consummation of the transactions contemplated hereby and thereby, do not and will not:
 - (a) conflict with or violate the Articles of Association or equivalent constitutional documents of the Company or any of its Subsidiaries;
 - (b) conflict with or violate any Applicable Law or by which any property or asset of the Company or any of its Subsidiaries is bound or affected; or
 - (c) result in any breach of, constitute a default (or an event that, with notice or lapse of time or both, would become a default) under, require any consent of any person pursuant to, give to others any right of termination, amendment, modification, acceleration or cancellation of, allow the imposition of any fees or penalties, require the offering or making of any payment or redemption, give rise to any increased, guaranteed, accelerated or additional rights or entitlements of any person or otherwise adversely affect any rights of the Company or any of its Subsidiaries under, or result in the creation of Liens on any property, asset or right of the Company or any of its Subsidiaries pursuant to, any note, bond, mortgage, indenture, agreement, lease, license, permit, franchise, instrument, obligation or other contract to which the Company or any of its Subsidiaries is a party or by which the Company, any of its Subsidiaries or any of its properties, assets or rights are bound or affected.
- 4.2 No person has any rights of first offer, rights of first refusal or other rights relating to the issuance of equity securities or consummation of the transaction contemplated by this Deed and the other Transaction Documents.
- 4.3 Neither the Company nor any of its Subsidiaries is required to file, seek or obtain any notice, authorisation, approval, order, permit or consent of or with any Governmental Entity in connection with the execution, delivery and performance by the Company of this Deed and the other Transaction Documents or the consummation of the transactions contemplated hereby or thereby or in order to prevent the termination of any right, privilege, license or qualification of the Company or any of its Subsidiaries, except for (a) the filing of the Articles of Association and (b) any filings required to be made under any acquisition notification filing Applicable Laws required with any Governmental Entity.
- 4.4 No "fair price," "interested stockholder," "business combination" or similar provision of any takeover Applicable Law is, or will be, applicable to the transactions contemplated by this Deed or the other Transaction Documents.

5. Accounts.

5.1 The Accounts and the Interim Accounts:

are correct and complete in all material respects and have been prepared in accordance with the books and records of the Company and its Subsidiaries;

- (a) have been prepared in accordance with UK GAAP consistently applied throughout the periods covered, except as noted in the Accounts;
- (b) reflect and provide in accordance with UK GAAP adequate reserves in respect of all known liabilities of the Company and its Subsidiaries, including all known contingent liabilities, as of such dates;
- (c) do not contain any items of a special or nonrecurring income or any other income not earned in the ordinary course of business except as expressly specified therein; and
- (d) present fairly the consolidated financial condition of the Company and its Subsidiaries at such dates and the consolidated results of their operations for the fiscal periods then ended.
- 5.2 Except as and to the extent adequately accrued or reserved against in the Company's balance sheet included in the Accounts or the Interim Accounts, the Company and its Subsidiaries have no liability or obligation of any nature, whether accrued, absolute, contingent or otherwise, whether known or unknown and whether or not required by UK GAAP to be reflected in a consolidated balance sheet of the Company and its Subsidiaries or disclosed in the notes thereto, except for liabilities and obligations incurred in the ordinary course of business consistent with past practice since the Accounts Date that are not, individually or in the aggregate, material to the Company or any of its Subsidiaries.
- 5.3 The Company and its Subsidiaries keep books, records and accounts that, in reasonable detail, accurately and fairly reflect (a) the transactions and dispositions of assets of the Company and its Subsidiaries, and (b) the value of inventory calculated in accordance with UK GAAP. The Company and its Subsidiaries have not, and no employee, agent or stockholder of the Company or any of its Subsidiaries (in their capacity as such or purporting to act in such capacity or on behalf of or in connection with the business or affairs of the Company or any of its Subsidiaries) has, directly or indirectly, made any payment of funds of the Company or any of its Subsidiaries or received or retained any funds in violation of any Applicable Law.

6. No Default.

Neither the Company nor any of its Subsidiaries is in breach, default or violation (and no event has occurred that with notice or the lapse of time or both would constitute a breach, default or violation) of any term, condition or provision of: (a) its Articles of Association or equivalent constitutional documents; (b) any Material Contract (as defined below); or (c) any Applicable Laws.

7. Absence of Certain Changes or Events.

Since the Accounts Date: (a) the Company and its Subsidiaries have conducted their business only in the ordinary course consistent with past practice; (b) there has not been any change, event or development that, individually or in the aggregate, has had, or would be reasonably expected to have, a Material Adverse Effect; (c) neither the Company nor any of its Subsidiaries has suffered any loss, damage, destruction or other casualty affecting any of its or their material properties or assets, whether or not covered by insurance; and (d) neither the Company nor any of its

Subsidiaries has taken any action that, if taken after the date of this Deed, would constitute a breach of any of the covenants set forth in <u>Schedule 6</u> of this Deed.

8. Permits and Compliance.

- 8.1 Each of the Company and its Subsidiaries is and at all times has been in compliance in all material respects with all Applicable Laws. None of the Company, any of its Subsidiaries or any of its or their executive officers has received since its incorporation, nor is there any reasonable basis for, any notice, order, complaint or other communication from any Governmental Entity or any other person that the Company or any of its Subsidiaries is not in compliance in any material respect with any Applicable Laws.
- 8.2 Each of the Company and its Subsidiaries is and at all times has been in possession of all material franchises, grants, authorisations, licenses, permits, easements, variances, exceptions, consents, certificates, approvals and orders of any Governmental Entity necessary for each of the Company and its Subsidiaries to own, lease and operate its properties or to carry on its business (the "**Company Permits**"), and no suspension or cancellation of any of the Company Permits is pending or, to the knowledge of the Company, threatened.
- 8.3 There are no contracts or agreements of the Company or any of its Subsidiaries having covenants not to compete that impair the ability of the Company or any of its Subsidiaries to conduct its business as currently conducted or that would apply to Qorvo following the consummation of the transactions contemplated by this Deed.
- 8.4 The Dutch Subsidiary or any third party has not done or agreed to do anything as a result of which
 - (a) any subsidy or government support received by the Dutch Subsidiary is or may be liable to be refunded in whole or in part; or
 - (b) any application made by the Dutch Subsidiary for such subsidy or government support shall or may be refused in whole or in part, and

neither this Deed nor any of the transactions contemplated by this Deed shall have any such result.

9. Tax Matters.

- 9.1 General Tax Matters
- 9.1.1 All notices, returns, reports, accounts, computations, statements, assessments, claims, disclaimers, elections and registrations and any other necessary information that have, or should have, been submitted by the Company or any Subsidiary to any Taxation Authority for the purposes of Taxation have been made on a proper basis, were submitted within applicable time limits and were accurate and complete in all material respects. None of the above is, or (so far as the Warrantors are aware) is likely to be, the subject of any material dispute with any Taxation Authority.
- 9.1.2 All Taxation for which the Company or any Subsidiary has been, or is, liable to account has been duly paid (insofar as such Taxation ought to have been paid) by the due dates and no penalties, fines, surcharges or interest have been incurred.

- 9.1.3 The Company and each Subsidiary maintain complete and accurate records, invoices and other information in relation to Taxation that meet all legal requirements and enable the tax liabilities of the Company and any Subsidiary to be calculated accurately in all material respects.
- 9.1.4 All Taxation deductible under any Taxation Statute has, so far as is required to be deducted, been deducted from all payments made (or treated as made) by the Company or any Subsidiary. All amounts due to be paid to the relevant Taxation Authority on or before the date of this Deed have been so paid.
- 9.1.5 Neither the Company nor any Subsidiary is, or so far as the Warrantors are aware will become, liable to make to any person (including any Taxation Authority) any payment in respect of any liability to Taxation which is primarily or directly chargeable against, or attributable to, any other person (other than the Company or any Subsidiary).
- 9.2 Capital Gains

The gross book value shown in, or adopted for the purposes of, the Accounts as the value of each of the tangible fixed assets of the Company does not exceed the amount which on a disposal of such asset at the date of this Deed would be deductible as acquisition cost in computing the capital gain or deductible loss.

- 9.3 [Reserved]
- 9.4 Distributions And Other Payments

No distribution or deemed distribution has been made (or deemed to have been made) by the Company or any Subsidiary, except dividends shown in their statutory accounts, and neither the Company nor any Subsidiary is bound to make any such distribution.

- 9.5 [Reserved]
- 9.6 Tax Groups And Fiscal Unities
- 9.6.1 The Disclosure Letter contains full particulars of:
- (a) all groups and consolidated groups for Taxation purposes and fiscal unities of which the Company or any Subsidiary is, or has been, a member within the last seven (7) years;
- (b) every agreement relating to the use of group relief or allowance to which the Company or any Subsidiary is, or has been, a party within the last seven (7) years; and
- (c) any arrangements for the payment of group Taxation liabilities to which the Company or any Subsidiary has ever been party.
- 9.6.2 All claims made by the Company and the Subsidiaries for group relief or allowance were valid when made and have been or will be allowed by way of relief from, or allowance or credit against, Taxation. All arrangements entered into by the Company and the Subsidiaries in relation to groups and consolidated groups for Taxation purposes and fiscal unities were valid when made and continue to be valid. The Company and the Subsidiaries have met all procedural and other requirements of all Taxation Statutes in respect of such claims, unities or groups.

- 9.6.3 Neither the execution of this Deed nor the exercise of the Option, nor any other event since the Accounts Date, will result in the clawback or disallowance of any group relief or allowance previously given.
- 9.7 Intangible Assets And Intellectual Property

For the purposes of this <u>paragraph 9.7</u>, references to intangible fixed assets mean intangible fixed assets, goodwill and intellectual property.

The Disclosure Letter contains full and complete details of the amount of expenditure on each of the intangible fixed assets of the Company and the Subsidiaries and provides the basis on which any deduction or allowance relating to that expenditure has been taken into account in the Accounts or, in relation to expenditure incurred since the Accounts Date, will be available to the Company or any Subsidiary. No circumstances have arisen since the Accounts Date by reason of which that basis might change.

- 9.8 Company Residence And Overseas Interests
- 9.8.1 The Company and the Subsidiaries have, throughout the past seven (7) years, been resident for Taxation purposes solely in the jurisdiction in which they are incorporated, and have not, at any time in the past seven (7) years, been treated as resident in any other national jurisdiction for the purposes of any double taxation arrangements.
- 9.8.2 Neither the Company nor any Subsidiary has, or within the last seven (7) years has had, a permanent establishment or taxable presence outside the national jurisdiction in which it is incorporated.
- 9.9 Transfer Pricing

All transactions or arrangements made by the Company or any Subsidiary have been made on arm's length terms and the processes by which prices and terms have been arrived at have, in each case, been fully documented. No notice, enquiry or adjustment has been made by any Taxation Authority in connection with any such transactions or arrangements.

9.10 Anti-Avoidance

Neither the Company nor any Subsidiary has been involved in any transaction or series of transactions the main purpose, or one of the main purposes of which, was the avoidance of Taxation, or any transaction that produced a loss for Taxation purposes with no corresponding commercial loss.

- 9.11 Indirect Taxes
- 9.11.1 The Company and the Subsidiaries are each taxable persons and are each registered for the purposes of Indirect Tax.
- 9.11.2 Neither the Company nor any Subsidiary is or has been, in the past seven (7) years, a member of a group for the purposes of Indirect Tax.

- 9.11.3 All supplies made by the Company or any Subsidiary are subject to Indirect Tax. Neither the Company nor any Subsidiary has been, or so far as the Warrantors are aware will be, denied full credit or allowance for all Indirect Tax paid or suffered by it.
- 9.12. Stamp Duties And Transfer Taxes
- 9.12.1 Any document that may be necessary in proving the title of the Company or any Subsidiary to any asset which is owned by the Company or any Subsidiary at Completion is duly stamped for stamp duty purposes or has had the transfer or registration tax due in respect of it paid.
- 9.12.2 Neither entering into this Deed nor Completion will result in the withdrawal of any stamp duty or transfer or registration tax relief granted on or before Completion which will affect the Company or any Subsidiary.
- 9.12.3 The Disclosure Letter contains full and accurate details of any asset acquired or held by the Company or any Subsidiary in respect of which the Warrantors are aware, or ought reasonably to be aware, that an additional Taxation return will be required to be filed with a Taxation Authority and/or a payment of stamp duty or transfer or registration tax made in the event of the exercise of the Option.

10. Actions and Proceedings.

There are no outstanding orders, judgments, injunctions, awards or decrees of any Governmental Entity against or involving the Company or any of its Subsidiaries, or against or involving any of the present or former directors, officers, managers, employees, consultants, agents or shareholders of the Company or any of its Subsidiaries, as such, any of its or their properties, assets or business or any Employee Share Scheme, nor is there any basis for any such action. There are no actions, suits or claims or legal, administrative or arbitrative proceedings or investigations (including claims for workers' compensation) pending or, to the knowledge of the Company, threatened against or involving the Company, any of its Subsidiaries or any of its or their present or former directors, officers, managers, employees, consultants, agents or shareholders, as such, or any of the properties, assets or business of the Company or any of its Subsidiaries or any Employee Share Scheme. There are no actions, suits or claims or legal, administrative or claims or legal, administrative proceedings or investigations pending or, to the knowledge of the Company, threatened seeking to prevent, hinder, modify, delay or investigations pending or, to the knowledge of the Company, threatened seeking to prevent, hinder, modify, delay or challenge the transactions contemplated by this Deed or the other Transaction Documents. There are no actions, suits or claims or legal, administrative or any of its Subsidiaries pending, or which the Company or any of its Subsidiaries have commenced preparations to initiate, against any other person.

11. Certain Agreements.

11.1 Neither the Company nor any of its Subsidiaries is a party to any oral or written agreement, program, plan or other arrangement relating to the compensation of employees of the Company or any of its Subsidiaries, including any employment agreement, severance agreement, stock option plan, stock appreciation rights plan, restricted stock plan or stock purchase plan, pension plan or welfare plan (collectively, the "**Compensation Agreements**"), any of the benefits of which will be increased, or the vesting of the benefits of which will be accelerated, by the occurrence of any of the transactions contemplated by this Deed or any of the other Transaction Documents or the value of any of the benefits of which will be calculated on the basis of any of the transactions

contemplated by this Deed or any of the other Transaction Documents. Section 11.1(a) of the Disclosure Letter sets forth the total amount of indebtedness owed to the Company or any of its Subsidiaries from each officer, director or employee of the Company or any of its Subsidiaries. Section 11.1(b) of the Disclosure Letter sets forth, as of the Completion Date, for each officer, director or employee of the Company or any of its Subsidiaries who is a party to, or will receive benefits under, any Compensation Agreement as a result of the transactions contemplated by this Deed or any of the other Transaction Documents, the total amount that each such officer, director or employee may receive, or is eligible to receive.

- 11.2 Set forth in Section 11.2 of the Disclosure Letter is a list of all Material Contracts (as defined below) to which the Company or any of its Subsidiaries is, or proposes to become, a party or to which any of its assets are, or are proposed by the Company or any of its Subsidiaries to be, bound. The Company has provided true and complete copies (or true and complete summaries with respect to oral contracts) of all such Material Contracts to Qorvo. "**Material Contracts**" means any of the following contracts, agreements or arrangements and any contracts, arrangements or agreements set forth in Section 11.2 of the Disclosure Letter, whether written or oral:
 - (a) any contract or commitment that involves a dollar amount in excess of USD \$[*****] over the term of such contract;
 - (b) any employment or consulting contracts with employees or consultants, including any investigator agreement, fee agreement or other contract with any physician;
 - (c) any contract with sales or other agents, brokers, franchisees, dealers or distributors;
 - (d) any partnership, joint venture agreement, joint development or similar arrangement;
 - (e) any lease or other occupancy or real estate use agreements;
 - (f) any options or rights of first refusal in or relating to the business of the Company or any of its Subsidiaries;
 - (g) any agreements giving any party the right to renegotiate or require a reduction in price or refund of payments previously made in connection with the business of the Company or any of its Subsidiaries;
 - (h) any agreements for the borrowing or lending of money with respect to the business of the Company or any of its Subsidiaries, any guaranty agreement and any other evidence of indebtedness;
 - (i) any agreement granting any person a Lien on any of the material assets of the Company or any of its Subsidiaries;
 - (j) any bonus, executive or deferred compensation, profit sharing, pension or retirement, stock option or stock purchase, hospitalisation, insurance, medical reimbursement or other plan, agreement or arrangement or practice providing employee or executive benefits to any officer or employee or former officer or former employee of the Company or any of its Subsidiaries;

- (k) any contract granting a license to or from the Company or any of its Subsidiaries for any Intellectual Property;
- (1) any material agreements that contain any provisions requiring the Company or any of its Subsidiaries to indemnify any other party thereto;
- (m) any non-competition, secrecy, confidentiality or other contract, in each case that restricts the right of the Company or any of its Subsidiaries to conduct the its or their business in any material respect at any time, in any manner or at any place in the world, or the expansion thereof to other geographical areas, customers, suppliers or lines of business, or that grants the other party or any third person "most favoured nation" status;
- any agreement purporting to grant a right to manufacture, produce, assemble, license, market or sell the Products or affecting the Company's or any of its Subsidiaries' exclusive right to develop, manufacture, assemble, distribute, market or sell the Products; or
- (o) any other agreement, arrangement or contract, whether oral or written, which could be considered material to the conduct of the business of the Company or any of its Subsidiaries.
- 11.3 Each Material Contract is a legal, valid and binding agreement of the Company or the Company's Subsidiary party thereto; the Company and its Subsidiaries (and to the knowledge of the Company, any other party thereto) are not in default under any Material Contract, and none of such Material Contracts has been cancelled by the other party thereto; each Material Contract is in full force and effect and no event has occurred which, with the passage of time or the giving of notice or both, would constitute a default, event of default or other breach by the Company or any of its Subsidiaries which would entitle the other party to such Material Contract to terminate the same or declare a default or event of default thereunder; and neither the Company nor any of its Subsidiaries is in receipt of any claim of default under any such agreement.

12. Employee Benefits.

- 12.1 Each Company Plan is listed in Section 12.1 of the Disclosure Letter. With respect to each Company Plan, the Company has made available to Qorvo a true and correct copy of:
 - (a) all annual reports (Form 5500) filed with the applicable government agency for the three (3) most recent plan years;
 - (b) each such Company Plan that has been reduced to writing and all amendments thereto;
 - (c) each trust agreement, insurance contract or administration agreement relating to each such Company Plan;
 - (d) a written summary of each unwritten Company Plan;
 - (e) the most recent summary plan description or other written explanation of each Company Plan provided to participants;
 - (f) the three most recent actuarial reports or valuations, if any, relating to each Company Plan;

- (g) the most recent determination letter or opinion letter, if any, issued by the IRS with respect to any Company Plan intended to be qualified under Section 401(a) of the Code;
- (h) any request or application for a determination currently pending before the IRS;
- (i) all material correspondence during the three (3) prior years with the IRS, the Department of Labor, or Pension Benefit Guaranty Corporation relating to any Company Plan;
- (j) the defined contribution Company Plan applicable to the Dutch Subsidiary insured with insurance company Achmea Pensioen- en Levensverzekeringen N.V. (called 'Flexibel Renteniers Plan') from 1 January 2015;
- (k) the defined benefit Company Plan of the Dutch Subsidiary;
- (l) any other written or oral individual and/or collective and contractual and/or statutory retirement benefit scheme (including any schemes in relation to pre-pension, early retirement or similar benefits payable on or following retirement, disability or death) for or in respect of any present and or former employee or managing director for any of the Dutch Subsidiary and/or their spouses or other dependants, including details of the financing arrangements made with the respective pension providers or applicable by Applicable Law;
- (m) the Share Option Plans (dated 22 July 2003 and 1 April 2007) for the employees of the Dutch Subsidiary and the Equity Incentive Plan (2007) for the other employees of the Company or any of its Subsidiaries other than the Dutch Subsidiary; and
- (n) any other terms of benefits policies, share incentive schemes, share option schemes, profit share schemes, bonus arrangements, or any other arrangements with respect to variable remuneration components, and any outstanding entitlements or conditional entitlements thereunder applicable to employees of the Dutch Subsidiary.

Each Company Plan complies in form and has complied in operation in all material respects with ERISA, the Code and all other applicable statutes and governmental rules and regulations.

12.2 No Company Plan has participated in, engaged in or been a party to any transaction that is prohibited under Section 4975 of the Code or Section 406 of ERISA that is not exempt under Section 4975 of the Code or Section 408 of ERISA, respectively which could result in liability to the Company. With respect to any Company Plan, (a) neither the Company nor any of its ERISA Affiliates has had asserted against it any claim for taxes under Chapter 43 of Subtitle D of the Code and Section 5000 of the Code or for penalties under ERISA Sections 502(c), 502(i) or 502(l), nor, to the knowledge of the Company, is there a basis for any such claim, and (b) to the knowledge of the Company, no officer, director or employee of the Company or any of its Subsidiaries has committed a breach of any fiduciary responsibility or obligation imposed by Title I of ERISA. Except for routine contributions due and owing and benefits due, with respect to the Company Plans, to the knowledge of the Company or any ERISA Affiliate could be subject to any material liability under the terms of such Company Plans, ERISA, the Code or any other Applicable Law. All Company Plans that are intended to be qualified under Section 401(a) of the Code have either (i) been determined by the IRS to be so qualified, or a timely application for such determination is now pending or (ii) are documented on

the form of an IRS pre-approved document, and to the knowledge of the Company, no event has occurred and there exists no condition or set of circumstances that would be reasonably expected to result in the loss of such qualified status. Neither the Company nor any of its ERISA Affiliates has any liability or obligation under any welfare plan to provide benefits after termination of employment to any employee or dependent other than as required by Section 4980B of the Code or applicable under state Applicable Law. The Company and its ERISA Affiliates have complied in all material respects with the provisions of Part 6 of Title I of ERISA and Sections 4980B, 9801, 9802, 9811 and 9812 of the Code.

- As used herein, (a) "Company Plan" means a "pension plan" (as defined in Section 3(2) of ERISA (including a Company 123 Multiemployer Plan, whether or not subject to ERISA)), a "welfare plan" (as defined in Section 3(1) of ERISA, whether or not subject to ERISA), any written or oral individual and/or collective and contractual and/or statutory retirement benefit scheme (including any schemes in relation to pre-pension, early retirement or similar benefits payable on or following retirement, disability or death) for or in respect of any present and or former employee or managing director for any of the Dutch Subsidiary and/or their spouses or other dependents, including details of the financing arrangements made with the respective pension providers or applicable by Applicable Law, and any other written or oral bonus, profit sharing, deferred compensation, incentive compensation, stock ownership, stock purchase, stock option, phantom stock, restricted stock, stock appreciation right, holiday pay, vacation, severance, medical, dental, vision, disability, death benefit, sick leave, fringe benefit, personnel policy, insurance or other material plan, program, agreement, or arrangement, in each case sponsored, established or maintained by the Company or any of its ERISA Affiliates or as to which the Company or any of its ERISA Affiliates otherwise may have any liability to any person that is providing, has provided or may in the future provide services to the Company or any of its ERISA Affiliates, (b) "Company Multiemployer Plan" means a "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA) to which the Company or any of its ERISA Affiliates is obligated to contribute or otherwise may have any liability, and (c) "ERISA Affiliate" means any trade or business (whether or not incorporated) which would be considered a single employer with the Company or any of its Subsidiaries pursuant to Section 414(b), (c), (m) or (o) of the Code and the regulations promulgated under those sections or pursuant to Section 4001(b) of ERISA and the regulations promulgated thereunder.
- 12.4 No Company Plan is subject to Section 302 or Title IV of ERISA or Section 412 or Section 430 of the Code, and neither the Company nor any ERISA Affiliate has any liability (contingent or otherwise) with respect to any such Company Plan. Neither the Company nor any of its ERISA Affiliates sponsors or has ever sponsored, maintained, contributed to, or incurred an obligation to contribute or incurred a liability (contingent or otherwise) with respect to any Company Multiemployer Plan, Multiple Employer Plan or pension plan that is subject to Section 302 or Title IV of ERISA or Section 412 of the Code. For these purposes, "Multiple Employer Plan" means any employee benefit plan sponsored by more than one employer, within the meaning of Sections 4063 or 4064 of ERISA or Section 413(c) of the Code.
- 12.5 Section 12.5 of the Disclosure Letter contains a list of all:
 - (a) severance, termination pay, change in control and employment agreements with employees of the Company and each Subsidiary;
 - (b) severance programs and policies of the Company and each Subsidiary with or relating to its employees; and

- (c) plans, programs, agreements and other arrangements of the Company and each ERISA Affiliate with or relating to its employees containing change of control or similar provisions.
- 12.6 Neither the Company nor any of its Subsidiaries is a party to any agreement, contract or arrangement that would reasonably be expected to result, separately or in the aggregate, in the payment, acceleration or enhancement of any benefit as a result of the transactions contemplated in this Deed or any of the other Transaction Documents (either alone or in combination with any other event) including the payment of any "excess parachute payments" within the meaning of Section 280G of the Code.
- 12.7 Each Company Plan that is a "nonqualified deferred compensation plan" that is subject to Section 409A of the Code is in material compliance with and has materially complied in form and operation with the requirements of Section 409A of the Code. Neither the Company nor any of its Subsidiaries has any obligation to provide any tax gross-up of any tax or interest charge imposed pursuant to Section 409A of the Code.
- 12.8 Except as designated in Section 12.1 of the Disclosure Letter, there is no Company Plan that is subject to the Applicable Laws of a foreign government or jurisdiction.
- 12.9 The Company Plans referred to in Sections 12.1(j), 12.1(k) and 12.1(l) of this <u>Schedule 8</u> are fully funded through a pension fund or third party insurance in accordance with the applicable pension agreements and Applicable Laws. All financial obligations of the Dutch Subsidiary due in respect of the Company Plans referred to in Sections 12.1(j), 12.1(k) and 12.1(l) of this <u>Schedule 8</u> (including past service liabilities) have been fully satisfied within the appropriate time period allowed. All Company Plans referred to in Sections 12.1(j), 12.1(k) and 12.1(l) of this <u>Schedule 8</u> are in line with Applicable Laws and market practice.
- 12.10 All employees of the Dutch Subsidiary have been registered with the correct applicable Company Plans referred to in Sections 12.1(j), 12.1(k) and 12.1(l) of this <u>Schedule 8</u>.

13. Product Liability.

- 13.1 Neither the Company nor any of its Subsidiaries has received a claim for, or based upon, breach of product or service warranty or guaranty or similar claim, strict liability in tort, negligent design of product, negligent provision of services or any other allegation of liability, including or arising from the materials, design, testing, manufacture, packaging, labelling (including instructions for use), or sale of the Products or from the provision of services; and, to the knowledge of the Company, there is no basis for any such claim.
- 13.2 As of the Completion Date, the Company shall have provided in Section 13.2 of the Disclosure Letter a schedule of products in development and planned introductions. To the knowledge of the Company, as of the Completion Date, there are no existing circumstances that would prevent the goals set forth therein from being achieved in all material respects. As of the Completion Date, the product and service engineering, development, manufacturing and quality control processes which have been and are being followed by the Company and its Subsidiaries are reasonably designed to produce products and services which (a) are consistent with the claims and statements made about them by or on behalf of the Company and its Subsidiaries, (b) comply with applicable regulatory requirements, and (c) avoid claims of the type described in <u>clause 13.1</u>.

14. Labor Matters.

- 14.1 Neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement or labour contract. Neither the Company nor any of its Subsidiaries have any obligation to notify any employee or service provider regarding the transactions contemplated by this Deed prior to engaging in any such transactions. The Company and its Subsidiaries have not engaged in any unfair labour practice with respect to any persons employed by or otherwise performing services primarily for the Company or any Subsidiary (the "**Company Business Personnel**"), and there is no unfair labour practice complaint or grievance against the Company or any of its Subsidiaries or by any person pursuant to the National Labor Relations Act of the United States or any comparable state agency or foreign Applicable Law pending or, to the knowledge of the Company, threatened with respect to the Company Business Personnel. There is no labour strike, dispute, slowdown or stoppage pending or, to the knowledge of the Company, threatened against or affecting the Company or any of its Subsidiaries that may interfere with the business activities of the Company or any Subsidiary. The Company and its Subsidiaries have complied in all material respects with all Applicable Laws relating to the employment of labour.
- 14.2 The Dutch Subsidiary has in relation to each of its current and former employees, trade unions, its current and former pension providers and any bodies representing employees (if any) at all times complied with its obligations under any Applicable Laws, collective bargaining agreements, individual employment agreements, reorganisation plans, social plans, executionary agreements and other agreements.
- 14.3 The Dutch Subsidiary has at all times punctually satisfied its payment obligations to current and former employees and to any relevant governmental or other social security agencies or institutions.
- 14.4 There has not been any accident, illness or injury which is not fully recovered or recoverable by insurance that is likely to give rise to any liability for the Dutch Subsidiary vis-à-vis any current or former employee, independent contractor or any other third party.
- 14.5 There has been no dispute between the Dutch Subsidiary and any of its current or former employees or current or former pension providers and there are no circumstances that are likely to give rise to any such dispute.

15. Intellectual Property.

15.1 As used herein, the term "Intellectual Property" means all intellectual property rights arising under Applicable Law: (a) trade names, trademarks and service marks (whether registered or unregistered), domain names, trade dress and similar rights and applications to register any of the foregoing (collectively, "Marks"); (b) patents and patent applications and rights in respect of utility models or industrial designs (collectively, "Patents"); (c) copyrights (whether registered or unregistered) and registrations and applications therefor (collectively, "Copyrights"); (d) know-how, inventions, discoveries, methods, processes, techniques, methodologies, formulae, algorithms, technical data, specifications, research and development information, technology, and other proprietary or confidential information, including customer lists, in each case, that derives economic value (actual or potential) from not being generally known to other persons who can obtain economic value from its disclosure, but excluding any Copyrights, Patents or Marks that cover or protect any of the foregoing (collectively, "Trade Secrets"); and (e) mask work and similar rights protecting integrated circuit or chip topographies or designs (registered or unregistered) and registrations and applications therefor (collectively, "**Mask Works**").

- 15.2 Section 15.2.1 of the Disclosure Letter sets forth an accurate and complete list of all registered Marks and applications for registration of Marks owned by the Company or any of its Subsidiaries (collectively, the "Company Registered Marks"), Section 15.2.2 of the Disclosure Letter sets forth an accurate and complete list of all Patents owned by the Company or any of its Subsidiaries (collectively, the "Company Patents"), and Section 15.2.3 of the Disclosure Letter sets forth an accurate and complete list of all registered Copyrights and all pending applications for registration of Copyrights owned by the Company or any of its Subsidiaries (collectively, the "Company Registered Copyrights" and, together with the Company Registered Marks and the Company Patents, the "Company Registered IP"). No Company Registered IP has been or is now involved in any derivation, interference, reexamination, opposition or cancellation proceeding and, to the knowledge of the Company, no such action is or has been threatened with respect to any of the Company Registered IP. All Company Registered IP has been registered or obtained in accordance with all applicable legal requirements and is currently in compliance in all material respects with all legal requirements (including the timely post-registration filing of affidavits of use and incontestability and renewal applications) other than any requirement that, if not satisfied, would not result in a cancellation of any such Company Registered IP or otherwise materially adversely affect the priority, validity and enforceability of such Company Registered IP. The Company Registered IP is subsisting and, to the knowledge of the Company, valid and enforceable, and no written notice or claim challenging the validity or enforceability or alleging the misuse of any of the Company Registered IP has been received by the Company or any of its Subsidiaries. The Company and its Subsidiaries have not taken any action or failed to take any action that could reasonably be expected to result in the abandonment, cancellation, forfeiture, relinquishment, invalidation or unenforceability of any of the Company Registered IP. All filing, examination, issuance, post registration and maintenance fees, annuities and the like associated with or required with respect to any of the Company Registered IP have been timely paid.
- 15.3 The Company and its Subsidiaries have taken all reasonable endeavours to protect its rights in the Intellectual Property owned by the Company or any of its Subsidiaries and have maintained the confidentiality of all of the Trade Secrets of the Company or any of its Subsidiaries. All current or former employees, consultants and contractors of the Company or any of its Subsidiaries who have participated in the creation of any Intellectual Property that is used by the Company or any of its Subsidiaries have entered into proprietary information, confidentiality and assignment agreements substantially in the Company's standard form (which has previously been provided to Qorvo). The Company is not aware that any of its employees, consultants or contractors are in violation thereof.
- 15.4 The Company or one of its Subsidiaries owns exclusively all right, title and interest to the Company Registered IP, free and clear of any Lien or other adverse claims or interests, and neither the Company nor any of its Subsidiaries has received any written notice or claim challenging the Company's or any of its Subsidiaries' ownership of any of such Intellectual Property. The Company or one of its Subsidiaries has a valid right pursuant to a valid and enforceable written license agreement to all other Intellectual Property used or planned to be used by the Company or any of its Subsidiaries, free and clear of any Lien or other adverse claims or interests, and neither the Company nor any of its Subsidiaries has received any written notice or claim challenging the Company's or any of its Subsidiaries' rights to any of such Intellectual Property. None of the Intellectual Property owned by the Company or any of its Subsidiaries that

is used by the Company or any of its Subsidiaries in the conduct of its business is subject to any outstanding order, judgment, or stipulation restricting the use thereof by the Company or such Subsidiary. No loss, impairment or expiration of any of the Intellectual Property owned by the Company or any of its Subsidiaries and used by the Company or any of its Subsidiaries in the conduct of its business is pending or, to the knowledge of the Company, threatened or reasonably foreseeable.

- 15.5 Section 15.5.1 of the Disclosure Letter sets forth a complete and accurate list of all agreements granting to the Company or any of its Subsidiaries any right under or with respect to any Intellectual Property owned by a third party that is used in connection with the business of the Company or any of its Subsidiaries other than any commercially available standard desktop software applications used generally in the Company's or any of its Subsidiaries' operations that are licensed for an aggregate license fee of no more than USD \$100,000 (collectively, the "Inbound License Agreements"), indicating for each the title and the parties thereto. Section 15.5.2 of the Disclosure Letter sets forth a complete and accurate list of all license agreements under which the Company or any of its Subsidiaries grants any rights under any Intellectual Property, excluding non-exclusive, end user licenses granted by the Company or any of its Subsidiaries in the ordinary course of business to purchasers of the Products in which any software is embedded. There is no outstanding or threatened dispute or disagreement with respect to any Inbound License Agreement or any license agreements under which the Company or any of its Subsidiaries grants any rights under any Intellectual Property (collectively, the "Outbound License Agreements") that could materially affect any of the respective rights and obligations of the parties thereunder. Except as set forth in Section 15.5.3 of the Disclosure Letter, the execution, delivery and performance by the Company of this Deed and the other Transaction Documents, and the consummation of the transactions contemplated hereby and thereby, will not result in a loss or impairment of, or give rise to any right of any third party to terminate or re-price or otherwise modify any of the Company's or any of its Subsidiaries' rights or obligations under, any Inbound License Agreement or any Outbound License Agreement. Except as set forth in Section 15.5.4 of the Disclosure Letter, the rights licensed under each Inbound License Agreement shall be exercisable by the Company or the Subsidiary thereof that is the party thereto on and after Completion to the same extent as by the Company or such Subsidiary, as applicable, prior to Completion. To the knowledge of the Company, no loss, impairment or expiration of any Intellectual Property licensed under any Inbound License Agreement is pending, reasonably foreseeable or threatened. Neither the Company nor any of its Subsidiaries has granted to any third party any exclusive rights under any Intellectual Property owned by the Company or such Subsidiary.
- 15.6 The Intellectual Property owned by or licensed (pursuant to a valid written license agreement) to the Company or one of its Subsidiaries constitutes all the Intellectual Property rights necessary for the conduct of the Company's and its Subsidiaries' business as it is currently conducted and as currently proposed to be conducted. Upon Completion, Qorvo shall succeed to all of the Intellectual Property rights held by the Company and its Subsidiaries for use in the conduct of the Company's and its Subsidiaries' business as it is currently conducted and as currently proposed to be conducted, and to the knowledge of the Company, all of such rights shall be exercisable by Qorvo to the same extent as by the Company and its Subsidiaries, as applicable prior to Completion.
- 15.7 Neither the conduct of the business of the Company or its Subsidiaries as currently conducted or as currently proposed to be conducted nor the Products, nor any technology or materials sold,

distributed or otherwise commercially exploited by or for the Company or any of its Subsidiaries, in any material respect, has infringed upon, misappropriated or violated or will infringe, misappropriate or violate any Intellectual Property of any third party, and neither the Company nor any of its Subsidiaries has received any written notice or claim or, to the knowledge of the Company, oral notice or claim asserting or suggesting that any such infringement, misappropriation, violation, unfair competition or trade practices has occurred, nor, to the knowledge of the Company, is there any reasonable basis therefor. To the knowledge of the Company, no third party is, in any material respect, misappropriating or infringing any Intellectual Property owned by the Company or any of its Subsidiaries and no third party has made any unauthorised disclosure of any Trade Secrets of the Company or any of its Subsidiaries.

- 15.8 No source code of any computer software owned by the Company or any of its Subsidiaries has been licensed or otherwise provided to another person other than an escrow agent pursuant to the terms of a source code escrow agreement in customary form, and the Company and its Subsidiaries have taken reasonable steps to protect all such source code as a Trade Secret of the Company or one of its Subsidiaries. Except as set forth in Section 15.8 of the Disclosure Letter, no software embedded in any of the Products contains any code that is owned by any third party, including any code that is licensed pursuant to the provisions of any "open source" license agreement, or any other license agreement that requires source code be generally distributed or made publicly available in connection with the distribution of the licensed software in object code form or that limits the amount of fees that may be charged in connection with sublicensing or distributing such licensed software (each, an "**Open Source License**"). None of the Products in which software is embedded, as a result of the intermingling or integration of proprietary code owned by the Company or any of its Subsidiaries with any "open source" software licensed under any Open Source License is, in whole or in part, subject to the provisions of any Open Source License.
- 15.9 To the knowledge of the Company, at no time during the conception of or reduction to practice of any Intellectual Property owned or developed by the Company or any of its Subsidiaries was any developer, inventor or other contributor to such Intellectual Property operating under any grants from any Governmental Entity or private source, performing research sponsored by any Governmental Entity or private source or subject to any employment agreement or invention assignment or non-disclosure agreement or other obligation with any third party that could adversely affect the Company's or any of its Subsidiaries' rights in such Intellectual Property. To the knowledge of the Company, there exist no inventions by current or former employees or consultants of the Company or any of its Subsidiaries made or otherwise conceived prior to their beginning employment or consultation with the Company that have been or will be incorporated into any of the Company's or any of its Subsidiaries' Intellectual Property or Products and to which the Company or such Subsidiary does not have rights to use in connection with such Intellectual Property or Products.

16. Inventories.

As of the Completion Date, (a) all inventories of the Company and its Subsidiaries consist of items of merchantable quality and quantity usable or saleable (free of any material defect or deficiency) in the ordinary course of business, are saleable at prevailing market prices that are not less than the book value amounts thereof or the price customarily charged by the Company and its Subsidiaries therefor, conform to the specifications established therefor, and have been manufactured in accordance with applicable regulatory requirements; and (b) the quantities of all inventories, materials, and supplies of the Company and its Subsidiaries (net of the obsolescence

reserves therefor shown in the Accounts and determined in the ordinary course of business consistent with past practice) are not obsolete, damaged, slow-moving, defective, or excessive, and are reasonable and balanced in the circumstances of the Company and its Subsidiaries.

17. Environmental Matters.

- 17.1 For purposes of this Deed, the following terms shall have the following meanings: (a) "Hazardous Substances" means (i) petroleum and petroleum products, by-products or breakdown products, radioactive materials, asbestos-containing materials and polychlorinated biphenyls, and (ii) any other chemicals, materials or substances regulated as toxic or hazardous or as a pollutant, contaminant or waste under any applicable Environmental Law; (b) "Environmental Law" means any Applicable Law and any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, or common law, relating to pollution or protection of the environment, health or safety or natural resources, including those relating to the use, handling, transportation, treatment, storage, disposal, release or discharge of Hazardous Substances; and (c) "Environmental Permit" means any permit, approval, identification number, license or other authorisation required under any applicable Environmental Law.
- 17.2 The Company and its Subsidiaries are, and have at all times been, in compliance with all Environmental Laws, has obtained all Environmental Permits and are, and have at all times operated, in compliance with their requirements, and has resolved all past non-compliance with Environmental Laws and Environmental Permits without any pending, ongoing or future obligation, cost or liability, except in each case for the notices set forth in Section 17.2 of the Disclosure Letter.
- 17.3 There is no pending or threatened investigation by any Governmental Entity, nor any pending or threatened action with respect to the Company or any of its Subsidiaries relating to Hazardous Substances or otherwise under any Environmental Law.
- 17.4 No Environmental Law imposes any obligation upon the Company or any of its Subsidiaries arising out of or as a condition to any transaction contemplated by this Deed, including any requirement to modify or to transfer any permit or license, any requirement to file any notice or other submission with any Governmental Entity, the placement of any notice, acknowledgment or covenant in any land records, or the modification of or provision of notice under any agreement, consent order or consent decree.
- 17.5 There are no environmental assessments or audit reports or other similar studies or analyses in the possession or control of the Company or any of its Subsidiaries relating to any real property currently or formerly owned, leased or occupied by the Company or any of its Subsidiaries, except in each case for those set forth in Section 17.5 of the Disclosure Letter.

18. Suppliers and Distributors.

18.1 Neither the Company nor any of its Subsidiaries has received any notice, oral or written, and none of them has any reason to believe, that any significant supplier, including any sole source supplier, will not sell raw materials, supplies, merchandise and other goods to the Company and its Subsidiaries at any time after the Completion or after the consummation of the transactions contemplated by this Deed on terms and conditions substantially similar to those used in its current sales to the Company and its Subsidiaries, subject only to general and customary price

increases, except for such raw materials, supplies, merchandise, or other goods as are readily available from other sources on comparable terms and conditions.

18.2 As of the Completion Date, neither the Company nor any of its Subsidiaries has received any notice, oral or written, and none of them has any reason to believe that any distributors, sales representatives, sales agents, or other third party sellers engaged by the Company or any of its Subsidiaries to sell or market the Products, will not sell or market the products or services of the Company and its Subsidiaries at any time after the consummation of the transaction contemplated by this Deed on terms and conditions substantially similar to those used in the current sales and distribution contracts of the Company and its Subsidiaries with such distributors, sales representatives, sales agents or third party sellers.

19. Insurance.

Section 19 of the Disclosure Letter contains a list of all policies of title, property, fire, casualty, liability, life, business interruption, product liability, sprinkler and water damage, workmen's compensation, libel and slander, and other forms of insurance of any kind relating to the business and operations of the Company or any of its Subsidiaries, in each case which are in force as of the Completion Date (the "**Insurance Policies**"). As of the Completion Date: (a) all of the Insurance Policies are maintained with reputable insurance carriers, provide adequate coverage for all normal risks incident to the business of the Company, its Subsidiaries and its and their properties and assets, and are in character and amount similar to that carried by similarly situated persons engaged in similar businesses and subject to the same or similar perils or hazards; (b) the Company and its Subsidiaries have made any and all payments required to maintain the Insurance Policies in full force and effect; and (c) neither the Company nor any of its Subsidiaries has received written notice of default under any Insurance Policy, and has not received written notice of any pending or threatened termination or cancellation, coverage limitation or reduction or premium increase with respect to any Insurance Policy.

20. Transactions with Affiliates.

- 20.1 For purposes of this Section 20, the term "**Affiliated Person**" means (a) any direct or indirect holder of more than five per cent (5%) of the Shares, (b) any director, officer, senior executive, consultant or adviser of the Company or any of its Subsidiaries, (c) any member of the immediate family of any of such persons, or (d) any person that is controlled by any of the foregoing.
- 20.2 Since the Accounts Date, neither the Company nor any of its Subsidiaries has, in the ordinary course of business or otherwise, (a) purchased, leased or otherwise acquired any property or assets or obtained any services from, (b) sold, leased or otherwise disposed of any property or assets or provided any services to (except with respect to remuneration for services rendered in the ordinary course of business as director, officer or employee of the Company or any Subsidiary), (c) entered into or modified in any manner any contract with, or (d) borrowed any money from, or made or forgiven any loan or other advance (other than expenses or similar advances made in the ordinary course of business) to, any Affiliated Person.
- 20.3 The (a) contracts of the Company and its Subsidiaries do not include any obligation or commitment between the Company or any of its Subsidiaries, on the one hand, and any Affiliated Person, on the other hand, (b) assets of the Company and its Subsidiaries do not include any receivable or other obligation or commitment from an Affiliated Person to the Company or any of its Subsidiaries and (c) liabilities of the Company and its Subsidiaries do not include any payable

or other obligation or commitment from the Company or any of its Subsidiaries to any Affiliated Person.

20.4 No Affiliated Person is a party to any contract with any customer or supplier of the Company or any of its Subsidiaries that affects in any manner the business, financial condition or results of operation of the Company or any of its Subsidiaries.

21. Accuracy of Information.

None of the representations or warranties of the Company contained in this Deed or any of the other Transaction Documents and none of the information contained in any schedule, certificate, or other document delivered pursuant hereto or thereto or in connection with the transactions contemplated hereby or thereby contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements herein or therein not misleading. No information furnished by the Company to Qorvo or its Representatives in connection with this Deed (including all information in the Disclosure Letter and the other Schedules to this Deed) or the other Transaction Documents contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements so made or information so delivered not misleading. On the date hereof, the Company has delivered to Qorvo DVDs that contain all information and documents contained in the Data Room. On the Completion Date, the Company will deliver to Qorvo DVDs that contain all information and documents contained in the Data Room as of Completion.

22. Title to and Sufficiency of Assets.

- 22.1 The Company and its Subsidiaries own good and marketable title to all of its and their assets constituting personal property (excluding, for purposes of this sentence, assets held under leases), free and clear of any and all Liens. Such assets, together with all assets held by the Company and its Subsidiaries under leases and licenses of Intellectual Property, include all tangible and intangible personal property, contracts and rights necessary or required for the operation of the businesses of the Company and its Subsidiaries as presently conducted.
- 22.2 Neither the Company nor any of its Subsidiaries owns any Real Estate. All Real Estate leases held by the Company and its Subsidiaries are adequate for the operation of the businesses of the Company and its Subsidiaries as presently conducted. The leases to all Real Estate occupied by the Company or any of its Subsidiaries are listed in Section 22.2 of the Disclosure Letter, are in full force and effect and no event has occurred which with the passage of time, the giving of notice, or both, would constitute a default or event of default by the Company or any of its Subsidiaries or, to the knowledge of the Company, any other person who is a party signatory thereto. For purposes of this Deed, "**Real Estate**" means all of the fee, if any, or leasehold ownership right, title and interest of the Company and its Subsidiaries, in and to all real estate and improvements owned or leased by the Company and its Subsidiaries.

23. Brokers.

No broker, investment banker or other person is entitled to any broker's, finder's or other similar fee or commission in connection with the transactions contemplated by this Deed or the other Transaction Documents based upon arrangements made by or on behalf of the Company or any of its Subsidiaries. Any such fee or commission set forth in the Disclosure Letter shows, with

specificity, what portion of such fee or commission is due with respect to the Completion and the transactions contemplated by this Deed, as the case may be.

24. Employees.

The employment of each officer, employee and consultant of the Company and each of its Subsidiaries (excluding the Dutch Subsidiary) is terminable at will. No employee of the Company or any of its Subsidiaries has been granted the right to continued employment by the Company or any of its Subsidiaries or to any material compensation following termination of employee or group of employees intends to terminate, has agreed or given notice of termination of his, her or their employment with the Company or any of its Subsidiaries, nor does the Company or any of its Subsidiaries have a present intention to terminate the employment of any such employees. No former employee or independent contractor or any third party otherwise engaged by the Company or any of the Subsidiaries has a statutory or contractual right to return to work for the Company or any Subsidiary. Neither the Company nor any of its Subsidiaries has entered into or made or proposed any agreement, arrangement or commitment: (a) to make any payment or to grant any loan or other benefit to any employee other than in respect of salary or standard benefits; or (b) to change any terms of employment or to increase or adjust any remuneration or to make any bonus or incentive payments or any benefits in kind; or (c) prohibiting the giving of notice or obliging the Dutch Subsidiary to observe a notice period in excess of four (4) months.

25. Effectiveness of this Deed.

- 25.1 The Articles of Association as set out in <u>Schedule 12</u> have been duly adopted as at the date of this Deed as the governing articles of association of the Company.
- 25.2 The Company has sufficient authority as at the date of this Deed to effect the issuance of sufficient share capital to enable the issuance of sufficient Series F Preferred Shares to Qorvo in the event that the Additional Funding occurs pursuant to clause 4.2 of this Deed.
- 25.3 The terms and conditions of the Employee Share Schemes as in force at the date of this Deed (and as amended pursuant to clause 8.2 and clause 8.5) (and any supplemental documents or agreements granting any awards of Contingent Securities pursuant to such Employee Share Schemes) permit and enable the implementation of this Deed with respect to the Contingent Securities.
- 25.4 The Committed Shareholders and the Major Investors, in aggregate, hold at least [*****] of the Ordinary Shares of the Company as at the date of this Deed and shall hold at least [*****] of the Ordinary Shares of the Company as at Completion.
- 25.5 The Committed Shareholders and the Major Investors, in aggregate, hold at least [*****] of the Series F Preferred Shares which are not held by Qorvo as at the date of this Deed and shall hold at least [*****] of the Series F Preferred Shares which are not held by Qorvo as at Completion.
- 25.6 The Committed Shareholders and the Major Investors, in aggregate, hold at least [*****] of each of:
 - (i) the Series A Preferred Shares;

- (ii) the Series B Preferred Shares;
- (iii) the Series B-1 Preferred Shares;
- (iv) the Series C Preferred Shares;
- (v) the Series D Preferred Shares; and
- (vi) the Series E Preferred Shares,

of the Company as at the date of this Deed and shall hold such amounts as at Completion.

25.7 The Executive Optionholders, in aggregate, hold at least [*****] of the Contingent Securities of the Company as at the date of this Deed and shall hold at least [*****] of the Contingent Securities of the Company as at Completion.

WARRANTORS

Each of the parties named below also enters into this Deed as a "Warrantor":

- 1. each of the [*****] Investors;
- 2. [*****];
- 3. each Committed Shareholder who executes this Deed on the date of this Deed, and any other Committed Shareholder who accedes to this Deed as a Warrantor in accordance with <u>clause 3</u> or otherwise; and
- 4. each Executive Optionholder who executes this Deed on the date of this Deed, and any other Executive Optionholder who accedes to this Deed as a Warrantor in accordance with <u>clause 3</u> or otherwise.

FORM OF DEED OF ADHERENCE

[OMITTED]

EFFECTIVE TIME STATEMENTS

PART A EFFECTIVE TIME STATEMENTS

- (1) After Completion, Qorvo shall procure the preparation on behalf of Qorvo of:
 - (a) draft Effective Time Statements in accordance with Schedule 14 (Accounting Policies); and
 - (b) a draft Payment Statement.
- (2) Qorvo shall procure that the draft Effective Time Statements and the draft Payment Statement are submitted to the Securityholder Agent within twenty (20) Business Days after (and excluding) the Completion Date.
- (3) In order to enable the Securityholder Agent, Selling Shareholders and the Sellers' Accountants to review the draft Effective Time Statements and the draft Payment Statement, Qorvo shall, on reasonable notice and subject to the execution of reasonably appropriate hold harmless letters and confidentiality obligations:
 - (a) make available to the Securityholder Agent and Sellers' Accountants during normal office hours:
 - (i) its working papers; and
 - such books and records relating to the business of the Group and in the possession or under the control of Qorvo as they may reasonably require; and
 - (b) cooperate with the Securityholder Agent and Selling Shareholders with regards to the provision of such information as is reasonably required by the Selling Shareholders in connection with their review of the draft Effective Time Statements and the draft Payment Statement and shall make available the services of the employees of the Group in connection with the provision of such information, provided that this does not materially interfere with the proper operation of the Group after Completion.
 - (c) cooperate with the Securityholder Agent and Selling Shareholders with regards to the provision of such information as is reasonably required by the Selling Shareholders in connection with their review of the draft Effective Time Statements and the draft

Payment Statement and shall make available the services of the employees of the Group in connection with the provision of such information, provided that this does not materially interfere with the proper operation of the Group after Completion.

- (4) Within twenty (20) Business Days after (and excluding) the date of receipt of the draft Effective Time Statements and the draft Payment Statement, and contingent upon Qorvo's fulfilment in all material respects of its obligations set forth in paragraph 3 of this <u>Schedule 11</u>, the Securityholder Agent shall serve notice on Qorvo that either:
 - (a) the draft Effective Time Statements are agreed; or
 - (b) the draft Effective Time Statements are not agreed, in which case such notice (a "Sellers' Disagreement Notice") shall set out in reasonable detail:
 - (i) those amounts with which the Selling Shareholders disagree;
 - (ii) the quantum (in the Selling Shareholders' opinion) of such amounts; and
 - (iii) the reasons why the Selling Shareholders disagree with those items,

and the Selling Shareholders and the Sellers' Accountants shall, subject to the execution of appropriate hold harmless letters, make available their working papers to Qorvo and Qorvo's Accountants.

- (5) If the Securityholder Agent:
 - (a) notifies Qorvo in accordance with paragraph 4(a) above that the draft Effective Time Statements and the draft Payment Statement are agreed; or
 - (b) fails to give notice to Qorvo as required by paragraph 4 above,

the draft Effective Time Statements and the draft Payment Statement shall be final and binding and shall, for the purposes of this Deed, be treated as agreed on the date on which such notice is served or the next date after the last date on which such notice could be served (as appropriate).

(6) If the Securityholder Agent serves a Sellers' Disagreement Notice in accordance with <u>paragraph 4(b)</u> above, Qorvo and the Securityholder Agent shall use reasonable endeavours to resolve the matters in dispute within twenty (20) Business Days after (and excluding) the day of receipt of the Sellers' Disagreement Notice.

- (7) If the disputed items are agreed within the period specified in <u>paragraph 6</u> above, the draft Effective Time Statements and the draft Payment Statement (in each case amended as agreed) shall be final and binding and shall, for the purposes of this Deed, be treated as agreed on the date on which such agreement is reached.
- (8) If the disputed matters are not agreed within the period specified in <u>paragraph 6</u> above:
 - (a) either Qorvo or the Securityholder Agent may refer the remaining matters in dispute to Ernst & Young LLP (or if Ernst & Young LLP is not available, such other independent firm of chartered accountants as may be agreed between them in writing); and
 - (b) if Ernst & Young LLP is not available and Qorvo and the Securityholder Agent are unable to agree on such appointment within a further fifteen (15) Business Days, the remaining matters in dispute shall be referred to such independent firm of chartered accountants as may be nominated by the President for the time being of the Institute of Chartered Accountants in England and Wales on the application of either Qorvo or the Securityholder Agent.
- (9) The independent firm of chartered accountants appointed in accordance with <u>paragraph 8</u> above (the "**Expert**") shall act on the following basis:
 - (a) the Expert shall act as an expert and not as an arbitrator;
 - (b) the Expert's terms of reference shall be limited to determining the remaining matters in dispute and, in doing so, the Expert shall apply the Accounting Policies set out in Schedule 14 (Accounting Policies);
 - (c) except to the extent that Qorvo and the Securityholder Agent agree otherwise, the Expert shall determine its own procedure but the procedure of the Expert shall:
 - (i) give the parties a reasonable opportunity to make written and oral representations;
 - (ii) require that the Selling Shareholders supply to Qorvo, and Qorvo supplies to the Securityholder Agent, a copy of any written representations at the same time as they are supplied to the Expert; and
 - (iii) permit each party to be present while oral representations are being made to the Expert by any other party; and

- (d) the Expert's costs (including any fees and costs of any advisers appointed by the Expert) shall be borne equally by the Selling Shareholders (jointly and severally) and Qorvo, or as the Expert may determine.
- (10) The parties shall cooperate with the Expert, and shall comply with all reasonable requests made by the Expert in connection with the carrying out of its duties. In particular, but without limitation, the parties shall provide the Expert with all information reasonably required by the Expert in relation to any Group Company, and the Expert shall be entitled (to the extent it considers appropriate) to base its determination on such information and on the accounting and other records of each Group Company.
- (11) The determination of the Expert pursuant to <u>paragraph 9</u> above shall:
 - (a) be made as soon as is reasonably practicable, and in any event within twenty (20) Business Days after receipt of all necessary information from the parties;
 - (b) be made in writing and, unless otherwise agreed by the Selling Shareholders and Qorvo, include reasons for each relevant determination;
 - (c) be sent simultaneously by the Expert to Qorvo and the Securityholder Agent by email to addresses provided by them for the purpose; and
 - (d) be final and binding on the parties in respect of any matter falling within the Expert's jurisdiction (save in the event of fraud or manifest error, in which case the relevant part of the determination shall be void and the matter shall be referred to another firm of independent accountants on the terms of this Schedule 11 (Effective Time Statements) for correction),

and the Effective Time Statements and the Payment Statement (in each case amended as determined by the Expert) shall, for the purposes of this Deed, be treated as determined on the date on which the Expert's determination is sent to the parties.

- (12) Without prejudice to <u>clause 39</u> (Confidentiality), (and subject to the last sentence of clause 39.1), each of the parties shall, and shall procure that their respective advisers and the Expert shall:
 - (a) keep confidential all information and documents provided to them pursuant to this Schedule 11 (Effective Time Statements); and

- (b) use or disclose such information and documents solely in connection with the preparation of the Effective Time Statements, the proceedings of the Expert or in defending any claim or argument or alleged claim or argument relating to this Deed or its subject matter.
- (13) Nothing in this <u>Schedule 11</u> (Effective Time Statements) shall entitle any party or the Expert access to any information or document that is protected by legal professional privilege or litigation privilege, provided that this <u>paragraph 13</u> shall not entitle any party to refuse to disclose such parts of any documents as contain only the facts on which the relevant claim or argument is based.

PART B ACTUAL NET DEBT STATEMENT

	Estimated Balances (\$000's) (Value A)	Balances at the Effective Time (\$000's) (Value B)	Difference (\$000's) (Value B less Value A)
[External Debt]	[•]	[X]	[X]
[Cash]	([•])	([X])	[X]
	Estimated Net Debt Amount	Actual Net Debt Amount	Difference
	[•]	[•]	[•]
Net Debt Shortfall (if any) (\$000's)			
Net Debt Excess (if any) (\$000's)			

		Estimated Balances (\$000's) (Value A)	Balances at the Effective Time (\$000's) (Value B)	Difference (\$000's) (Value B less Value A)
1	[Trade debtors]	[•]	[X]	[X]
2	[Stock]	[•]	[X]	[X]
3	[Prepayments]	[•]	[X]	[X]
4	[Other current assets]	[•]	[X]	[X]
Current Assets		[•]	[X]	[X]
5	[Trade creditors]	([•])	([X])	[X]
6	[Accruals – payroll]	([•])	([X])	[X]
7	[Accruals – holiday pay]	([•])	([X])	[X]
8	[Accruals – PAYE and National Insurance Contributions]	([•])	([X])	[X]
9	[Accruals – Severance Obligations]	([•])	([X])	[X]
10	[Accruals – other]	([•])	([X])	[X]
11	[VAT]			
12	[Other current liabilities]	([•])	([X])	[X]
Current Liabilities		([•])	([X])	[X]
		Working Capital Target	Actual Working Capital Amount	Difference
		[•]	[•]	[•]
Worki	ing Capital Shortfall (if any) (\$000's)	1		
Worki	ing Capital Excess (if any) (\$000's)			

PART C ACTUAL WORKING CAPITAL STATEMENT

ARTICLES OF ASSOCIATION

[OMITTED]

TAX COVENANT

1. USING DEFINITIONS

1.1 The definitions and rules of interpretation in this paragraph apply in this <u>Schedule 13</u>.

Accounts Relief: a Relief the availability of which has been:

- (a) shown as an asset (including a right to repayment of Tax) in the Actual Net Debt Statement; or
- (b) taken into account in computing (and so reducing or eliminating) any provision for deferred Tax in the Actual Net Debt Statement (or which, but for the Relief, would have appeared in the Actual Net Debt Statement).

Auditors: the auditors for the time being of a Group Company and in any case where that shall mean more than one firm of auditors, the auditors of the company that has incurred the Tax Liability at issue.

Buyer's Relief: any:

- (a) Accounts Relief;
- (b) Post-Completion Relief; or
- (c) Relief arising to any member of Buyer's Tax Group (other than a Group Company) at any time.

Buyer's Tax Group: Qorvo and any company in the same Tax Group from time to time.

Default Rate: 4% over the base rate from time to time of Barclays Bank.

Event: includes (without limitation) any event, transaction (including, without limitation, the execution of and Completion of this Deed), payment, action, circumstance, dealing, state of affairs, expiry of any time period, default, omission or occurrence of any nature whatsoever and whether or not a Group Company or Qorvo is a party to it, and also the death or the winding up or dissolution of any person, any change in residence of a person for the purposes of any Tax, any failure to take action that would have prevented or avoided an apportionment or deemed distribution of income (regardless of whether any action taken after the Effective Time could have prevented or avoided the apportionment or deemed distribution), a Group Company becoming, being or ceasing to be a member of a group of companies (however defined) or becoming or ceasing to be associated or connected with any person for the purposes of any Tax, and in any one case or more whether alone or in any combination.

Group Relief: any Relief available between members of a Tax Group.

Income, Profits or Gains: income, profits, gains and any other consideration, value, receipt or measure by reference to which Tax is chargeable or assessed.

Indirect Tax: value added, turnover, sales, use, distribution or corresponding Tax.

Overprovision: the amount by which any provision in the Actual Net Debt Statement relating to Tax (other than a provision for deferred Tax) is overstated (except to the extent that any overstatement results from the utilisation of a Buyer's Relief), applying the accounting policies, principles and practices adopted in relation to the preparation of the Actual Net Debt Statement (and ignoring the effect of any change in law made after the Effective Time).

PAYE: any system for the deduction and withholding of Tax or National Insurance or social security contributions (or similar or corresponding obligations) or both from sums paid to employees in respect of their employment.

Post-Completion Date Relief: a Relief that arises to a Group Company as a result of or in connection with any:

- (a) Event occurring; or
- (b) Income, Profits or Gains earned, accrued or received,

after the Effective Time.

Relief: includes any relief, loss, allowance, credit, deduction, exemption or set-off in respect of any Tax or relevant to the computation of any Income, Profits or Gains for the purposes of any Tax, or any right to a repayment of or saving of Tax (including any repayment supplement or interest in respect of Tax), and:

- (c) any reference to the use or set-off of a Relief shall be construed accordingly and shall include use or set-off in part;
- (d) any reference to the loss of a Relief (including the loss of any Accounts Relief, Post-Completion Relief and any other defined Relief) shall include the absence, failure to obtain, non-existence, non-availability, disallowance, withdrawal, clawback or cancellation of any Relief, or its utilisation or set-off by any person other than a member of Buyer's Tax Group and shall also include this Relief being available only in a reduced amount and "lost" as it relates to a Relief shall be construed accordingly.

Tax Claim: in each case from which it appears that there is, or is likely to be, a liability or increased liability of a Group Company in respect of which the Warrantors may be liable under <u>paragraph 2</u>:

(e) any assessment, self-assessment, notice, letter, determination, demand or other document issued or action taken by or on behalf of any Taxation Authority (whether issued or taken before or after the date of this Deed and whether satisfied or not at the date of this Deed) including, for the avoidance of doubt, the imposition of any withholding of or on account of tax; and

(f) the preparation or submission to a Taxation Authority of any return, amended return, computation, accounts or any other documents by Qorvo, a Group Company or another person.

Tax Counsel: an advocate specialising in tax matters.

Tax Group: those companies treated for the purposes of determining the amount of or liability for or relief from any Tax as being members of the same group of companies or fiscal unity.

Tax Liability: means:

- (g) a liability to make an actual payment or increased payment of, or in respect of, or on account of, Tax (including making a payment in settlement of a Tax liability and whether or not presently payable and whether satisfied or unsatisfied at the Effective Time and also including any liability of a Group Company to make any payment of Tax in connection with the exercise, release, vesting, variation or cancellation of a right obtained before the Effective Time to acquire, or otherwise relating to, shares) ("Actual Tax Liability"), in which case the amount of the Tax Liability shall be the amount of the actual payment or increased payment;
- (h) the loss otherwise than by utilisation or set off of an Accounts Relief, in which case the amount of the tax liability shall be the amount of Tax that would have been saved but for such loss (assuming that a Group Company had used the Relief in full and that, to the extent that there is an Actual Tax Liability as a result of the loss, the Tax would have been charged at the Tax rates applying to that period and, where there is no Actual Tax Liability, the Tax that would have been saved would have been charged at the Tax rates current at the Accounts Date) or, where the Relief is a right to repayment of Tax, the amount of the repayment; or
- (i) the use or setting off of any Buyer's Relief in circumstances where, but for such use or setting off, a Group Company would have had an Actual Tax Liability in respect of which the Warrantors would have been liable under this <u>Schedule 13</u>, in which case the amount of the Tax Liability shall be the amount for which the Warrantors would have been liable under this <u>Schedule 13</u> but for such set-off or utilisation.

Tax Warranties: the Warranties which relate to Taxation.

- 1.2 In this Schedule:
 - (a) references to Income, Profits or Gains earned, accrued or received (or to an Event occurring) on or before a particular date (including Completion) or in respect of a particular period shall include Income, Profits or Gains that are deemed for the purposes of any Tax to have been earned, accrued or received (or, as the case may be, an Event that is deemed to have occurred) on or before that date or in respect of that period as the case may be;
 - (b) references to a payment or distribution made on or before a particular date shall include:

- $(i) \quad any \ payment \ or \ distribution \ that \ on \ or \ before \ that \ date \ has \ fallen \ due; \ and$
- (ii) any act or transaction that has occurred on or before that date and is or has been deemed to be a payment or distribution for the purposes of any Tax.
- (c) words and expressions defined in this Deed have the same meanings in this Schedule and any provisions in this Deed concerning matters of interpretation or construction shall apply equally to this Schedule except as otherwise provided in this Schedule;
- (d) references to paragraphs are references to paragraphs of this Schedule unless stated otherwise;
- (e) reference to any Event shall include the combined result of two or more Events, the first of which has taken place or the commencement of which has occurred on or before the Effective Time, and for the avoidance of doubt this shall include the combined result of Events all occurring on or before the Effective Time and the combined result of Events where more than one Event occurs on or before the Effective Time where the Events occurring on or before the Effective Time are outside the ordinary course of business of a Group Company as carried on at the time of the Event and the Events occurring after the Effective Time are inside the ordinary course of business of a Group Company as carried on at the Effective Time;
- (f) references to a repayment of Tax include any repayment supplement or interest in respect of it;
- (g) any stamp duty or transfer or registration tax that would be payable on any document, provided such document is either necessary to establish the title of a Group Company to any asset or is a document in the enforcement or production of which a Group Company is interested, and any interest, fine or penalty relating to any such duty or tax, will be deemed to be an Actual Tax Liability of a Group Company arising in consequence of an Event occurring on the last date it would have been necessary to pay such duty or tax to avoid any liability to interest or penalties on paying it;
- (h) for the avoidance of doubt, references to any Tax Liability of a Group Company that results from any gains earned, received or accruing on or before the Effective Time or any Event on or before the Effective Time include a reference to any Tax Liability of a Group Company arising as a result of Completion or of entering into this Deed, or of the satisfaction of any condition in this Deed, or as a result of forming the intention, or of entering into arrangements, to enter into this Deed (including any liability as the result of any company ceasing to be, or ceasing to be treated as, a member of a group of companies for the purpose of any Tax);
- (i) the word **connected** shall be construed in accordance with the provisions of section 1122 of the Corporation Tax Act 2010;
- (j) where references are made to any notice to be given, or to any other communication that is to be made, to the Warrantors by Qorvo, any reference to the Warrantors shall include

any representative of the Warrantors agreed between the parties under this Deed or otherwise;

- (k) references to the **due date for payment** of any Tax shall be read and construed as a reference to the last day on which such Tax may by law be paid without incurring a penalty or liability for any interest, charge, surcharge, penalty, fine or other similar imposition accruing (after taking into account any postponement of such date that is obtained for such Tax); and
- (1) for the avoidance of doubt, references to any Tax Liability of a Group Company in consequence of or in respect of or by reference to an Event which occurred on or before the Effective Time shall include a reference to any Tax Liability of a Group Company resulting from the sale of the Shares and Contingent Securities under this Deed.
- 1.3 All payments made by the Warrantors to Qorvo or by Qorvo to the Warrantors under this Schedule, shall, so far as possible, be made by way of adjustment to the consideration for the sale of the Shares and Contingent Securities.

2. COVENANT

Under a covenant on the terms set out in this paragraph 2 and subject to paragraph 3, the Warrantors shall pay to Qorvo an amount equal to the Indemnifiable Portion of:

- 2.1 Any Tax Liability of a Group Company that arises:
 - (a) in consequence of or in respect of or by reference to an Event that occurred on or before the Effective Time;
 - (b) in consequence of or in respect of or by reference to any Income, Profits or Gains that were earned, accrued or received on or before the Effective Time;
 - (c) from the grant, exercise, release, vesting, variation or cancellation at any time of a right acquired before the Effective Time to acquire securities or an interest in securities; or
 - (d) in consequence of any failure to discharge or default in discharging any of the Warrantors' obligations under paragraph 6 and paragraph 9, including any failure to meet any relevant time limit.
- 2.2 Any Tax Liability that is the liability of another person ("**Primary Person**") for which a Group Company, Qorvo or any other member of Buyer's Tax Group is liable in consequence of:
 - (a) the Primary Person failing to discharge such Tax Liability; or
 - (b) a Group Company at any time before the Effective Time being:
 - (i) a member of the same Tax Group as the Primary Person; or

- (ii) treated for the purpose of any Tax as having control of, being controlled by, or being otherwise connected with, the Primary Person or being controlled by or connected with the same person as the Primary Person for any Tax purpose.
- 2.3 Any Tax Liability of a Group Company arising as a result of or in connection with:
 - (a) any liability to operate PAYE or to deduct or account for National Insurance or social security contributions (or similar or corresponding obligations) as a result of or in connection with the issue or transfer of securities or an interest in securities on or before the Effective Time; or
 - (b) any payments to the Selling Shareholders under this Deed.
- 2.4 Any Tax Liability in respect of a gain that arises as a result of the disposal at any time of any asset acquired by a Group Company before the Effective Time, to the extent that such Tax Liability would not have arisen if the expenditure allowable or deductible in respect of the asset (ignoring any other Relief) had not been less than the value of the asset stated or recognised in the Accounts or, in the case of an asset acquired since the Accounts Date, the cost of that asset.
- 2.5 Any liability of a Group Company to make a payment, or to make a repayment of the whole or any part of any payment, to any person (other than a member of Buyer's Tax Group) in respect of Group Relief under any arrangement or agreement entered into by a Group Company on or before the Effective Time except to the extent that such payment or repayment is reflected in the Actual Net Debt Statement.
- 2.6 The loss in whole or in part of the right of a Group Company to receive any payment (other than from a member of Buyer's Tax Group) for Group Relief under any arrangement or agreement entered into on or before the Effective Time where such payment was taken into account in the Actual Net Debt Statement.
- 2.7 Any liability of a Group Company to make a payment or repayment under any indemnity, covenant, warranty, mortgage, guarantee or charge entered into or created on or before the Effective Time of a sum equivalent to or by reference to another person's Tax liability.
- 2.8 Any reasonable costs and expenses suffered or properly incurred by Qorvo or a Group Company or any member of Buyer's Tax Group in connection with:
 - (a) any Tax Claim in respect of which the Warrantors may be liable under paragraph 2.1 to paragraph 2.8; or
 - (b) taking or defending any action under this Schedule.

3. EXCLUSIONS

- 3.1 The Warrantors shall not be liable under paragraph 2 (or, as appropriate, the liability of the Warrantors shall be reduced) in respect of any liability of a Group Company:
 - (a) to the extent that such liability was taken into account in computing the amount of the External Debt for the purposes of the Actual Net Debt Statement;

- (b) to the extent that the liability is increased as a result only of any increase in rates of Tax that is announced and comes into force after the Effective Time with retrospective effect or of any change in law that is announced and comes into force after the Effective Time with retrospective effect but this paragraph 3.1(b) shall not apply to any amount payable under paragraph 11;
- (c) to the extent that the liability (other than a Tax Liability mentioned in paragraph 2.4) would not have arisen but for a voluntary act, omission or transaction of a Group Company after the Effective Time or of Qorvo or any member of Buyer's Tax Group except where such act, omission or transaction was:
 - (i) carried out or effected under a legally binding obligation entered into on or before the Effective Time;
 - (ii) required by law, any regulatory, financial reporting or accounting practice or requirement;
 - (iii) carried out with the written consent of the Warrantors;
 - (iv) in the ordinary course of business of a Group Company (or, as appropriate, of Qorvo or a member of Buyer's Tax Group);
 - (v) the preparation or submission of any tax return or computation or other information to a Taxation Authority;
 - (vi) one that Qorvo or the party in question could not foresee or could not reasonably have foreseen would give rise to that liability; or
 - (vii) one that could not reasonably have been avoided; and
- (d) to the extent recovery (less costs and expenses of recovery) has been made by Qorvo under the Warranties or any other provision of the Agreement.
- 3.2 The provisions of paragraph 3.1 will not apply to any claim by Qorvo against the Warrantors that arises or is delayed as a result of fraud, dishonesty, wilful misconduct or deliberate concealment by the Warrantors.
- 3.3 The provisions of paragraph 3.1 and clause 13.8 of the Agreement shall not apply to any claims in respect of any liability falling within paragraph 2.2 or paragraph 2.8.

4. LIMITATIONS

- 4.1 Notwithstanding the provisions of clause 13.7 of the Agreement, the liability of the Warrantors under paragraph 2 will terminate on:
 - (a) the twenty-second anniversary of the Completion Date, in respect of any claim under paragraph 2 for a liability arising from a loss of tax brought about deliberately by a Group Company or any related person; or

(b) the seventh anniversary of the Completion Date, in any other case, except in respect of any claim under paragraph 2 of which notice is given or deemed to have been given to the Warrantors before that date in accordance with paragraph 6.1 or paragraph 6.2, as the case may be.

5. RECOVERY FROM THIRD PARTIES

- 5.1 Where a payment has been made by the Warrantors under paragraph 2 in full discharge of a liability under that paragraph and a Group Company or any other member of Buyer's Tax Group has recovered or subsequently recovers from any person (other than Qorvo, a Group Company or any member of Buyer's Tax Group or any employee or director of any of them but including a Taxation Authority) any sum in respect of such Tax Liability (other than by reason of the use or set off of a Buyer's Relief), Qorvo shall inform the Securityholder Agent and shall at the written request of the Securityholder Agent (and shall procure that a Group Company or the relevant member of Buyer's Tax Group shall) account to the Warrantors for the lesser of:
 - (a) any amount recovered and actually received by a Group Company or the member of Buyer's Tax Group as applicable net of any Tax on such amount and net of any Tax that would have been payable but for the use or set off of any Relief; and
 - (b) the amount paid by the Warrantors under paragraph 2 in respect of that Tax Liability.
- 5.2 Any payment by Qorvo to the Warrantors under this paragraph 5 shall be reduced by the amount of any losses, damages, liabilities, costs and expenses suffered or incurred by a Group Company and Qorvo and any other member of Buyer's Tax Group in recovering any amount from any third party and shall be further reduced by any amount of the payment by the Warrantors referred to in paragraph 5.1 that has previously been repaid to the Warrantors.

6. CONDUCT OF CLAIMS

- 6.1 If Qorvo or a Group Company becomes aware of a Tax Claim that could give rise to a liability of the Warrantors under paragraph 2, Qorvo shall or shall procure that a Group Company shall, within a reasonable time, give written notice to the Securityholder Agent of the Tax Claim containing, to the extent reasonably practicable, a description of such claim and the estimated total amount of the claim, provided that the giving of such notice shall not be a condition precedent to the liability of the Warrantors.
- 6.2 Subject to the following provisions of this paragraph 6, Qorvo shall (and where relevant, shall procure that a Group Company shall) take such action as the Securityholder Agent may reasonably request by notice in writing to avoid, dispute, defend, resist, appeal or compromise any Tax Claim ("**Disputed Tax Claim**") or any matter relating to the Disputed Tax Claim but only provided that the Warrantors shall first (to Qorvo's reasonable satisfaction) indemnify Qorvo and/or a Group Company (as required by Qorvo) against the Indemnifiable Portion of any losses, damages, costs or expenses that it may suffer or incur as a result of taking this action (including any additional liability to Tax).
- 6.3 By agreement in writing between Qorvo and the Securityholder Agent, and on the terms as they may agree from time to time, the Warrantors may instruct independent lawyers or other

professional advisers (appointed by them for this purpose) to take any action mentioned in paragraph 6.2 provided that unless Qorvo and the Securityholder Agent specifically agree otherwise in writing, the Warrantors shall:

- (a) keep Qorvo fully and promptly informed of all matters known to them, or to the Warrantors' advisers, concerning the Disputed Tax Claim;
- (b) promptly provide Qorvo with copies of all documents, notes (including notes of conversations and meetings) and correspondence relating to the Disputed Tax Claim;
- (c) obtain Qorvo's prior written approval (not to be unreasonably withheld or delayed) to the appointment of lawyers or other professional advisers;
- (d) submit to Qorvo for prior written approval (not to be unreasonably withheld or delayed) any communication (written or otherwise) relating to the Disputed Tax Claim and shall make any amendments Qorvo shall request; and
- (e) not settle or compromise the Disputed Tax Claim or agree any matter in the conduct of the Disputed Tax Claim without Qorvo's prior written approval (not to be unreasonably withheld and delayed).
- 6.4 Without prejudice to the liability of the Warrantors under this Schedule, Qorvo shall not be obliged to take or procure the taking of any action under paragraph 6.2 in respect of any Tax Claim:
 - (a) to the extent that it would involve a Group Company contesting any Disputed Tax Claim before any court or other appellate body (excluding the Taxation Authority that has made the Disputed Tax Claim) unless in the written opinion of Tax Counsel of appropriate relevant experience appointed by agreement between the Securityholder Agent and Qorvo an appeal would be reasonable in all the circumstances having regard to the amounts involved, the likelihood of success and any future Tax Liability or other liability of a Group Company or Buyer's Tax Group to the extent that such liability is not or may not be recoverable under paragraph 2;
 - (b) where the Tax Claim or action derives from or arises out of or is in connection with any dishonest or fraudulent act or omission or wilful default by or of any of the Warrantors at any time or by or of a Group Company before the Effective Time or if Qorvo has reasonable grounds to so believe;
 - (c) should any of the Warrantors fail to comply in a material respect with their obligations under paragraph 6.3;
 - (d) following receipt of written notice of the Tax Claim from Qorvo in accordance with paragraph 6.1 should:
 - (i) the Securityholder Agent fail within twenty (20) Business Days (or, in a case where there is a time limit for an appeal that has been notified to the Sellers' Representative, fail within ten (10) calendar days) to serve notice on Qorvo under paragraph 6.2;

- (ii) the Securityholder Agent fail within twenty (20) Business Days to notify Qorvo in writing of any further action to be taken by Qorvo or a Group Company under paragraph 6.2 where Qorvo at any time seeks instructions from the Securityholder Agent;
- (iii) the Warrantors fail to adequately indemnify and secure Qorvo and/or a Group Company (as appropriate) under paragraph 6.2; or
- (iv) the Warrantors fail to further indemnify Qorvo or fail to provide additional security to the reasonable satisfaction of Qorvo in the event that the indemnity or security previously provided becomes, or is likely to become, in the reasonable opinion of Qorvo insufficient;
- (e) should any of the Warrantors:
 - become insolvent and corporate action or other steps are taken or legal proceedings are started for its winding up, dissolution or administration or for the appointment of a receiver, administrator, trustee or similar officer of the Warrantor or any of its assets; or
 - be unable to pay its debts as they fall due, start negotiations with a creditor with a view to the general readjustment or rescheduling of its indebtedness or make a general assignment for the benefit of, or a composition with, its creditors;
- (f) if, in Qorvo's reasonable opinion, the action requested by the Securityholder Agent under paragraph 6.2 is likely to materially affect adversely the liability of Qorvo or a Group Company to Tax or the business or financial interests of any of them or of any person connected with any of them or is unreasonable or contrary to the legal obligations of any of them or of any person connected with any of them;
- (g) if a Group Company would be required to appeal against any assessment or demand for Tax where it is a requirement for such an appeal that the Taxation be paid unless payment has previously been made by the Warrantors to Qorvo of an amount equal to that Taxation and in respect of it; or
- (h) that requires a Group Company to take any action against any person who is at the time in question either an employee or director of any member of Buyer's Tax Group, or any company that is at the time in question a member of Buyer's Tax Group.
- 6.5 If paragraph 6.2 does not apply by virtue of any provision in paragraph 6.4, Qorvo or a Group Company (as the case may be) may satisfy or settle the Tax Liability on terms as it may in its absolute discretion think fit without prejudice to any of Qorvo's rights under this Schedule.
- 6.6 Neither Qorvo nor a Group Company shall be subject to any claim by or liability to the Warrantors for non-compliance with any of the foregoing provisions of this paragraph 6 if Qorvo or a Group Company has acted in good faith in accordance with the instructions of the Warrantors.

7. DATE OF PAYMENT

- 7.1 Payment by the Warrantors in respect of any liability under this Schedule must be made in cleared and immediately available funds on the following days:
 - (a) in the case of a Tax Liability that is an Actual Tax Liability, the later of ten Business Days before the due date for payment and ten Business Days after the date Qorvo serves notice on the Warrantors requesting payment;
 - (b) in the case of a liability under paragraph 2.5, paragraph 2.6 and paragraph 2.7, ten Business Days following the date Qorvo serves notice on the Warrantors requesting payment;
 - (c) in respect of a Tax Liability that is not an Actual Tax Liability, the later of ten Business Days after the date Qorvo serves notice on the Warrantors requesting payment and:
 - (i) the due date for payment of the Tax that would have been relieved (disregarding the availability of any other Relief) in the case of the loss of or failure to obtain an Accounts Relief that is not a right to repayment of Tax;
 - (ii) the date on which the Tax would otherwise have been repaid in the case of the loss of or failure to obtain an Accounts Relief that is a right to repayment of Tax; or
 - (iii) the due date for payment of the Tax which a Group Company would have had to pay but for the use or set off of a Buyer's Relief in the case of the use or set-off of a Buyer's Relief against an Actual Tax Liability in respect of which the Warrantors would otherwise have been liable;
- 7.2 If there is any dispute as to the amount of the liability of the Warrantors under this Schedule in respect of any Tax Liability to which paragraph 7.1 applies, the matter may be referred by either the Securityholder Agent or Qorvo for determination by the Auditors (acting as experts and not as arbitrators) in which case the parties shall be bound to accept the determination of the Auditors except in cases of manifest error and the expense of the Auditors shall be borne by the parties in the same proportions as the ratio of the percentage error of their respective estimates of the amount.

8. INTEREST ON LATE PAYMENT

- 8.1 Interest shall be paid at the Default Rate on any amount due under this Schedule that has not been paid in accordance with paragraph 7, such interest to accrue from day to day (after as well as before any judgement for the same) up to and including the date on which any such amount is actually paid.
- 8.2 Interest shall be compounded on a quarterly basis and shall be payable on demand.

9. CONDUCT OF TAX AFFAIRS

- 9.1 Qorvo or its duly authorised agents shall have sole conduct of all Tax affairs of a Group Company and shall be entitled to deal with such Tax affairs in any way in which it, in its absolute discretion, considers fit.
- 9.2 Qorvo shall procure that a Group Company shall provide to the Securityholder Agent all Tax Returns relating to the any period ended on or before the Accounts Date no later than twenty (20) Business Days before the date on which such Tax Returns are required to be filed with the appropriate Taxation Authority without incurring interest or penalties. Qorvo shall further procure that a Group Company shall take the Securityholder Agent's reasonable comments into account before those Tax Returns are submitted to the appropriate Taxation Authority.
- 9.3 The Warrantors shall (at the Warrantors' expense) use reasonable endeavours to provide Qorvo and a Group Company with all reasonable assistance, co-operation and information in respect of any accounting period commencing before the Effective Time including (but not limited to) information and co-operation requested in connection with Tax Returns outstanding at the Effective Time and in connection with all negotiations, correspondence and agreements in respect of a Group Company's Tax Liability.

10. CORRESPONDING SAVINGS AND OVERPROVISIONS

- 10.1 Where a liability of the Warrantors under paragraph 2 that has been discharged in full by payment in cleared funds is in respect of a Tax Liability that has given rise to a Relief, Qorvo shall inform the Securityholder Agent and the Securityholder Agent may (at the expense of the Securityholder Agent) instruct the Auditors (acting as experts and not as arbitrators) to certify by way of a written determination:
 - (a) whether a Group Company has obtained a reduction or the elimination of a liability to make an actual payment of Tax (in respect of which paragraph 2 does not provide for a liability of the Warrantors to Qorvo) as a result of the use or set-off of such a Relief ("**Saving**"); and
 - (b) the amount of the Saving.
- 10.2 Any Saving, less any amount that has been repaid to the Warrantors (or set off against any liability of the Warrantors) under any provision of this Deed or otherwise, or which represents interest or Qorvo's costs and expenses, and further reduced by the amount of any reasonable costs and expenses of Qorvo in connection with the Saving, shall:
 - (a) first be set off against any payment then due from the Warrantors under this Deed; and
 - (b) to the extent there is an excess, Qorvo shall procure that a refund shall be made to the Warrantors of any previous payments made by the Warrantors under paragraph 2 and (not previously refunded) up to the amount of such excess.

11. DEDUCTIONS FROM PAYMENTS

- 11.1 All sums payable by the Warrantors to Qorvo under this Schedule shall be paid without any rights of counterclaim and setoff and free and clear of all deductions or withholdings of any kind, save only as may be required by law.
- 11.2 Should any of the Warrantors be required by law to make a deduction or withholding from any payment referred to in this Deed, the Warrantors shall provide such evidence of the relevant withholding as Qorvo may reasonably require and shall pay to Qorvo the sum as will, after the deduction or withholding has been made, leave Qorvo with the same amount as Qorvo would have received had no deduction or withholding been made.
- 11.3 If any sum payable by any of the Warrantors to Qorvo under this Deed is subject to Tax in the hands of Qorvo, the Warrantors shall pay such additional amount as shall ensure that the net amount received by Qorvo shall be the amount that Qorvo would have received if the payment had not been subject to Tax.
- 11.4 If Qorvo would, but for the availability of a Buyer's Relief, incur a Taxation liability falling within paragraph 11.3, it shall be deemed for the purposes of that paragraph to have incurred and paid that liability.

12. GENERAL

- 12.1 No delay or omission by Qorvo in exercising any rights under this Schedule shall prejudice such rights or be construed as a waiver or partial waiver of such rights, nor shall it exclude the further exercise of such rights.
- 12.2 Qorvo shall in its absolute discretion decide whether to make a claim under this Schedule or the Tax Warranties or both.
- 12.3 For the sake of clarity, there shall be no duplication of recovery by Qorvo under this <u>Schedule 13</u> and <u>clause 13</u> of the Deed. Also for the sake of clarity, the limitations on liability of the Warrantors set forth in <u>clause 13</u> shall not apply to this <u>Schedule 13</u>.

ACCOUNTING POLICIES

PART A PREPARATION OF EFFECTIVE TIME STATEMENTS

- (1) The Estimated Net Debt Statement and the Effective Time Statements shall be prepared:
 - (a) adopting the principles, policies and practices set out in Part D of this Schedule 14 (Accounting Policies); and then
 - (b) adopting the same accounting principles, policies, treatments and categorisations as were used in the preparation of the Accounts, as then applied, including in relation to the exercise of accounting discretion and judgement; and then
 - (c) if none of the specific accounting treatments set out in this Schedule or the Accounts deal with a matter, by reference to UK GAAP.
- (2) For the avoidance of doubt, <u>paragraph (a)</u> above shall take precedence over <u>paragraphs (b)</u> and (c) above, and <u>paragraph (b)</u> above shall take precedence over <u>paragraph (c)</u> above.

PART B ACTUAL NET DEBT AMOUNT

- (1) The Actual Net Debt Amount shall be calculated by deducting the amount of Cash as at the Effective Time from the amount of External Debt at the Effective Time.
- (2) The Actual Net Debt Statement shall take the form of the worked example set out in <u>Part B</u> of <u>Schedule 11</u> (Effective Time Statements) and shall set out the constituent elements of the Actual Net Debt Amount (including the constituent elements of the External Debt as at the Effective Time).

PART C ACTUAL WORKING CAPITAL AMOUNT

- (1) The Actual Working Capital Amount shall be calculated by deducting the Working Capital Liabilities from the Working Capital Assets (each as defined below).
- (2) The "**Working Capital Assets**" shall comprise the aggregate of the following line items in the Company's consolidation system, which is the source for the preparation of the Accounts, in each case as at the Effective Time:
 - (a) [Net Trade Debtors];

- (b) [Total Prepayments and Other Debtors];
- (c) [Total Stock],

provided that the Actual Working Capital Amount shall not include any Trapped Cash.

- (3) The "**Working Capital Liabilities**" shall comprise the aggregate of the following line items in the Company's consolidation system, which is the source for the preparation of the Accounts, in each case as at the Effective Time:
 - (a) [Total Trade Creditors];
 - (b) [Accruals Payroll];
 - (c) [Accruals Holiday Pay];
 - (d) [Accruals PAYE and National Insurance];
 - (e) [Accruals Severance Obligations];
 - (f) [Accruals Other]; and
 - (g) [VAT].
- (4) The Actual Working Capital Statement shall take the form of the worked example set out in <u>Part C</u> of <u>Schedule 11</u> (Effective Time Statements) and shall set out the constituent elements of the Actual Working Capital Amount.

PART D SPECIFIC ACCOUNTING POLICIES

- (1) The Effective Time Statements shall be prepared:
 - (a) so as to include no charge, provision, reserve or write off in respect of any costs, liabilities or charges incurred or to be incurred after the Effective Time arising as a consequence of the change of ownership of any Group Company or any change in management strategy, direction or priority which results from the change in ownership contemplated by this Deed, any other effect of this Deed or any action of Qorvo following Completion;
 - (b) to the extent not otherwise covered by the specific policies set out at paragraphs (4) to (7) (inclusive) below:

- where Qorvo receives goods or services after the Effective Time, but the liability has been settled by or will fall due to the Sellers or any of them, an asset will be recorded in the Effective Time Statements; and
- (ii) where goods or services were received by the Selling Shareholders or any of them on or before the Effective Time, but the liability will be met by or will fall due to Qorvo or any member of the Group after the Effective Time, a liability will be recorded in the Effective Time Statements.
- (2) No account shall be taken of post balance sheet events or of information that becomes available after the delivery of the draft Effective Time Statements by Qorvo to the Securityholder Agent.
- (3) There shall be no double counting of any account balances, so that no account balance is included in both the Actual Working Capital Amount and the Actual Net Debt Amount.
- (4) Save as otherwise provided in this <u>Schedule 14</u> (Accounting Policies), trade creditors and accruals shall be included in the Effective Time Statements to the extent that a liability exists and is still outstanding at the Effective Time in respect of goods and services that have been received or supplied on or before the Effective Time.
- (5) Employee bonuses earned but not paid in respect of periods terminating before the Effective Time shall be included in the Effective Time Statements. Employee bonus accruals in respect of bonus scheme periods which bridge the Effective Time will be calculated on a pro rata basis, based on performance against the bonus target up to the Effective Time.
- (6) Any severance obligations or similar obligations with respect to any employee of the Group who will not be continuing as an employee with Qorvo or the Company or any other member of the Group following the Completion shall be accrued as of the Effective Time and be included in Working Capital Liabilities.
- (7) If any Group Company (a) makes any Approved Retention Bonus (as defined below) before Completion, the amount of such Approved Retention Bonus shall be added back into and be included in Working Capital Assets, or (b) is obligated to make any Approved Retention Bonus upon or after Completion, then the amount of such Approved Retention Bonus shall not be included in Working Capital Liabilities. "Approved Retention Bonus" means any retention bonus or other similar payment that the Company makes before Completion, or the Company is obligated to make upon or after Completion, to any Employee to incentivize such Employee to accept employment with Qorvo or continue employment with a Group Company following Completion, but only if and to the extent that such retention bonus or similar payment is approved in writing by Qorvo in advance of any such payment or obligation.

- (8) The Actual Working Capital Amount shall exclude any amounts included within current assets that relate to amounts receivable in relation to the sale of properties from which any Group Company was trading on the Accounts Date or any amount receivable in relation to any property from which any Group Company was trading at the Accounts Date which was the subject of a sale and leaseback transaction prior to the Effective Time.
- (9) Both the amount receivable in relation to any insurance claim relating to loss of or damage to any asset and the corresponding liability (whether or not a legal liability exists) for reinstatement of the asset shall be included in the Actual Working Capital Amount.
- (10) Any Leakage in the period between the Effective Time and Completion shall, for the purposes of the Effective Time Statements, be deemed to have occurred immediately prior to the Effective Time and shall be taken account of in the Effective Time Statements.

DETERMINATION OF LEGACY PRODUCT REVENUE

The Legacy Product Revenue shall be set forth as a component of the Revenue Payment on the Payment Statement to be prepared and delivered pursuant to <u>Schedule 11</u>.

The "Legacy Product Revenue" shall be calculated for the Company's fiscal quarter ended immediately prior to the date of the Option Notice, and shall mean the following with respect to such fiscal quarter:

(a) the amounts invoiced and recognized as revenue in accordance with UK GAAP by the Company or any other Group Company with respect to the sale by the Company or any other Group Company to a Third Party (including bona fide distributors) of (i) the devices indicated by the following product codes as at the date of this Deed: (A) 32CK301; (B) 32CK402; (C) 32CK417; (D) 32CK503; and (E) 32CK505; and (ii) any other antenna tuner devices that are successors or subsequent generations of such devices, but excluding any [*****];

less

(b) the sum of the following for such fiscal quarter (but only to the extent included in or related to the amounts calculated under <u>subparagraph (a)</u> above):

(i) trade discounts, allowances and rebates actually accrued, granted or paid to Third Parties; plus

(ii) allowances and adjustments actually accrued or credited to customers for a product that is damaged, outdated, obsolete, returned or otherwise recalled; <u>plus</u>

(iii) charges for freight, postage, shipping, delivery, service and insurance charges; <u>plus</u>

(iv) any Tax, duty or other governmental charge; <u>plus</u>

(v) write-offs or allowances for bad debts; <u>plus</u>

(vi) ny running royalties accrued, granted and payable to a Third Party based on the selling price of a product for which revenues have been included in <u>subparagraph (a)</u>; <u>plus</u>

(vii) any ancillary revenues or related amounts invoiced and recognized as revenue in accordance with UK GAAP by the Company or any other Group Company from a Third Party in connection with the sale of any product described in <u>subparagraph (a)</u> above that does not constitute direct revenue from the sale of such a product, including (A) any milestone payments, license fees, up-front payments, expediting fees, pre-payments or deposits, (B) non-recurring engineering revenues and fees and other amounts for reimbursement of research and development expenses, (C) amounts for

reimbursement of manufacturing expenses, (D) amounts in payment for the provision of any service by the Company or any other Group Company, including consulting, design, assembly and testing services, or (E) payments made in consideration for the purchase of any other tangible or intangible assets or equity interests of the Company or any other Group Company.

Legacy Product Revenue shall be determined in accordance with UK GAAP consistently applied by the Company, except as modified or supplemented by this <u>Schedule 15</u>. In calculating Legacy Product Revenue, the payment, delivery and other relevant contractual terms with the Third Party customer must be persuasively evidenced with a written contract or other documents that show a binding obligation of the customer to take delivery of and to pay for the applicable products on or before the last day of the fiscal quarter in question.

In calculating Legacy Product Revenue sold through distribution or reseller channels, in the absence of a written contractual term in favor of the Company under which the distributor/reseller has no right to return products to the Company, revenue will not be recognized until the distributor/reseller completes the sale of the product to its customer.

In calculating Legacy Product Revenue, product pricing must be fixed and determinable at the time of sale and all revenue recognized must fully reflect future price concessions that have been granted at the close of the period.

In calculating Legacy Product Revenue, any transfer from the Company to any other Group Company, or from any other Group Company to the Company shall be disregarded and the calculation shall instead be based on the first transfer to a Third Party.

For the avoidance of doubt, "Legacy Product Revenue" shall only include revenue earned by the Company or any other Group Company in respect of the sale of applicable products that have been released for production, and shall not include any amounts of revenue from the sale or other disposition of prototypes, engineering samples, design kits, test vehicles, evaluation boards, demonstration boards and other similar items.

PREPRODUCTION RELEASE OF THE SPECIFIED PRODUCT

Purpose

This document sets forth the specifications and conditions that constitute a "Preproduction Release of the Specified Product" under the Contingent Acquisition Implementation Agreement. This would be a product consisting of [*****].

FORM OF DEED OF RELEASE AND WAIVER

AMENDMENT #1 TO EMPLOYEE SHARE SCHEME GOVERNING LTD OPTIONS

FORM OF ESCROW AGREEMENT

FORM OF PAYMENTS AGREEMENT

EXECUTED as a DEED on behalf of TRIQUINT SEMICONDUCTOR, INC., a Delaware corporation doing business as "Qorvo", by Robert A. Bruggeworth its President being a person who, in accordance with the laws of that territory, is acting under the authority of that company

/s/ Robert A. Bruggeworth____

EXECUTED as a DEED on behalf of Cavendish Kinetics Limited, a private limited company incorporated in England, by Paul Dal Santo being a person who, in accordance with the laws of that territory, is acting under the authority of that company

/s/ Paul Dal Santo____

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Robert A. Bruggeworth, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Qorvo, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2015

/s/ ROBERT A. BRUGGEWORTH

Robert A. Bruggeworth President and Chief Executive Officer

EXHIBIT 31.2

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Steven J. Buhaly, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Qorvo, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2015

/s/ STEVEN J. BUHALY

Steven J. Buhaly Chief Financial Officer

EXHIBIT 32.1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Robert A. Bruggeworth, President and Chief Executive Officer of Qorvo, Inc. (the "Company"), certify pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that, to my knowledge:

- (1) the Quarterly Report on Form 10-Q of the Company for the fiscal quarter ended October 3, 2015 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

<u>/s/ ROBERT A. BRUGGEWORTH</u> Robert A. Bruggeworth President and Chief Executive Officer

November 12, 2015

EXHIBIT 32.2

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Steven J. Buhaly, Chief Financial Officer and Secretary of Qorvo, Inc. (the "Company"), certify pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that, to my knowledge:

- (1) the Quarterly Report on Form 10-Q of the Company for the fiscal quarter ended October 3, 2015 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ STEVEN J. BUHALY Steven J. Buhaly Chief Financial Officer

November 12, 2015